

Greater Fall River Vocational School District

REQUEST FOR DESIGNER SERVICES (RFS)
FOR THE DIMAN REGIONAL VOCATIONAL TECHNICAL HIGH SCHOOL PROJECT

November 20, 2019

PREPARED BY:
Alan Minkus, AIA
Director, Project Management Services
Colliers Project Leaders

PREPARED FOR:
Deborah Kenney
Business Manager
Diman Regional Vocational Technical High School

```
THINONALLY LEFT BLANK
```

GREATER FALL RIVER VOCATIONAL SCHOOL DISTRICT

REQUEST FOR DESIGNER SERVICES (RFS)

DIMAN REGIONAL VOCATIONAL TECHNICAL HIGH SCHOOL

251 STONEHAVEN ROAD, FALL RIVER, MA 02723

November 20, 2019

Invitation: The Greater Fall River Vocational School District ("Owner") is seeking the services of a qualified "Designer" within the meaning of M.G.L. Chapter 7C, Section 44 to provide professional design and construction administration services for the Diman Regional Vocational Technical High School in Fall River, Massachusetts. Selection of a Designer will be made by the Designer Selection Panel of the Massachusetts School Building Authority ("MSBA") in accordance with the MSBA's Designer Selection Procedures.

The Owner is seeking design services to conduct a Feasibility Study which will include the development and evaluation of potential alternative solutions and continue through the Schematic Design Phase of the preferred alternative initially. Subject to the approval of a Project by the MSBA and further subject to adequate funding authorized by the Owner, the contract between the Owner and the Designer may be amended to include continued designer services through design development, construction contract documents, bidding, award of construction contract(s), construction administration, final closeout and warranty period of the potential Project. A potential approved Project may include a renovation of the existing School, a renovation of and addition to the existing School and/or new construction.

The estimated construction budget of an approved potential Project may range from \$40,000,000 to \$240,000,000 depending upon the solution that is agreed upon by the Owner and the MSBA and that is ultimately approved by a vote of the MSBA's Board of Directors. The fee for Basic Services will be negotiated.

Pursuant to M.G.L. Chapter 7C, Section 6, the Designer must agree to contract with minority and women-owned businesses as certified by the Supplier Diversity Office (SDO). The amount of participation that shall be reserved for such enterprises shall not be less than seventeen and nine tenths percent (17.9%) of the contract price for combined minority business enterprises (MBE) and women-owned business enterprises (WBE).

Applicants must include a reasonable representation of both MBE and WBE firms that meets or exceeds the combined goal. Proposed MBE/WBE participation plans that include solely MBE or solely WBE participation, or do not include a reasonable amount of participation by both MBE and WBE firms to meet the combined goal, will not be considered responsive. Applications from MBE and WBE firms as prime designers are encouraged. Where the prime Designer is an SDO certified MBE or WBE, the Designer must bring a reasonable amount of participation by a firm or firms that hold the certification which is not held by the prime Designer on the project.

The minority and women-owned business enterprises must be selected from those categories of work identified in Section F of this RFS or be assigned to tasks required under Basic Services as specifically set forth in the Contract for Designer Services as amended. Applicants are strongly encouraged to utilize multiple disciplines and firms to meet their MBE/WBE goals. Consultants to the prime Designer can team within their disciplines in order to meet the MBE/WBE goals but must state this relationship on the organizational chart (Section 6 of the application form).

For additional information on Designer qualifications see Sections E. and F. in this RFS.

A. Background:

The Greater Fall River Vocational School District consists of four member communities (Fall River, Somerset, Swansea, and Westport) and is located at 251 Stonehaven Road, Fall River, MA 02723 and currently serves approximately 1400 students in grades 9-12. The School District also currently serves 125 postgraduate students in three dedicated post-secondary programs (LPN, Electricity & Plumbing) for adult learners.

Diman Regional Vocational Technical High School ("Diman Regional") currently provides educational services in 18 vocational program areas. In addition, Diman Regional also provides education in all the traditional academic programs.

The School Building Committee ("SBC") is composed of the approved categories of representation as identified by the MSBA. In addition, the SBC members include alumni, selectmen, and town managers from the District's four member communities, demonstrating the broad range of stakeholders engaged in this Feasibility Study.

B. Project Goals and General Scope:

On or about March 2017, the Owner submitted a Statement of Interest (Attachment A) to the MSBA for Diman Regional Vocational Technical High School. The MSBA is an independent public authority that administers and funds a program for grants to eligible cities, towns, and regional school districts for school construction and renovation projects. The MSBA's grant program is discretionary, and no city, town, or regional school district has any entitlement to any funds from the MSBA. At the June 26, 2019 Board of Directors meeting, the MSBA voted to issue an invitation to the Owner to conduct a feasibility study for this Statement of Interest to identify and study possible solutions and, through a collaborative process with the MSBA, reach a mutually agreed upon solution. The MSBA has not approved a Project and the results of this feasibility study may or may not result in a Project approved by the MSBA.

It is anticipated that the feasibility study will review the problems identified in the Statement of Interest at the Diman Regional Vocational Technical High School, Attachment A.

The Feasibility Study shall include a study of all alternatives and contain all information required by 963 CMR 2.10(8) and any other applicable rules, policies, guidelines and directives of the Authority, including, but not limited to, a final design program, space summary, budget statement for educational objectives, and a proposed total project budget. The Schematic Design shall include, but not limited to, a site development plan, environmental assessment, geotechnical assessment, geotechnical analysis, code analysis, utility analysis, schematic building floor plans, schematic exterior building elevations, narrative building systems descriptions, NE-CHPS or LEED-S scorecard, outline specifications, cost estimates, project schedule and proposed total project budget.

The intended outcomes of the study will determine whether Greater Fall River needs a remodel, addition, or rebuild. Currently, assistance is needed to resolve issues dealing with overcrowding in classrooms and technical areas, hallway and evacuation concerns, and safety concerns in common areas such as the student cafeteria.

Of utmost importance, a functional, practical design solution that offers the best value to Diman Regional will be expected and stressed throughout all phases of design as well as during construction.

Diman Regional first opened its doors to students in 1968. The original school had 12 Chapter 74 programs and 6 post-graduate programs. In the past 49 years, the building has been adapted in many areas based upon the changing vocational market needs. Such changes have resulted in the current configuration of eighteen Chapter 74 programs and three post-secondary programs (LPN, Electricity and Plumbing). The current high school is made up of a single structure. Sections of the building consist of two stories while other parts are single story. In the 1980's, a wing was added to the original footprint of the school. This allowed for the expansion of several technical programs and theory classrooms.

Of the eighteen vocational programs at Diman Regional, only Machine Tool Technology has sufficient space to meet DESE's minimum recommended requirements for the number of students in the program. Of the remaining seventeen programs, square foot shortages range from 300 to 5,000 square feet.

To emphasize this point, please see the examples below:

- The Heating, Air-conditioning, Ventilation, and Refrigeration program (HVACR) has approximately 75 square feet per student, which is 125 square feet short of the per student recommendation by DESE.
- Health Assisting has approximately 31 square feet per student, which is 94 square feet short of the per student recommendation by DESE. Continuity of instruction is interrupted by physical limitations inherent in using classrooms as shops.
- Dental Assisting affords 84 square feet per student, which is 41 square feet short of the per student recommendation by DESE.
- Business Technology allows for 43 square feet per student, which is 67 square feet short of the per student recommendation by DESE.

The Health Assisting, Dental Assisting and Business Technology programs are critically inadequate in meeting the demand of students for these programs.

Numerous academic STEM/STEAM course offerings are not available to Diman Regional students due to the physical limitations of the plant. For example, the District's science programs cannot provide the laboratory research or clinical learning milieu to prepare students for post-secondary success. This is particularly true relative to laboratory curriculum that is severely limited to safety, space and practical limitations. In addition, the District has no ability to offer students courses in music or art as a result of the limits inherent in our building. Further, the ability to meet our students' social and emotional learning ("SEL") needs continues to be problematic as spaces for counseling, social group activities and other aspects of a strong SEL program is being denied our students.

Potential respondents are asked to review the Owner's SOI in the attachments section of this document.

The initial building was 275,000 square feet and served approximately 750 male high school students. Female enrollment at Diman Regional commenced in 1973.

The following grade configurations will be evaluated as a part of the feasibility study for the Diman Regional Vocational Technical High School in accordance with mutually agreed upon MSBA Study Enrollment Certification:

Enrollment for Grades 9-12 as Currently Configured	Enrollment for Grades 9-12 with Proposed Expansion of Chapter 74 Programming	Enrollment for Grades 9- 12 with Expansion of Chapter 74 Programming
1,400 students	1,645 students	Between 1,400 - 1,645 students

Sixteen thousand (16,000) square feet of additional classroom and shop space was added in 1989. A separate 4,000 square foot building serving the Practical Nursing Program was added in 2009 allowing for additional instructional space for grades 9-12 in the main building. The girls' locker room was expanded in 2013 to accommodate the growing female athlete population. The building currently consists of approximately 293,000 square feet, and is need of renovation consisting of ADA compliance, full roof replacement, relocation/redesign of current shop areas, updating of science lab areas and fire alarm/suppression systems as well as many other infrastructure systems that have outlived their intended useful life. To this end, many technical and academic programs cannot provide students with equitable educational opportunities to counterparts in more modern facilities. Several of the technical areas need additional space to offer students new competencies. Over the last few decades, Chapter 74 programs have updated frameworks, curricula scopes, and teaching sequences. With the limited space available to the current school, faculty members do not have adequate space to conduct appropriate project-based learning opportunities for students.

Additionally, academic frameworks have also been updated in the areas of mathematics and ELA. Antiquated classrooms and inadequate space take away from potential cooperative learning and research opportunities for faculty and students alike.

The following conditions adversely impact the ability of the District to renovate, build, or deliver the District's desired educational program/curriculum. It is the intent as part of the feasibility study that existing conditions will be evaluated by the selected designer:

- The current building is plagued by asbestos-containing building materials; i.e. spray-on asbestos, tiles and mastic, wall and ceiling coverings, etc.;
- The lack of verticality access for students with disabilities creates a safety concern in the event of a multi-hazard event;
- The current structure of the building, particularly in the STEM shop and laboratory areas, are not suitable for industry standard instruction and has become functionally obsolescent and creates a distinct educational disadvantage for students of the district;
- The inability to address an obsolescent heating and air handling system creates a bifurcated temperature milieu that result in sub-standard learning environments for students and teachers alike. Additionally, air quality suffers as a result of poor mechanical air exchange leading to increased illness for building inhabitants.

The ability to be "green" is non-existent at Diman Regional due to the inability to place solar panels on the compromised roof structure that will not support additional weight. Additionally, the District currently uses No. 4 oil that is burned in 1967 oil-fired boilers, resulting in a dirty footprint and inefficient heating source that results in exorbitant energy costs.

Furthermore, the building does not have adequate storage to support the programs and requires pods or out of building storage that is costly and hazardous during inclement weather. Security issues continue to be a major concern for the administration. Most shops do not have air conditioning and require ventilation through open doors and windows. Additionally, programs that serve the public cannot be isolated from the building proper and as a result, create an unsafe environment. Also, the ability to provide community service through our vocational programs is severely limited by the physical structure of the vocational programs as they currently exist.

The brick facade of the building is in critical need of re-pointing as gaps in the mortar have resulted in moisture penetration adding to the unhealthy learning environment.

Project Objectives under consideration by the Owner include:

- a. Serve as a model for career and workforce education by achieving sustainable results using strategies that focus on providing innovative, data driven, best practices;
- b. Support a professional learning community that cultivates a less rigid management model in favor of a highly adaptable model that targets gains in student learning;
- c. Increase productivity and cost-effectiveness by providing a balance of academic environments and applied learning laboratories and technical areas that mirror the workplace and facilitate current instructional technologies;
- d. Correct faulty design flow approaches and provide students and the general public with an experience that creates and nurtures a desire for a continuing partnership with a state-of-the-art career and technical education training center;
- e. Northeast Collaborative for High Performance Schools (NE-CHPS) criteria or US Green Building Council's LEED for Schools (LEED-S) Rating System;
- f. Give students and member communities the opportunity to safely participate in physical education activities by making long-overdue improvements and additions to Diman Regional's fields (both existing and needed), track, tennis courts, parking areas, and traffic patterns.
- g. Identification of community concerns that may impact study options;
- h. Identification of specific milestone requirements and/or constraints of the District e.g. Town votes, swing space, occupancy issues;
- i. Identification of alternative sites;
- j. Life cycle costs of operating the School as it relates to future operational budgets;
- k. CM-at-Risk Delivery Method.
- l. Functional, practical and best value will be stressed throughout all phases of design as well as construction.

C. Scope of Services:

The required scope of services is set forth in the MSBA's standard Contract for Designer Services (Contract), a copy of which is attached hereto and incorporated herein by reference. If the Owner decides to proceed with the Project beyond the Schematic Design Phase and when the project delivery method is decided (Design/Bid/Build or Construction Manager at Risk), the Contract will be amended accordingly. Copies of Designer Services Contract Amendments for Design/Bid/Build and Construction Manager at Risk are also attached hereto and incorporated herein by reference. Unless specifically excluded, the Designer's Basic Services consist of the tasks described in the Contract for Designer Services as amended and this RFS including all investigative work (to the extent provided for in the Contract), feasibility study, schematic design, and, at the Owner's option, design work, preparation of construction documents, bidding period administration, construction administration, and other related work reasonably inferred in the opinion of the Owner and the Authority as being necessary to meet the project's stated scope and goals.

This RFS will be appended to and become part of the Contract for Designer Services. Any Designer selected as a result of this RFS will be required to execute the Contract for Designer Services and applicable amendment that are attached hereto.

Basic Services include, but are not limited to, verification of existing record information including building dimensions, details and general existing conditions, cost estimating, architecture, civil, sanitary, mechanical, electrical, plumbing, fire protection, structural, site planning and landscape architecture, basic environmental permitting, graphics, lighting design, acoustics, data and communication, educational consultants, any specialty consultants for sustainable design (LEED-S/NE-CHPS), laboratory, library/media center and kitchen space, code consultants, accessibility, energy evaluations, detailed cost estimates; preparation of construction documents; bidding and administering the Construction Contract Documents and other design and consulting services incidental and required to fulfill the project goals. Please refer to the Contract and amendments for a complete summary of Basic Services.

Extra and reimbursable expenses are defined in Articles 8 and 9 of the Contract in Attachment B.

D. Project Schedule:

Work under this RFS is divided into the Project Phases as listed in Article 7 of the Contract as amended and as may be augmented in this RFS. Each Project Phase will consist of one or more required submissions, and may include site visits, meetings with the Owner, Owner's Project Manager, the Authority and others, and other tasks as described.

The milestone dates listed below are estimates only. Actual dates may vary depending upon the agreed upon solution, the extent of required document revisions, the time required for regulatory approvals, and the construction contractor's performance. Such variances will not, in and of themselves, constitute a justification for an increased Fee for Basic Services.

Projected Date
03/2/2020
10/28/2020
06/16/2021
12/25/2021
11/09/2021
09/05/2022
08/15/2025
08/29/2025

E. Minimum Qualifications:

Selection will be made by the MSBA Designer Selection Panel in accordance with the Authority's Designer Selection Procedures, attached hereto as Attachment E. The Respondent must certify in its cover letter that it meets the following minimum requirements. Any respondent that fails to include such certification in its response, demonstrating that these criteria have been met, will be rejected without further consideration. To be eligible for selection, the Designer must meet <u>all</u> of the following qualifications:

- a. Be a qualified Designer within the meaning of M.G.L. Chapter 7C, Section 44, employing a Massachusetts registered Architect responsible for and being in control of the services to be provided pursuant to the Contract.
- b. The Massachusetts registered Architect responsible for and in control of the services to be provided has successfully completed the Massachusetts Certified Public Purchasing Official Program ("MCPPO") seminar "Certification for School Project Designers and Owner's Project Managers" as administered by the Office of the Inspector General of the Commonwealth of Massachusetts, and must maintain certification by completing the "Recertification for School Project Designers and Owner's Project Managers" seminar every three years thereafter. Proof of recertification or registration in the next recertification seminar for which space is available must be provided.
- c. Pursuant to M.G.L. Chapter 7C, Section 6, the Designer must agree to contract with minority and women-owned businesses as certified by the Supplier Diversity Office (SDO). The amount of participation that shall be reserved for such enterprises shall not be less than seventeen and nine tenths percent (17.9%) of the design contract price for combined minority business enterprises and women- owned business enterprises. Applicants must include a reasonable representation of both MBE and WBE firms that meets or exceeds the combined goal.

F. Selection Criteria:

In evaluating proposals, the Owner and Designer Selection Panel will consider the members of the proposed design team. Identify those member(s) of the proposed design team who will be responsible for the following categories of work: (Firm's name, individual's name and professional registration or license number, as applicable, must be listed in the application for each category of work, as well as whether the firm is SDO certified as an MBE and/or WBE).

- 1. Architecture
- 2. Educational Programming
- 3. Civil Engineering
- 4. Landscape Architecture
- 5. Structural Engineering
- 6. Fire Protection Engineering
- 7. Plumbing Engineering
- 8. HVAC Engineering
- 9. Electrical/Lighting
- 10. Data/Communications
- 11. Environmental Permitting
- 12. Geotechnical Engineering
- 13. Geoenvironmental Engineering
- 14. Hazardous Materials
- 15. Cost Estimating
- 16. Kitchen/Food Service Consultant
- 17. Acoustical Consultant
- 18. Specifications Consultant

- 19. Library/Media
- 20. Technology Consultant/Audio Visual Consultant
- 21. Industrial Hygienist
- 22. Sustainable/Green Design/Renewable Energy Consultant
- 23. Code Consultant
- 24. Accessibility Consultant
- 25. Traffic Consultant
- 26. Furniture, Fixtures and Equipment Consultant
- 27. Site Surveying
- 28. Security Consultant
- 29. Laboratory Consultant
- 30. Theatrical Consultant
- ** N.B. -

Applicants must address each category of work listed above in their application whether it is to be performed by in-house staff or by sub-consultant(s).

The members of the team for each of the categories of work listed above must be identified including the firm's name, individual's name and professional registration or license number, as applicable, as well as whether the firm is SDO certified as an MBE and/or WBE.

Failure to address <u>each</u> category may result in the elimination of the applicant from consideration on this project.

Applicants should not list any consultants other than those for the categories of work listed above.

The minority and women-owned business enterprises must be selected to perform services addressing the categories of work listed above or be assigned to tasks required under Basic Services as specifically set forth in the Contract for Designer Services as amended. Consultants other than those proposed for the categories of work listed above or required to perform Basic Services may not be used for purposes of meeting M/WBE requirements. Applicants are strongly encouraged to utilize multiple disciplines and firms to meet their MBE/WBE goals. Consultants to the prime Designer can team within their disciplines in order to meet the MBE/WBE goals but must state this relationship on the organizational chart (Section 6 of the application form).

The Owner and Designer Selection Panel will consider the following additional criteria in evaluating proposals:

- 1. Prior similar experience best illustrating current qualifications for the specific project.
- 2. Past performance of the firm, if any with regard to public, private, DOE-funded, and MSBA funded projects across the Commonwealth, with respect to:
 - a. Quality of project design.
 - b. Quality, clarity, completeness and accuracy of plans and contract documents.
 - c. Ability to meet established program requirements within allotted budget.
 - d. Ability to meet schedules including submission of design and contract documents, processing of shop drawings, contractor requisitions and change orders.
 - e. Coordination and management of consultants.
 - f. Working relationship with contractors, subcontractors, local awarding authority and MSBA staff and local officials.
- 3. Current workload and ability to undertake the contract based on the number and scope of projects for which the firm is currently under contract.
- 4. The identity and qualifications of the consultants who will work on the project.
- 5. The financial stability of the firm.
- 6. The qualifications of the personnel to be assigned to the project.
- 7. Geographical proximity of the firm to the project site or willingness of the firm to make site visits and attend local meetings as required by the client.
- 8. Additional criteria that the MSBA Designer Selection Panel considers relevant to the project.

G. Proposal Requirements:

Persons or firms interested in applying must meet the following requirements:

- a. Applicants must have an up-to-date Master File Brochure on file at the Massachusetts School Building Authority.
- b. Applications shall be on "Standard Designer Application Form for Municipalities and Public Agencies not within DSB Jurisdiction (Updated July 2016)" as developed by the Designer Selection Board of the Commonwealth of Massachusetts. Applications (one original, twenty (20) hard copies, and two (2) digital copies in PDF format on separate compact disks or flash drives) must be received on or before 1:00 PM, Wednesday, December 18, 2019. Applications must be completed using no smaller than the same font size as in the application (10 font Arial Narrow). Applications should be printed double-sided and spiral bound on the left short edge, landscape orientation, in order that the pages lie and remain flat when opened. Applications should not be provided with acetate covers.
- c. Applications must be accompanied by a concise cover letter that is a maximum of two pages in length. A copy of the cover letter should be attached to each copy of the application. The cover letter must include the certifications as noted in Section E of this RFS. (A copy of the MCPPO certification must be attached to the cover letter as well as any SDO letters.)
- d. Applicants may supplement this proposal with graphic materials and photographs that best demonstrate design capabilities of the team proposed for this project subject to the page limitations as set forth in the Standard Designer Application Form.
- e. Proposals must be received on or before 1:00 PM, Wednesday, December 18, 2019 and shall be addressed to:

Ms. Deborah Kenney Business Manager

Diman Regional Vocational Technical High School 251 Stonehaven Road Fall River, MA 02723 508-678-2891, ext. 1160

Email: <u>dkenney@dimanregional.org</u>

f. Proposals must be clearly identified by marking the package or envelope with the following:

Designer Services, Diman Regional Vocational Technical High School Project, Fall River, MA

g. All questions regarding this RFS must be received on or before 11:00 AM, Wednesday, December 11, 2019 and should be addressed exclusively in writing via email to:

Alan Minkus, AIA Director, Project Management Services

Colliers Project Leaders

67 Hunt Street

Agawam Corporate Center, Suite 119 Agawam, MA 01001

413-592-0030 ext. 171

Email: alan.minkus@colliers.com

H. Pre-Proposal Meeting:

All interested parties are "Strongly Encouraged" to attend a briefing session at Diman Regional Vocational Technical High School, 251 Stonehaven Road, Fall River, MA 02723 scheduled for Thursday, December 5, 2019 at 11:00 AM.

While "Strongly Encouraged", the briefing session is non-mandatory.

I. Withdrawal:

Applicants may withdraw an application as long as the written request to withdraw is received by the Owner prior to the time and date of the proposal opening.

J. Public Record:

All responses and information submitted in response to this RFS are subject to the Massachusetts Public Records Law, M.G.L. c. 66, § 10 and c. 4, § 7(26). Any statements in submitted responses that are inconsistent with the provisions of these statutes shall be disregarded.

K. Waiver/Cure of Minor Informalities, Errors and Omissions:

The Owner reserves the right to waive or permit cure of minor informalities, errors or omissions prior to the selection of a Respondent, and to conduct discussions with any qualified Respondents and to take any other measures with respect to this RFS in any manner necessary to serve the best interest of the Owner and its beneficiaries.

L. Rejection of Responses, Modification of RFS:

The Owner reserves the right to reject any and all responses if the Owner determines, within its own discretion, that it is in the Owner's best interests to do so. This RFS does not commit the Owner to select any Respondent, award any contract, pay any costs in preparing a response, or procure a contract for any services. The Owner also reserves the right to cancel or modify this RFS in part or in its entirety, or to change the RFS guidelines. A Respondent may not alter the RFS or its components.

ATTACHMENTS:

Attachment A: Statement of Interest

Attachment B: Contract for Designer Services - Base Contract for Design Bid Build or CM-at-Risk Project

(http://www.massschoolbuildings.org/sites/default/files/editcontentfile/Guidelines Forms/Contracts Forms/Base%20Contract%20v 02 25.pdf)

Designer Services Contract Amendment for Design/Bid/Build (http://www.massschoolbuildings.org/sites/default/files/edit-contentfile/Guidelines Forms/Contracts Forms/DBB%20v 02 25.pdf)

Designer Services Contract Amendment for CM-at-Risk

(http://www.massschoolbuildings.org/sites/default/files/editcontentfile/Guidelines Forms/Contracts Forms/CM-R%20v 02 25.pdf)

Attachment C: Standard Designer Application Form for Municipalities and Public Agencies not within DSB Jurisdiction (Updated October 2019)

(https://www.mass.gov/files/citiestownsapplication2016 1.doc)

Attachment D: Certifications: None Included

Attachment E: MSBA's Designer Selection Panel's Procedures

End of Request for Designer Services

```
THINONALLY LEFT BLANK
```

ATTACHMENT A:

STATEMENT OF INTEREST



```
THINONALLY LEFT BLANK
```

Massachusetts School Building Authority

Next Steps to Finalize Submission of your FY 2017 Statement of Interest

Thank you for submitting your FY 2017 Statement of Interest (SOI) to the MSBA electronically. **Please note, the District's submission is not yet complete**. The District is required to print and mail a hard copy of the SOI to the MSBA along with the required supporting documentation, which is described below.

Each SOI has two Certification pages that must be signed by the Superintendent, the School Committee Chair, and the Chief Executive Officer*. Please make sure that **both** certifications contained in the SOI have been signed and dated by each of the specified parties and that the hardcopy SOI is submitted to the MSBA with **original signatures**.

SIGNATURES: Each SOI has two (2) Certification pages that must be signed by the District.

In some Districts, two of the required signatures may be that of the same person. If this is the case, please have that person sign in both locations. Please do not leave any of the signature lines blank or submit photocopied signatures, as your SOI will be incomplete.

*Local chief executive officer: In a city or town with a manager form of government, the manager of the municipality; in other cities, the mayor; and in other towns, the board of selectmen unless, in a city or town, some other municipal office is designated as the chief executive office under the provisions of a local charter.

VOTES: Each SOI must be submitted with the proper vote documentation. This means that (1) the required governing bodies have voted to submit each SOI, (2) the specific vote language required by the MSBA has been used, and (3) the District has submitted a record of the vote in the format required by the MSBA.

- School Committee Vote: Submittal of all SOIs must be approved by a vote of the School Committee.
 - For documentation of the vote of the School Committee, Minutes of the School Committee meeting at which the vote was taken must be submitted with the original signature of the Committee Chairperson. The Minutes must contain the actual text of the vote taken which should be substantially the same as the MSBA's SOI vote language.
- Municipal Body Vote: SOIs that are submitted by cities and towns must be approved by a vote of the appropriate municipal body (e.g., City Council/ Aldermen/Board of Selectmen) in addition to a vote of the School Committee.
 - Regional School Districts do not need to submit a vote of the municipal body.
 - For the vote of the municipal governing body, a copy of the text of the vote, which shall be substantially the same as the MSBA's SOI vote language, must be submitted with a certification of the City/Town Clerk that the vote was taken and duly recorded, and the date of the vote must be provided.

CLOSED SCHOOLS: Districts must download the report from the "Closed School" tab, which can be found on the District Main page. Please print this report, which then must be signed by the Superintendent, the School Committee Chair, and the Chief Executive Officer. A signed report, with original signatures must be included with the District's hard copy SOI submittal. **If a District submits multiple SOIs, only one copy of the Closed School information is required.**

ADDITIONAL DOCUMENTATION FOR SOI PRIORITIES #1 AND #3: If a District selects Priority #1 and/or Priority #3, the District is required to submit additional documentation with its SOI.

- If a District selects Priority #1, Replacement or renovation of a building which is structurally unsound or otherwise in a condition seriously jeopardizing the health and safety of the school children, where no alternative exists, the MSBA requires a hard copy of the engineering or other report detailing the nature and severity of the problem and a written professional opinion of how imminent the system failure is likely to manifest itself. The District also must submit photographs of the problematic building area or system to the MSBA.
- If a District selects Priority #3, Prevention of a loss of accreditation, the MSBA requires the full accreditation report(s) and any supporting correspondence between the District and the accrediting entity.

ADDITIONAL INFORMATION: In addition to the information required with the SOI hard copy submittal, the District may also provide any reports, pictures, or other information they feel will give the MSBA a better understanding of the issues identified at a facility.

If you have any questions about the SOI process please contact Diane Sullivan at 617-720-4466 or Diane.Sullivan@massschoolbuildings.org.

Massachusetts School Building Authority

School District Greater Fall River

District Contact Thomas F Aubin TEL: (508) 678-2891

Name of School Diman Reg Voc Tech High

Submission Date 3/3/2017

SOI CERTIFICATION

To be eligible to submit a Statement of Interest (SOI), a district must certify the following:

- The district hereby acknowledges and agrees that this SOI is NOT an application for funding and that submission of this SOI in no way commits the MSBA to accept an application, approve an application, provide a grant or any other type of funding, or places any other obligation on the MSBA.
- The district hereby acknowledges that no district shall have any entitlement to funds from the MSBA, pursuant to M.G.L. c. 70B or the provisions of 963 CMR 2.00.
- The district hereby acknowledges that the provisions of 963 CMR 2.00 shall apply to the district and all projects for which the district is seeking and/or receiving funds for any portion of a municipally-owned or regionally-owned school facility from the MSBA pursuant to M.G.L. c. 70B.
- The district hereby acknowledges that this SOI is for one existing municipally-owned or regionally-owned public school facility in the district that is currently used or will be used to educate public PreK-12 students and that the facility for which the SOI is being submitted does not serve a solely early childhood or Pre-K student population.
- After the district completes and submits this SOI electronically, the district must sign the required certifications and submit one signed original hard copy of the SOI to the MSBA, with all of the required documentation described under the "Vote" tab, on or before the deadline.
- The district will schedule and hold a meeting at which the School Committee will vote, using the specific language contained in the "Vote" tab, to authorize the submission of this SOI. This is required for cities, towns, and regional school districts.
- Prior to the submission of the hard copy of the SOI, the district will schedule and hold a meeting at which the City Council/Board of Aldermen or Board of Selectmen/equivalent governing body will vote, using the specific language contained in the "Vote" tab, to authorize the submission of this SOI. This is not required for regional school districts.
- On or before the SOI deadline, the district will submit the minutes of the meeting at which the School Committee votes to authorize the Superintendent to submit this SOI. The District will use the MSBA's vote template and the vote will specifically reference the school and the priorities for which the SOI is being submitted. The minutes will be signed by the School Committee Chair. This is required for cities, towns, and regional school districts.
- The district has arranged with the City/Town Clerk to certify the vote of the City Council/Board of Aldermen or Board of Selectmen/equivalent governing body to authorize the Superintendent to submit this SOI. The district will use the MSBA's vote template and submit the full text of this vote, which will specifically reference the school and the priorities for which the SOI is being submitted, to the MSBA on or before the SOI deadline. This is not required for regional school districts.
- The district hereby acknowledges that this SOI submission will not be complete until the MSBA has received all of the required vote documentation and certification signatures in a format acceptable to the MSBA. If Priority 1 is selected, your Statement of Interest will not be considered complete unless and until you provide the required engineering (or other) report, a professional opinion regarding the problem, and photographs of the problematic area or system.

Chief Executive Officer *	School Committee Chair	Superintendent of Schools
Thomas F. Aubin	Paul Jennings	Thomas F. Aubin
Superintendent		
(signature)	(signature)	(signature)
Date	Date	Date

^{*} Local chief executive officer: In a city or town with a manager form of government, the manager of the municipality; in other cities, the mayor; and in other towns, the board of selectmen unless, in a city or town, some other municipal office is designated to the chief executive office under the provisions of a local charter. Please note, in districts where the Superintendent is also the Local Chief Executive Officer, it is required for the same person to sign the Statement of Interest Certifications twice. Please do not leave any signature lines blank.

Massachusetts School Building Authority

School District Greater Fall River

District Contact Thomas F Aubin TEL: (508) 678-2891

Name of School Diman Reg Voc Tech High

Submission Date $\frac{3}{3}$

Note

The following Priorities have been included in the Statement of Interest:

- 1. Explacement or renovation of a building which is structurally unsound or otherwise in a condition seriously jeopardizing the health and safety of school children, where no alternative exists.
- 2. Elimination of existing severe overcrowding.
- 3. Prevention of the loss of accreditation.
- 4. Prevention of severe overcrowding expected to result from increased enrollments.
- 5. Explacement, renovation or modernization of school facility systems, such as roofs, windows, boilers, heating and ventilation systems, to increase energy conservation and decrease energy related costs in a school facility.
- 6. [€] Short term enrollment growth.
- 7. Be Replacement of or addition to obsolete buildings in order to provide for a full range of programs consistent with state and approved local requirements.
- 8. E Transition from court-ordered and approved racial balance school districts to walk-to, so-called, or other school districts.

SOI Vote Requirement

B I acknowledge that I have reviewed the MSBA's vote requirements for submitting an SOI which are set forth in the Vote Tab of this SOI. I understand that the MSBA requires votes from specific parties/governing bodies, in a specific format using the language provided by the MSBA. Further, I understand that the MSBA requires certified and signed vote documentation to be submitted with the SOI. I acknowledge that my SOI will not be considered complete and, therefore, will not be reviewed by the MSBA unless the required accompanying vote documentation is submitted to the satisfaction of the MSBA.

Potential Project Scope: Renovation/ Addition

Is this SOI the District Priority SOI? YES

School name of the District Priority SOI: Diman Reg Voc Tech High

Is this part of a larger facilities plan? NO

If "YES", please provide the following:

Facilities Plan Date:

Planning Firm:

Please provide an overview of the plan including as much detail as necessary to describe the plan, its goals and how the school facility that is the subject of this SOI fits into that plan:

Please provide the current student to teacher ratios at the school facility that is the subject of this SOI: 11 students per teacher

Please provide the originally planned student to teacher ratios at the school facility that is the subject of this SOI: 11 students per teacher

Does the District have a Master Educational Plan that includes facility goals for this building and all school buildings in District? NO

Does the District have related report(s)/document(s) that detail its facilities, student configurations at each facility, and District operational budget information, both current and proposed?

NO

If "NO", please note that:

If, based on the SOI review process, a facility rises to the level of need and urgency and is invited into the Eligibility Period, the District will need to provide to the MSBA a detailed Educational Plan for not only that facility, but all facilities in the District in order to move forward in the MSBA's school building construction process.

Is there overcrowding at the school facility? YES

If "YES", please describe in detail, including specific examples of the overcrowding.

The Greater Fall River Vocational School District has utilized every available space within the building for instructional and support services. As a result of this space limitation, Diman cannot meet the market demand for its educational services. Consequently, 40% of students who apply on average are turned away each year.

Of the sixteen vocational programs at Diman, only Machine Tool Technology has sufficient space to meet DESE's minimum recommended requirements for the number of students in the program. Of the remaining fifteen programs, square foot shortages range from 300 to 5,000 square feet.

To emphasize this point, please see examples below:

- The Heating, Air-conditioning, Ventilation, and Refrigeration program (HVACR) has approximately 75 square feet per student, which is 125 square feet short of the per student recommendation by DESE.
- Health Assisting has approximately 31 square feet per student, which is 94 square feet short of the per student recommendation by DESE. Continuity of instruction is interrupted by physical limitations inherent in using classrooms as shops.
- Dental Assisting affords 84 square feet per student, which is 41 square feet short of the per student recommendation by DESE.
- Business Technology allows for 43 square feet per student, which is 67 square feet short of the per student recommendation by DESE.
- * Health Assisting, Dental Assisting and Business Technology programs are critically short in meeting the demand of students for these programs. This is particularly troubling as the District is unable to provide equity of educational opportunity for its female population. Only 19% of the school's female population will have enrollment opportunities for these traditional female orientated programs. Efforts to mitigate this gender-inequality through additional vocational programs have been stymied by the physical limitations of our existing plant.

Additionally, numerous academic STEM/STEAM course offerings are not available to Diman students due to the physical limitations of the plant. For example, the District's science programs cannot provide the laboratory research or clinical learning milieu to prepare students for post-secondary success. This is particularly true relative to laboratory curriculum that is severely limited to safety, spacial or practical limitations. Additionally, the District has no ability to offer students courses in music or art as a result of the limits inherent in our building. Also, the ability to meet our students' social and emotional learning (SEL) needs continues to be problematic as spaces for counseling, social group activities and other aspects of a strong SEL program is being denied our students.

Furthermore, the structure of the cafeteria creates critical mass both in housing and feeding our students. For example, the school currently conducts five lunch periods due to chronic overcrowding. Despite this, overcrowding remains both a practical and safety concern for the school. Equally concerning is the District's inability to further expand the culinary offerings to our students that reflect an understanding of the nexus between nutrition and learning. Specifically, lack of food preparation area, refrigeration, and storage severely limit the variety of food offerings that reflect a nutritious and appealing diet for students.

In addition, the current library space at Diman does not meet the educational needs of our students from both a research and instructional perspective, as space limitations do not afford the dual activity of classroom instruction and research. This is especially concerning for students who must compete in a global society that requires myriad research skills coupled with familiarity of varied instructional settings that reflect current trends for college and career readiness.

The gymnasium and athletic fields cannot meet the needs of our athletic teams during any of the three-sport seasons. For example, the school is currently using the football/soccer field for approximately nine teams during practices in the fall. Additionally, the gymnasium is over-subscribed during the winter sports season as six teams are scheduled to practice each day after school. This leads to extended practice times that run well past 7:00 p.m. Adding to this problem is the fact that there is no area for students to safely congregate while waiting to practice. The consequence of this situation is that students' time to study is hindered. Furthermore, safety concerns exist as a result of supervision limitations for teachers, administration and parents alike. The District does not have the ability to meet the ongoing requests of students and parents alike to add new sports programs to the current offerings at Diman.

Has the district had any recent teacher layoffs or reductions?

NO

If "YES", how many teaching positions were affected? 0

At which schools in the district?

Please describe the types of teacher positions that were eliminated (e.g., art, math, science, physical education, etc.).

Has the district had any recent staff layoffs or reductions?

NO

If "YES", how many staff positions were affected? 0

At which schools in the district?

Please describe the types of staff positions that were eliminated (e.g., guidance, administrative, maintenance, etc.).

Please provide a description of the program modifications as a consequence of these teacher and/or staff reductions, including the impact on district class sizes and curriculum.

Does Not Apply

Please provide a detailed description of your most recent budget approval process including a description of any budget reductions and the impact of those reductions on the district's school facilities, class sizes, and educational program.

The District's budgetary process is dictated by the Greater Fall River Vocational School District's regional agreement that was drafted in 1962. Throughout Diman's history, the sending communities have never exceeded the minimum spending allocations as dictated by the Department of Elementary and Secondary Education (DESE). Consequently, a history of austere measures were required to accrue liquidity for previous capital projects at Diman. Over the past five years, revenue streams from Chapters 70, 71, 74 and other funding sources have negatively impacted the ability of the District to financially meet the ever-growing industry and educational requirements for Diman students. As the ability of the District to adequately institutionalize the cost of both expansion and modernization of our educational facility is declining due to the aforementioned cuts in funding sources, options for the District continue to dwindle. Due to the fact that enrollment demands have increased over the past several years, monies that were originally earmarked for capital projects have been redistributed towards instructional purposes. This financial reality is creating an educational climate of "haves" and "have-nots", and places Diman and its' students in the later category.

General Description

BRIEF BUILDING HISTORY: Please provide a detailed description of when the original building was built, and the date(s) and project scopes(s) of any additions and renovations (maximum of 5000 characters).

Diman Regional Vocational Technical High School first opened its doors to students in 1968. The original school had 12 Chapter 74 programs and six post-graduate programs. In the past 49 years, changes have resulted in the current configuration of sixteen Chapter 74 programs and a post-secondary Practical Nursing Program. The school also has a non-Chapter 74 grade 9 vocational program that provides remedial technical competencies for students.

The initial building was 275,000 square feet and served approximately 750 male high school students. Female enrollment at Diman commenced in 1973. Sixteen thousand (16,000) square feet of additional classroom and shop space was added in 1989. A separate 4,000 square foot building serving the Practical Nursing Program was added in 2009 allowing for additional instructional space for grades 9-12 in the main building.

The girls' locker room was expanded in 2013 to accommodate the growing female athlete population and attempt to provide gender equity in sports offerings.

TOTAL BUILDING SQUARE FOOTAGE: Please provide the original building square footage PLUS the square footage of any additions.

293000

SITE DESCRIPTION: Please provide a detailed description of the current site and any known existing conditions that would impact a potential project at the site. Please note whether there are any other buildings, public or private, that share this current site with the school facility. What is the use(s) of this building(s)? (maximum of 5000 characters).

Diman Regional Vocational Technical High School is located in the east end of Fall River, Massachusetts and occupies 26 acres of land. The original footprint of the building was approximately 275,000 square feet, with an addition that was completed in 1989 that added 16,000 square feet to the original building.

The existence of ledge has limited the financial ability of the district to institutionalize the cost of both renovations and additions that would alleviate the chronic and acute overcrowding issue that is impacting the delivery of instruction to our students.

The following conditions adversely impact the ability of the district to renovate or build:

- *The current building is plagued by asbestos-containing building materials; i.e. spray-on asbestos, tiles and mastic, wall and ceiling coverings, etc;
- *The lack of verticality access for students with disabilities creates a safety concern in the event of a multi-hazard event;

 *The current structure of the building, particularly in the STEM shop and laboratory areas, are not suitable for industrystandard instruction and has become functionally obsolescent and creates a distinct educational disadvantage for students

of the district;

*The inability to address an obsolescent heating and air handling system creates a bifurcated temperature milieu that result

*The inability to address an obsolescent heating and air handling system creates a bifurcated temperature milieu that result in sub-standard learning environments for students and teachers alike. Additionally, air quality suffers as a result of poor mechanical air exchange leading to increased illness for building inhabitants.

The ability to be "green" is non-existent at Diman due to the inability to place solar panels on the compromised roof

structure that will not support additional weight. Additionally, the district currently uses No. 4 oil that is burned in 1967 oil-fired boilers resulting in a dirty footprint and inefficient heating source that results in exorbitant energy costs.

Furthermore, the building does not have adequate storage to support the programs and requires pods or out of building storage that is costly and hazardous during inclement weather.

Security issues continue to be a major concern for the administration. Most shops do not have air conditioning and require ventilation through open doors and windows. Additionally, programs that serve the public cannot be isolated from the building proper and as a result, create an environment that would provide easy access to intruders intent on doing harm. Also, the ability to provide community service through our vocational programs is severely limited by the physical structure of the vocational programs as they currently exist.

The brick facade of the building is in critical need of re-pointing as gaps in the mortar have resulted in moisture penetration adding to the unhealthy learning environment.

ADDRESS OF FACILITY: Please type address, including number, street name and city/town, if available, or describe the location of the site. (Maximum of 300 characters)

Diman Regional Vocational Technical High School 251 Stonehaven Road Fall River, MA 02723

BUILDING ENVELOPE: Please provide a detailed description of the building envelope, types of construction materials used, and any known problems or existing conditions (maximum of 5000 characters).

The building is over 50 years old. The construction of the building was a combination of window-wall panels (kalwall), double-hung window units, and precast concrete panels along with standard brick and mortar construction enveloping a steel superstructure. As a result of the period in which the building was constructed, various types of asbestos were used throughout.

Has there been a Major Repair or Replacement of the EXTERIOR WALLS? YES

Year of Last Major Repair or Replacement:(YYYY) 1999

Description of Last Major Repair or Replacement:

The block wall system in D and E shop wings were removed and replaced due to a block system failure.

The C-wing north roof wall was rebuilt due to shifting and failure.

Roof Section A

Is the District seeking replacement of the Roof Section? NO

Area of Section (square feet) 150274

Type of ROOF (e.g., PVC, EPDM, Shingle, Slate, Tar & Gravel, Other (please describe)

Tar and gravel

Age of Section (number of years since the Roof was installed or replaced) 18

Description of repairs, if applicable, in the last three years. Include year of repair:

None

Roof Section B

Is the District seeking replacement of the Roof Section? NO

Area of Section (square feet) 18400

Type of ROOF (e.g., PVC, EPDM, Shingle, Slate, Tar & Gravel, Other (please describe)

Built up

Age of Section (number of years since the Roof was installed or replaced) 47

Description of repairs, if applicable, in the last three years. Include year of repair:

None

Roof Section C

Is the District seeking replacement of the Roof Section?

Area of Section (square feet)

Type of ROOF (e.g., PVC, EPDM, Shingle, Slate, Tar & Gravel, Other (please describe)

Age of Section (number of years since the Roof was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Roof Section D

Is the District seeking replacement of the Roof Section?

Area of Section (square feet)

Type of ROOF (e.g., PVC, EPDM, Shingle, Slate, Tar & Gravel, Other (please describe)

Age of Section (number of years since the Roof was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Roof Section E

Is the District seeking replacement of the Roof Section?

Area of Section (square feet)

Type of ROOF (e.g., PVC, EPDM, Shingle, Slate, Tar & Gravel, Other (please describe)

Age of Section (number of years since the Roof was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Roof Section F

Is the District seeking replacement of the Roof Section?

Area of Section (square feet)

Type of ROOF (e.g., PVC, EPDM, Shingle, Slate, Tar & Gravel, Other (please describe)

Age of Section (number of years since the Roof was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Roof Section G

Is the District seeking replacement of the Roof Section?

Area of Section (square feet)

Type of ROOF (e.g., PVC, EPDM, Shingle, Slate, Tar & Gravel, Other (please describe)

Age of Section (number of years since the Roof was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Roof Section H

Is the District seeking replacement of the Roof Section?

Area of Section (square feet)

Type of ROOF (e.g., PVC, EPDM, Shingle, Slate, Tar & Gravel, Other (please describe)

Age of Section (number of years since the Roof was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Roof Section 1

Is the District seeking replacement of the Roof Section?

Area of Section (square feet)

Type of ROOF (e.g., PVC, EPDM, Shingle, Slate, Tar & Gravel, Other (please describe)

Age of Section (number of years since the Roof was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Roof Section J

Is the District seeking replacement of the Roof Section?

Area of Section (square feet)

Type of ROOF (e.g., PVC, EPDM, Shingle, Slate, Tar & Gravel, Other (please describe)

Age of Section (number of years since the Roof was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Window Section A

Is the District seeking replacement of the Windows Section? NO

Windows in Section (count) 222

Type of WINDOWS (e.g., Single Pane, Double Pane, Other (please describe))

Double hung, double pane classroom windows

Age of Section (number of years since the Windows were installed or replaced) 16

Description of repairs, if applicable, in the last three years. Include year of repair:

None

Window Section B

Is the District seeking replacement of the Windows Section? NO

Windows in Section (count) 49

Type of WINDOWS (e.g., Single Pane, Double Pane, Other (please describe))

Double hung/hopper

Age of Section (number of years since the Windows were installed or replaced) 12

Description of repairs, if applicable, in the last three years. Include year of repair:

None

Window Section C

Is the District seeking replacement of the Windows Section? NO

Windows in Section (count) 95

Type of WINDOWS (e.g., Single Pane, Double Pane, Other (please describe))

Kalwall window panels

Age of Section (number of years since the Windows were installed or replaced) 12

Description of repairs, if applicable, in the last three years. Include year of repair:

None

Window Section D

Is the District seeking replacement of the Windows Section?

Windows in Section (count)

Type of WINDOWS (e.g., Single Pane, Double Pane, Other (please describe))

Age of Section (number of years since the Windows were installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Window Section E

Is the District seeking replacement of the Windows Section?

Windows in Section (count)

Type of WINDOWS (e.g., Single Pane, Double Pane, Other (please describe))

Age of Section (number of years since the Windows were installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Window Section F

Is the District seeking replacement of the Windows Section?

Windows in Section (count)

Type of WINDOWS (e.g., Single Pane, Double Pane, Other (please describe))

Age of Section (number of years since the Windows were installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Window Section G

Is the District seeking replacement of the Windows Section?

Windows in Section (count)

Type of WINDOWS (e.g., Single Pane, Double Pane, Other (please describe))

Age of Section (number of years since the Windows were installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Window Section H

Is the District seeking replacement of the Windows Section?

Windows in Section (count)

Type of WINDOWS (e.g., Single Pane, Double Pane, Other (please describe))

Age of Section (number of years since the Windows were installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Window Section I

Is the District seeking replacement of the Windows Section?

Windows in Section (count)

Type of WINDOWS (e.g., Single Pane, Double Pane, Other (please describe))

Age of Section (number of years since the Windows were installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Window Section J

Is the District seeking replacement of the Windows Section?

Windows in Section (count)

Type of WINDOWS (e.g., Single Pane, Double Pane, Other (please describe))

Age of Section (number of years since the Windows were installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

MECHANICAL and ELECTRICAL SYSTEMS: Please provide a detailed description of the current mechanical and electrical systems and any known problems or existing conditions (maximum of 5000 characters).

Electrical service size is 4000A 480V-Y (4wire)

Transformer evaluation is needed to ensure it won't hinder additional equipment upgrades and/or replacement(s).

277V Lighting circuit 120/240/480V Service panels

All original panels and main switch gear is manufactured by Square "D".

All original panels need replacement.

Genset replaced in 2002 is a Cummins diesel-powered unit with self contained "base" fuel tank.

175Kw max electrical output 120/240v

Switch gears and panels to support this back-up power grid throughout the complex are original equipment and in need of replacement and or upgrade.

3 Boilers are manufactured by Cleaver Brooks

1967 installation date

Fired with #4-1% heavy fuel oil

20000gl underground fuel tank with overfill/tank monitor Veeter-Root

All main line valving in need of replacement and upgrade

All distribution lines (piping) in need of replacement and upgrade

*lack of water treatment has contributed to this systems failure.

Hot water fancoil (powered) heating AHU's in all classrooms, shops and common areas in our original building are 40+

years old, obsolete and parts are no longer available for these units.

This includes cabinet exhaust units in class areas.

Roof top exhaust units are maintained but CFM and air exchange are not adequate for our occupancy.

HW Boiler:

OA Smith DuraMax Gas Fired (indirect) Hot water boiler

Model DW720 Mfg Date 2003

720000 BTU

Installed 2003

This boiler as well as our building heat boilers are coupled to a tube & shell type hot water heater.

Adams Tank Co.

MFG 1967

2600 gal storage capacity

Retubed in 2001

This tank shell is original equipment.

Phone System

Omega Phone ADIX 200

Model 1X200 PWS

MFG by Iwatsu 1997

Installed 1997

Digital and Analog phone management system

Cabling and punch block systems in need of upgrading due to original cabling and configurations. This infrastructure does not properly support upgraded technology

Building Communication (PA) System

Simplex 5100

Installed 1997

Cabling throughout and all patch panels are original equipment and not designed to support updated technology.

This system is integrated with Simplex Time Control System,

Model 6400

Installed 1998

This system is integrated with the bell/tone system

Distribution cabling is inadequate to properly support this interface and replacement is needed.

Fire alarm system is a Simplex product.

Model 4100

Installed in 1998

It is currently monitoring 15 Zones

Equipment includes a sub-panel and graphic Annunciator.

Heat Detectors, Pull stations, Strobe and horns are throughout the facility

Sprinkler, wet systems are in D-wing south addition only, and in equipment (paint booths) as required.

Magnetic fire door releases are found only in the A-wing and D-wing south hall doors.

All fire safety equipment is in need of updating to meet current new building safety regulations.

Boiler Section 1

Is the District seeking replacement of the Boiler? NO Is there more than one boiler room in the School? NO

What percentage of the School is heated by the Boiler? 100

Type of heating fuel (e.g., Heating Oil, Natural Gas, Propane, Other)

#4 Heating Oil

Age of Boiler (number of years since the Boiler was installed or replaced) 48

Description of repairs, if applicable, in the last three years. Include year of repair:

Routine maintenance.

Boiler Section 2

Is the District seeking replacement of the Boiler?

Is there more than one boiler room in the School?

What percentage of the School is heated by the Boiler?

Type of heating fuel (e.g., Heating Oil, Natural Gas, Propane, Other)

Age of Boiler (number of years since the Boiler was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Boiler Section 3

Is the District seeking replacement of the Boiler?

Is there more than one boiler room in the School?

What percentage of the School is heated by the Boiler?

Type of heating fuel (e.g., Heating Oil, Natural Gas, Propane, Other)

Age of Boiler (number of years since the Boiler was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Boiler Section 4

Is the District seeking replacement of the Boiler?

Is there more than one boiler room in the School?

What percentage of the School is heated by the Boiler?

Type of heating fuel (e.g., Heating Oil, Natural Gas, Propane, Other)

Age of Boiler (number of years since the Boiler was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Boiler Section 5

Is the District seeking replacement of the Boiler?

Is there more than one boiler room in the School?

What percentage of the School is heated by the Boiler?

Type of heating fuel (e.g., Heating Oil, Natural Gas, Propane, Other)

Age of Boiler (number of years since the Boiler was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Boiler Section 6

Is the District seeking replacement of the Boiler?

Is there more than one boiler room in the School?

What percentage of the School is heated by the Boiler?

Type of heating fuel (e.g., Heating Oil, Natural Gas, Propane, Other)

Age of Boiler (number of years since the Boiler was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Boiler Section 7

Is the District seeking replacement of the Boiler?

Is there more than one boiler room in the School?

What percentage of the School is heated by the Boiler?

Type of heating fuel (e.g., Heating Oil, Natural Gas, Propane, Other)

Age of Boiler (number of years since the Boiler was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Boiler Section 8

Is the District seeking replacement of the Boiler?

Is there more than one boiler room in the School?

What percentage of the School is heated by the Boiler?

Type of heating fuel (e.g., Heating Oil, Natural Gas, Propane, Other)

Age of Boiler (number of years since the Boiler was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Boiler Section 9

Is the District seeking replacement of the Boiler?

Is there more than one boiler room in the School?

What percentage of the School is heated by the Boiler?

Type of heating fuel (e.g., Heating Oil, Natural Gas, Propane, Other)

Age of Boiler (number of years since the Boiler was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Boiler Section 10

Is the District seeking replacement of the Boiler?

Is there more than one boiler room in the School?

What percentage of the School is heated by the Boiler?

Type of heating fuel (e.g., Heating Oil, Natural Gas, Propane, Other)

Age of Boiler (number of years since the Boiler was installed or replaced)

Description of repairs, if applicable, in the last three years. Include year of repair:

Has there been a Major Repair or Replacement of the HVAC SYSTEM? NO

Year of Last Major Repair or Replacement: (YYYY) 2006

Description of Last Major Repair or Replacement:

The air handling unit in Metal Fabrication was replaced 2006.

The remainder of all heating mechanical equipment is original and in need of upgrade for energy efficiency and air quality.

All distribution piping, valving, and controls for the forced hot water system is also nearing the end of their life cycle. Air conditioning units have been installed throughout the D-wing, C-wing second floor, B-wing second floor, the two computer labs, and in the auditorium. There are building construction and asbestos constraints to providing air conditioning for the first floor.

Has there been a Major Repair or Replacement of the ELECTRICAL SERVICES AND DISTRIBUTION SYSTEM? YES

Year of Last Major Repair or Replacement:(YYYY) 2012

Description of Last Major Repair or Replacement:

The Culinary Arts service panel was relocated and updated in 2012. The main disconnect of D-wing failed and was replaced in 2010. Main switch gear was partially updated 2006 (main building power disconnect only). All other main panels are original vintage and in need of replacement for safety and energy efficiency. School emergency generator was replaced in 2002 (175,000 kvw).

BUILDING INTERIOR: Please provide a detailed description of the current building interior including a description of the flooring systems, finishes, ceilings, lighting, etc. (maximum of 5000 characters).

Plaster wall systems and block wall systems throughout the building. Some of the walls are painted and others are tiled. The floors are a combination of Terrazzo, VCT tile, ceramic tile, and cement material flooring. Ceilings are plaster painted and suspended acoustical panels throughout. Florescent lighting fixtures and ballasts have all been either upgraded or

replaced for energy efficiency in 2008 and on-going as rebates become available.

PROGRAMS and OPERATIONS: Please provide a detailed description of the current programs offered and grades served, and indicate whether there are program components that cannot be offered due to facility constraints, operational constraints, etc. (maximum of 5000 characters).

Diman Regional Vocational Technical High School has sixteen programs including Exploratory. Each of these vocational shops are approved, fully accredited Chapter 74 programs. Those are:

Automotive Collision Repair & Refinishing (470603)

Automotive Technology (470604)

Building and Property Maintenance (460401)

Business Technology (520407)

Carpentry (460201)

Culinary Arts (120500)

Dental Assisting (510601)

Drafting (151301)

Electricity (460302)

Electronics (150303)

Exploratory (990100)

Graphic Communications (100301)

Health Assisting (510000)

Heating - Ventilation - Air Conditioning - Refrigeration (470201)

Machine Tool Technology (480501)

Metal Fabrication & Joining Technologies (480599)

Plumbing (460503)

Diman is unable to offer Chapter 74 programs in Biotechnology, Hospitality Management, Information & Support Services, Legal & Protective Services, or Programming & Web Development due to space limitations. These are all considered emerging markets in Southeastern Massachusetts. Diman is also unable to expand the Medical/Health Assisting program to meet student interest and community needs due to space limitations.

The Painting & Decorating Program was closed in 2003. The space was then repurposed for the HVAC shop.

Diman also has a General Shop Program that provides a vocational/technical education to special needs students who have difficulty participating in the larger shops. Unlike the full-day week-about shop programs, General Shop is 180 half-day program in a small group highly structured environment. Although not an approved Chapter 74 program, General Shop teaches students the hands-on vocational, safety, and employability skills necessary for full inclusion in the approved programs.

CORE EDUCATIONAL SPACES: Please provide a detailed description of the Core Educational Spaces within the facility, a description of the number and sizes (in square feet) of classrooms, a description of science rooms/labs including ages and most recent updates, a description of the cafeteria, gym and/or auditorium and a description of the media center/library (maximum of 5000 characters).

The Core Educational Spaces are comprised of 32 academic classrooms that are between approximately 725 and 750 square feet. Seven classrooms were initially created as shop related classrooms of between approximately 575 and 600 square feet for up to fifteen students and have been converted to academic classrooms for up to 25 students. There are eight special education classrooms, four Reading Language Arts classrooms, and an English Language Learners classroom that have all been converted from other spaces. The special education classrooms are between 280 and 450 square feet, the reading language arts classrooms are between 406 and 580 square feet. The ELL classroom is approximately 300

square feet and needs to be relocated to a classroom space for FY17 to address a CESE CPR citing. All academic classrooms have teacher computer stations with printers, SmartBoards, and most have multiple student computer stations. Teachers also have access to one class-size computer labs.

In addition to the 51 academic classrooms listed above, there are four science laboratories. Only one (biology) of the four science laboratories is original to the building; the other three (biology, physics, and chemistry) were created from other instructional areas.

The Physics laboratory was converted from a machine tool technology related classroom into a science lab in 1986 and has 928 square feet. It has 12 stations, each with a High Voltage Power Supply, Oscilloscope, and Digital Multimeter. It also has six Frequency Generators, six Analog Multimeters, and seven Hot Plates. It should be noted that the Machine Tool Technology related classes were moved into a converted storage room within their shop.

The Chemistry laboratory was converted from an Auto Collision related classroom into a science lab in 2004 and has 957 square feet. It is well-stocked and has natural gas, water, and compressed air at each of 12 stations. It has ample glassware and triple beam balances, and two digital balances. It should be noted that the Auto Collision related classes were moved into a converted storage room.

There are two biology laboratories, one is 1,761 square feet and is the only science lab original to the building when constructed and opened in 1968. The other biology laboratory was created from an academic classroom in 2004 and has 986 square feet. Both have natural gas, water, electricity, and compressed air at each of 12 lab stations. All labs have compound microscopes for each student. Additional equipment includes a minicentrifuge, electrophoresis apparatus (6gel), three magnetic stirrer/hotplates, one spectrophotometer, two small waterbaths and one incubator.

Even with the additional laboratory classrooms, approximately 30 lab classes are taught in standard classrooms with limited equipment and laboratory resources.

The library was structured for a maximum 750 student body. Recognizing the inadequacy of the space, it was renovated and enlarged by absorbing an adjacent classroom in 2002 and is now 2600 square feet. The current library has thirteen tables which each could accommodate six students each. The second floor computer carts (25 laptops each) are housed in the library and available for student use in that setting. The shelf capacity is 11,000 for books and other materials.

The LPN area of the library was converted into a Teacher Resource Room to provide a work area for teachers who are unable to use their assigned classrooms during their preparation periods. The LPN resource books were moved to the LPN building.

CAPACITY and UTILIZATION: Please provide a detailed description of the current capacity and utilization of the school facility. If the school is overcrowded, please describe steps taken by the administration to address capacity issues. Please also describe in detail any spaces that have been converted from their intended use to be used as classroom space (maximum of 5000 characters).

- The girls' locker and shower room was recently remodeled and expanded. What was previously the music/band room (used as a math classroom from 1980 when band was dropped from the course of study) was incorporated into the girls' locker and shower room facility. This was a NEASC recommendation in 2003 and was a potential Title IX issue.
- The Drafting program space was reconfigured to move the related classroom, reduce the size of the freshman/sophomore shop, and eliminate the storage closet. This provided sufficient space for a full-size classroom to replace the classroom that was lost in the expansion of the girls' locker and shower room.
- The Health Assisting vocational program was converted from School of Practical Nursing offices to its present configuration.
- The Dental Assisting vocational program was converted from a post-graduate classroom area to its present configuration.
- A locker room originally designed for post-graduate students has been converted to a Special Education classroom.
- A storage area for the Auto Collision shop has been converted into two classrooms, one a Special Education classroom and the other a shop-related classroom.

- A teachers' workroom has been converted to a Special Education classroom.
- Storage space in the Machine Tool Technology Shop was renovated and re-purposed into an Engineering/Technology Center
- A platform was installed outside the building next to HVAC-R for a commercial refrigeration unit.
- A storage area in Pupil Personnel Services was assimilated into Guidance/Dean's Office renovation to provide sufficient space for a shared conference area.
- A second floor office area/teacher work-room has been converted to a Special Education classroom.
- Another second floor office area/teacher work-room has been converted to a Reading Language Arts classroom.
- The Computer Labs on the first floor were originally designed for a post-secondary Electronics program which was dropped when the space was needed for the high school programs.
- The PPS office was converted from a post-secondary Computer and Data Processing program.
- A storage area for three vocational shops was converted into a vocational related classroom.
- A school book storage area was converted to a Fitness Center for student instruction and cardiovascular training.
- A storage/electrical control panel closet that was used at various time in the past as an office, an AV room, and an ELL classroom is now used as storage for the Business Technology program.
- The auditorium lighting was upgraded and the seats replaced with units that have attached writing trays so that the space could be used an alternative instructional space and testing center.
- Storage trailers/pods have been purchased and are located outside of the shops. There are fourteen such storage trailers/pods currently in use.

Unfortunately, lack of vocational program space has necessitated the limiting of enrollment to approximately 70% of the students who apply.

MAINTENANCE and CAPITAL REPAIR: Please provide a detailed description of the district's current maintenance practices, its capital repair program, and the maintenance program in place at the facility that is the subject of this SOI. Please include specific examples of capital repair projects undertaken in the past, including any override or debt exclusion votes that were necessary (maximum of 5000 characters).

Preventive maintenance programs are electronically emailed for daily casual repairs. Contracted time and materials preventive maintenance is seasonally scheduled for all refrigerating and air conditioning equipment.

Summer scheduled maintenance includes the shutdown of the boiler system and maintenance to the system, specifically repairs and maintenance to exhaust fans, heat air handling units, motor belt and filter maintenance. Any non-emergency network infrastructure upgrades, which would include telephone, computer networks, security system, and electrical needs are made during the summer vacations.

All major facility structures and systems are inspected at least annually and repaired over the summer. For example, the roof is inspected in the early spring to identify any areas of concern, which are then evaluated and prioritized. Repairs and preventive maintenance is then scheduled over the summer months to be completed before the start of the new school year.

Priority 2

Question 1: Please describe the existing conditions that constitute severe overcrowding.

In an effort to meet the growing stakeholder demand from our sending communities for technical education, the district increased enrollment from 1,120 students in 2002 to the current enrollment of 1,400 students. This has resulted in a number of infrastructural and logistical problems in both the daily operation of the organization and the plant proper. For example:

- Hallway congestion during passing periods is problematic throughout the school day and has required adjustments in the schedule to mitigate the overcrowding, resulting in less instructional time for students;
- Mechanical and lavatory facilities are continually overtaxed as the original infrastructure is serving a larger capacity than were originally designed. As a consequence of this student overcrowding, excessive use of both human and financial capital is required to keep the facility functioning;
- The location of student drop-offs via automobile creates a safety hazard for students as driveway design limitations require students to exit vehicles into oncoming traffic;
- The current enrollment at Diman, coupled with space limitations, prohibits the school from conducting en masse meetings unless they are conducted at an outdoor location. Consequently, this impacts the ability of the school to communicate messages in a singular voice, possibly leading to an unevenness in the prescribed message. Additionally, this limitation also prohibits indoor all-school activities or celebrations that foster esprit de corps within and among the school community. Furthermore, as this limitation impacts all school events, stakeholder inclusivity cannot happen and as a result, effectively eliminates opportunities to expand community engagement;
- Due to a decrease in the District's leveling along with the moral obligation to offer an equitable educational experience, Diman would like to institute special education inclusion in the areas of mathematics and English. Due to the overcrowding of Diman's physical plant, this option is currently unavailable;
- The school is struggling to meet one of its 2017 District Improvement Goals to improve the social emotional learning opportunities for its students as space limitations prohibit group and independent counseling opportunities that have proven effective to reduce student anxiety that allows for barrier free learning. For example, the school has difficulty meeting the needs of students who require additional space to obtain services that include counseling or consultants as part of their 504/IEP Plan; and,
- Parking for stakeholders, especially parents is generally unavailable. It requires parents/guardians to walk an unreasonable distance to pick up their children or meet with instructional staff, causing them at times to park illegally or in fire lanes.

Priority 2

Question 2: Please describe the measures the School District has taken to mitigate the problem(s) described above.

The only tangible response that the District has relative to reducing the overcrowding problem is to reduce enrollment. This is politically untenable as the District realizes a minimum of forty percent surplus in admission applications each year. Efforts to institutionalize the cost of renovations/additions have been negated as a result of annual reductions in revenues streams at the state and national level. Additionally, the four sending communities have never, in the history of Diman Regional, agreed to pay over the minimum contribution as required by law. Furthermore, the creation of a campus off-site has been hampered by the increase in transportation costs coupled with reductions in Chapter 71 funding. Also, anecdotal polling of teachers and parents alike, indicates that support for a building separate from Diman proper has not gained momentum. As such, the possibility of leasing or purchasing off-site is financially and politically untenable.

Question 3: Please provide a detailed explanation of the impact of the problem described in this priority on your district's educational program. Please include specific examples of how the problem prevents the district from delivering the educational program it is required to deliver and how students and/or teachers are directly affected by the problem identified.

The science labs, as currently constructed at Diman, are not capable of providing students with the STEM education that their counterparts in neighboring communities are experiencing. This is especially true when referencing laboratory activities that are required both under the MassCore and the Next Generation Science Standards. In fact, in the school's current situation, many lab science course work is occurring in math and English classrooms.

Diman is forced to cap certain classes at 15 instead of the traditional 25 due to size of the room and inadequate resources resulting in an increased instructional and benefits cost. It also limits flexibility in the master schedule and limits the number of electives the school can offer. Diman does not have appropriate classroom space for music or art programs.

Diman's current square footage does not allow for expansion of vocational programs geared towards female occupations. Consequently, an unacceptably large percentage of Diman's female student population is denied educational equity in program opportunities, compared to their male counterparts.

Currently, due to space issues, four of Diman's sixteen vocational programs do not provide related/theory instruction within the academic schedule. Three of the four programs are geared toward female occupations, once again, creating a possible Title IX issue.

Please also provide the following:

Cafeteria Seating Capacity:	288
-----------------------------	-----

Number of lunch seatings per day: 5

Are modular units currently present on-site and being used for classroom space?: NO

If "YES", indicate the number of years that the modular units have been in use:

Number of Modular Units:

Classroom count in Modular Units:

Seating Capacity of Modular classrooms:

What was the original anticipated useful life in years of the modular units when they were installed?:

Have non-traditional classroom spaces been converted to be used for classroom space?: YE

If "YES", indicate the number of non-traditional classroom spaces in use:

Please provide a description of each non-traditional classroom space, its originally-intended use and how it is currently used (maximum of 1000 characters).:

Since 2002, the District has annually metamorphosized numerous areas throughout the building in an effort to meet its educational obligations to all students. The District is currently at a point where infrastructural changes are no longer viable, leaving the District in a static position relative to growth and innovations.

Please explain any recent changes to the district's educational program, school assignment polices, grade configurations, class size policy, school closures, changes in administrative space, or any other changes that impact the district's enrollment capacity (maximum of 5000 characters).:

In the period from SY02 to SY17, the student population at Diman grew from 1120 to over 1400.

What are the district's current class size policies (maximum of 500 characters)?:

Class Size Policy: By District contract with the Diman Teachers Association, twenty-five (25) students, exclusive of

	physical education and Special Education classes, is considered a maximum classroom number. In the vocational
	programs, the recommended student maximum guidelines are established by the Massachusetts Department of Elementary and Secondary Education department.
L	

Name of School

---- SAMPLE SCHOOL [DRAFT] ----

Name of School

As Diman has been a beacon of educational excellence in the area of technical education since 1968, the expectation on the part of the school and community-at-large is that graduates will have the requisite acquisition of skills and competencies required for economic verticality regardless of whether the chosen pathway is college or career. This has resulted in an annual tradition of rejecting over 200 students each year who are seeking, but will not obtain, the opportunity to experience the aformentioned technical educational experience. To support this claim, comprehensive districts are trying to increase technical program			
opportunities and a	are using the growing demand for skilled laborers as a mechanism by which to achieve this objective.		

Question 2: Please describe the measures the School District has taken or is planning to take in the immediate future to mitigate the problem(s) described above.

The only tangible response that the District has relative to reducing the overcrowding problem is to reduce enrollment. This is politically untenable as the District realizes a minimum of forty percent surplus in admission applications each year. Efforts to institutionalize the cost of renovations/additions have been negated as a result of annual reductions in revenues streams at the state and national level. Additionally, the four sending communities have never, in the history of Diman Regional, agreed to pay over the minimum contribution as required by law. Furthermore, the creation of a campus off-site has been hampered by the increase in transportation costs coupled with reductions in Chapter 71 funding. Also, anecdotal polling of teachers and parents alike, indicates that support for a building separate from Diman proper has not gained momentum. As such, the possibility of leasing or purchasing off-site is financially and politically untenable.

Additionally, the District has reviewed the possibility to create an annex for the non-industrial programs. Unfortunately, logistical problems including transportation, counseling, meal service, etc. are cost prohibitive and therefore, limiting the District's flexibility regarding this strategy.

Question 3: Please provide a detailed explanation of the impact of the problem described in this priority on your district's educational program. Please include specific examples of how the problem prevents the district from delivering the educational program it is required to deliver and how students and/or teachers are directly affected by the problem identified.

The transformation of vocational education that has taken place as a result of both technological innovation and globalization has created a condition of future functional obsolescence for Diman as an educational institution, impacting the students therein. To be specific, the design of many of Diman's vocational programs, while adequate and appropriate for the 20th century, have now become antiquated. For example, the Automotive Technology program will need to be completely re-designed as a result of the impact of artificial intelligence relative to driver-less vehicles. Further difficulties exist in our construction cluster trades where verticality is required to create industry-standard simulations that will prepare our students for employment upon graduation. Additionally, the Machine Tool Technology program will also require a complete overhaul as the nature of the program has transitioned from a manufacturing to a laboratory environment.

More importantly still is the school's inability to meet the needs of the community relative to health and dental care that could be provided with the addition of a public clinic in each of these trade areas. This type of experiential learning environment would benefit our students in preparation for a career in these fields. Also, the inability to house both a school store and bank have proven detrimental to our students. In particular, students in the construction cluster who would have access to banking and financial fundamentals and would likely engage in investment strategies while in attendance that would afford them the opportunity for greater financial verticality as they invest and "flip" properties within their communities.

Please also provide the following:

Cafeteria Seating Capacity: 288

Number of lunch seatings per day: 5

Are modular units currently present on-site and being used for classroom space?: NO

If "YES", indicate the number of years that the modular units have been in use:

Number of Modular Units:

Classroom count in Modular Units:

Seating Capacity of Modular classrooms:

What was the original anticipated useful life in years of the modular units when they were installed?:

Have non-traditional classroom spaces been converted to be used for classroom space?: NO

If "YES", indicate the number of non-traditional classroom spaces in use:

Please provide a description of each non-traditional classroom space, its originally-intended use and how it is currently used (maximum of 1000 characters).:

Please explain any recent changes to the district's educational program, school assignment polices, grade configurations, class size policy, school closures, changes in administrative space, or any other changes that impact the district's enrollment capacity (maximum of 5000 characters).:

The District continually rejects over 200 students each year as a result of space limitations. Additionally, the ability to expand our student population to meet the technological changes and subsequent industry demands for new programs and employees cannot be met by the District, as no space exists to house new programs. The result of this is an increase in requests by our community schools to create their own Chapter 74 programs. Unfortunately, the quality and rigor of the education that is afforded students at a traditional technical school is being substituted with the aforementioned programs within a comprehensive setting. Practically speaking, the competencies and skills that are generally associated with full-time technical instruction are mitigated within the comprehensive setting, leading to a less highly-skilled employee at the exact time where industry demands are just the opposite.

What are the district's current class size policies (maximum of 500 characters)?: Through mutual agreement between the Greater Fall River Vocational School District Committee and the Diman Teachers' Association, the maximum number of students per class is twenty-five (25), exclusive of physical education and Special Education classes. Vocational program limits are set by DESE, as well as other industry or state
recommendations.

Name of School

---- SAMPLE SCHOOL [DRAFT] ----

Question 1: Please provide a detailed description of the issues surrounding the school facility systems (e.g., roof, windows, boilers, HVAC system, and/or electrical service and distribution system) that you are indicating require repair or replacement. Please describe all deficiencies to all systems in sufficient detail to explain the problem.

The forced hot water heating system is currently 52 years old and is currently burning #4 heavy fuel, creating both environmental and health related concerns as particulate matter is locally spewed in the populated school areas. To accommodate this antiquated heating system, the school has the burden of maintaining a 20,000 gallon underground oil tank, creating the real possibility of a hazardous spill that would require costly mitigation as the school is located in close proximity to the city drinking water. The building mechanical controls for monitoring heating is non-existent, creating an unhealthy and widely varied temperatures throughout the building.

Additionally, the District has no ability to employ a cleaner source of energy, as the school is "gas-starved", the result of distribution limits from the gas supplier. What is more, the roof structure of the building proper does not meet current weight and wind-shear loads for any necessary equipment to update the air quality. Furthermore, the ability of the District to incorporate solar applications to reduce the energy costs of the District and the carbon footprint created by the school's energy use. Automated air exchange that provides a healthy or safe school environment is not an application that is currently in service at Diman due to the age and cost for retrofitting. Also, the inability of the District to install energy efficient air cooling units is well documented as installation involves roof and structural matters that cannot be resolved.

Throughout the facility, electrical power is being distributed with original electrical equipment that does not efficient nor meets code. More concerning is the fact that replacement of panel boards and electrical switch gear is impossible as replacement parts do not exist for the current application. The safety issues that this creates includes: grounding, neutrals, and load imbalances.

The majority of the building's waste pipe has deteriorated and requires constant patching to address health issues. The District is partnering with the Department of Environmental Management and is monitoring lead and copper levels throughout the building, which is a situation that will unavoidably increase as the building continues to age. The school continues to have copper pipe degradation leading to failures causing water damage to the equipment and building proper. Furthermore, maintenance within the building proper becomes a monumental task as a result of hazardous building materials including asbestos that is throughout the building including ceilings, walls, caulking, etc.

Building-wide pneumatic systems need to be replaced. It is difficult to retrofit lighting fixtures due to asbestos prevalence throughout the ceiling area surfaces, resulting in significantly higher energy costs for the District.

Communication is a safety issue because of antiquated cabling throughout the facility. This impacts the ability to properly monitor life safety issues. The District does not have wet fire suppression throughout the facility, thus resulting in sub-standard protection relative to current standards. The school's current telephone system is a cable-based system integrated with original cabling. Classroom safety issues requiring immediate communication via the phone system is compromised. ADA compliance is far short of standards for new construction. Mechanical lifts fall short of the expectations for building accommodations.

The building envelope is compromised throughout the facility due to water penetration through masonry failure, as mortar repointing has deteriorated and will be profoundly costly to repair. This also creates moisture issues that de-laminates plaster material and seeps into interior learning spaces. Many doors and frames have deteriorated and are no longer plum, creating possible safety/security issues with regard to access and egress.

Question 2: Please describe the measures the district has already taken to mitigate the problem/issues described in Question 1 above.

The District maintains the oil-fired boilers in an effort to maintain the heating source while attempting to improve energy efficiency.

The District has participated in multiple energy audits in an attempt to mitigate energy costs and in an attempt to create a cleaner learning environment. Also, mechanical operations are scheduled routinely through the school's job order system. In addition, the maintenance department has a prevention and maintenance schedule that applies to all mechanical systems that require routine maintenance.

The electrical department and private contractors work continuously to update electrical mechanical systems as best as possible relative to keeping the existing electrical infrastructure operating safely.

The compressors which operate the pneumatic systems are routinely serviced by an independent contractor. Nonetheless, routine maintenance has become excessively expensive and ineffectual due to the age of the equipment.

The school has made software updates to the public address phone system for expansion. However, the current climate relative to school security and effective communication between classes and offices is hampered by the obsolescence of the current system.

Within the last 18 years, the District has replaced the roofs, windows, and wall systems. However, the masonry portion of the school envelope is woefully deteriorated due to the limits of the District's financial capability to invest in capital projects.

Question 3: Please provide a detailed explanation of the impact of the problem/issues described in Question 1 above on your district's educational program. Please include specific examples of how the problem prevents the district from delivering the educational program it is required to deliver and how students and/or teachers are directly affected by the problem identified.

At the most basic level, the inability of the school to deliver consistent heating or cooling creates an environment that is a complete distraction to students and teachers to effectively deliver rigorous content that stimulates higher order thinking.

The "gas starving" situation existing as mentioned in question 1 is resulting in a number of conditions that limit the ability of the instructional staff to deliver the most up-to-date curricula relative to gas-dependent equipment. For example, the HVAC and Plumbing shops are denied an opportunity to teach experientially lessons on high efficiency gas-fired appliances; Automotive Collision program is limited in their ability to operate their paint spray booths, which is a critical aspect of the curriculum; the District has deferred the installation of a spray booth in the Carpentry program due to this limitations as well; and, the District also cannot install additional gas-fired equipment within the Culinary Arts program.

As a corollary, the expense of using more expensive energy sources to power the equipment both in building proper and vocational programs is resulting in significant costs that take away from further improvements that can be made.

Over 60% of our students are eligible for free and reduced lunch, the ability to access free drinking water safe from impurities should be a minimum standard for all school buildings. Unfortunately, this is not the case at Diman where the piping infrastructure consists of copper and lead piping, which was the standard for buildings in 1967.

In January 2017, the District experienced a major controls failure in the Electronic Technology program resulting in over \$150,000 of building and equipment damage. Subsequent damage is not a matter of "if", but "when".

The quantity and quality of air to support our vocational program and the building proper is far short for current needs and expansion of the latest technology and equipment. Additionally, the ability of the District to provide clean mechanically controlled air to support the overcrowded programs and school building is unavailable.

As security and communication infrastructure are inherently tied together, the fact that Diman's current cabling distribution system is antiquated beyond maintenance or repair jeopardizes the safety of all as minutes count in an emergency. For example, the inability to dial out directly from the classroom to emergency services causes delays that impact timely emergency response.

The fact that lifts cannot be operated independently by those who need have disabilities reduces their feeling of independence and is anothema to the philosophy of the District in providing equal access to all stakeholders.

The infiltration of moisture via precipitation into the learning environment results in an unsafe walking condition that requires constant monitoring and maintenance. This also causes the drywall to de-laminate and becomes friable leading to aesthetic and structural issues.

Doors and door frames that are compromised due to structural failure result in doors that are difficult to open, as well as difficult to secure. This creates a problem for egress during emergencies as well as problems securing access points within the school.

Question 4: Please describe how addressing the school facility systems you identified in Question 1 above will extend the useful life of the facility that is the subject of this SOI and how it will improve your district's educational program.

The ability to prioritize and address deficiencies of mechanical systems is part of the District's Capital Improvement Plan that is currently being developed. All capital improvements are prioritized based on providing a safe and engaging learning environment. This requires, at the most basic level, the ability of the District to maintain learning milieus that are temperate, free from environmental contaminants or safety hazards. As previously mentioned, fresh air, clean water, dependable heat and cooling systems, reliable communication and mechanical infrastructure, as well as safe and secure means of entrance and egress are all part of the improvements that the District is planning to institutionalize starting in the 2018 school year. All of these improvements will satisfy the lowest level of the Maslovian Hierarchy, allowing students to focus on learning strategies with the goal of maximizing their learning potential.

Please also provide the following:

Have the systems identified above been examined by an engineer or other trained building professional?:

NO

If "YES", please provide the name of the individual and his/her professional affiliation (maximum of 250 characters):

The date of the inspection:

A summary of the findings (maximum of 5000 characters):

N/A

Question 1: Please provide a detailed description of the programs not currently available due to facility constraints, the state or local requirement for such programs, and the facility limitations precluding the programs from being offered.

Currently, the Commonwealth of Massachusetts CVTE Division offers over 45 vocational/technical trade areas. Unfortunately for Diman, due to the limitations of the building, which prohibits us from the desired expansion we would like to offer our students, we currently offer only 16 trade areas. These space and program limitations are particularly troubling relative to providing equity of education for our female population, many of whom, unfortunately, are excluded from the trade areas they desire due to our limited offerings.

Equally troubling for the District is the stock reality that our infrastructure and building limitations are restricting the ability of the instructors and students alike to create a learning environment that is technologically progressive and meets the standards of a global market. This condition is leading to an instructional environment that is inferior in comparison to other programs that have been fortunate enough to renovate or rebuild to today's industry standards. The reality is that our academic and vocational areas are rapidly becoming functionally obsolescent and without major changes, will create a bifurcated educational condition that places our students at a distinct disadvantage to their counterparts. As an aside, the school is struggling to meet one of its 2017 District Improvement Goals to improve the social emotional learning opportunities for its students as space limitations prohibit group and independent counseling opportunities that have proven effective to reduce student anxiety that allows for barrier free learning.

Question 2: Please describe the measures the district has taken or is planning to take in the immediate future to mitigate the problem(s) described above.

The ability to prioritize and address instructional deficiencies resulting from space limitations is partially being addressed through the District's Capital Improvement Plan. However, it is an admitted fact that the majority of initiatives within said plan are focused on operational deficiencies rather than expansion possibilities. The District plans to implement a new Chapter 74 program called Computer Programming and Web Development for the 2018 school year. That said, the implementation of additional programs are being prepared beyond the 2018 school year, with the realization that space limits will impact further program expansion.

Question 3: Please provide a detailed explanation of the impact of the problem described in this priority on your district's educational program. Please include specific examples of how the problem prevents the district from delivering the educational program it is required to deliver and how students and/or teachers are directly affected by the problem identified.

Due to the current space limitations that exists at Diman, every vocational program suffers the real possibility of becoming functionally obsolescent. For example, the Automotive Technology program will need to be completely re-designed as a result of the impact of artificial intelligence relative to the advent of driver-less vehicles. Further difficulties exist in our construction cluster trades where verticality is required to create industry-standard simulations that will prepare our students for employment upon graduation. Additionally, the Machine Tool Technology program will also require a complete overhaul as the nature of the program has transitioned from a manufacturing to a laboratory environment. The Automotive Collision program will need reconfiguration as a result of the market demand for energy efficient aluminum-framed vehicles that cannot be safely repaired with steel-framed vehicles.

Also, many of our vocational programs are transformed academic classrooms which self-evidently lack the environment in which students will gain employment post-graduation.

Additionally, the ability to create an environment that reflects the highest level of learning through technology, science, engineering and other areas cannot be realized as the infrastructural changes that are required to this building to meet these goals cannot be done without mitigation of hazardous materials to achieve this objective. The impracticality and costs associated with ameliorating this condition is well documented.

REQUIRED FORM OF VOTE TO SUBMIT AN SOI

REQUIRED VOTES

If the SOI is being submitted by a City or Town, a vote in the following form is required from both the City Council/Board of Aldermen **OR** the Board of Selectmen/equivalent governing body **AND** the School Committee.

If the SOI is being submitted by a regional school district, a vote in the following form is required from the Regional School Committee only. FORM OF VOTE Please use the text below to prepare your City's, Town's or District's required vote(s).

FORM OF VOTE

Please use the text below to prepare your City's, Town's or District's re	quired vote(s).
Resolved: Having convened in an open meeting on	_, prior to the closing date, the
	[City Council/Board of Aldermen,
Board of Selectmen/Equivalent Governing Body/School Committee] Of	[City/Town], in
accordance with its charter, by-laws, and ordinances, has voted to author	rize the Superintendent to submit
to the Massachusetts School Building Authority the Statement of Interest	dated for the
	[Address] which
describes and explains the following deficiencies and the priority category	y(s) for which an application
may be submitted to the Massachusetts School Building Authority in the	future
; [In	sert a description of the priority(s) checked off
on the Statement of Interest Form and a brief description of the deficiency described therein for each prior	rity]; and hereby further
specifically acknowledges that by submitting this Statement of Interest F	Form, the Massachusetts School
Building Authority in no way guarantees the acceptance or the approval of	of an application, the awarding of
a grant or any other funding commitment from the Massachusetts School	Building Authority, or commits
the City/Town/Regional School District to filing an application for funding	ng with the Massachusetts School
Building Authority.	

CERTIFICATIONS

The undersigned hereby certifies that, to the best of his/her knowledge, information and belief, the statements and information contained in this statement of Interest and attached hereto are true and accurate and that this Statement of Interest has been prepared under the direction of the district school committee and the undersigned is duly authorized to submit this Statement of Interest to the Massachusetts School Building Authority. The undersigned also hereby acknowledges and agrees to provide the Massachusetts School Building Authority, upon request by the Authority, any additional information relating to this Statement of Interest that may be required by the Authority.

Chief Executive Officer *	School Committee Chair	Superintendent of Schools	
Thomas F. Aubin Paul Jennings		Thomas F. Aubin	
Superintendent			
(signature)	(signature)	(signature)	
Date	Date	Date	

^{*} Local Chief Executive Officer: In a city or town with a manager form of government, the manager of the municipality; in other cities, the mayor; and in other towns, the board of selectmen unless, in a city or town, some other municipal office is designated to the chief executive office under the provisions of a local charter. Please note, in districts where the Superintendent is also the Local Chief Executive Officer, it is required for the same person to sign the Statement of Interest Certifications twice. Please do not leave any signature lines blank.

- If a District selects Priority #1, Replacement or renovation of a building which is structurally unsound or otherwise in a condition seriously jeopardizing the health and safety of the school children, where no alternative exists, the MSBA requires a hard copy of the engineering or other report detailing the nature and severity of the problem and a written professional opinion of how imminent the system failure is likely to manifest itself. The District also must submit photographs of the problematic building area or system to the MSBA.
- If a District selects Priority #3, Prevention of a loss of accreditation, the MSBA requires the full accreditation report(s) and any supporting correspondence between the District and the accrediting entity.

ADDITIONAL INFORMATION: In addition to the information required with the SOI hard copy submittal, the District may also provide any reports, pictures, or other information they feel will give the MSBA a better understanding of the issues identified at a facility.

If you have any questions about the SOI process please contact Diane Sullivan at 617-720-4466 or Diane.Sullivan@massschoolbuildings.org.

ATTACHMENT B:

Contract for Designer Services –

Base Contract for Design Bid Build or CM-at-Risk Project



```
THINONALLY LEFT BLANK
```

CONTRACT FOR DESIGNER SERVICES (BASE CONTRACT FOR DESIGN BID BUILD OR CM at RISK PROJECT)

This Contract is made as of this	day of	in the year		between
the	(day)	(month)	(year)	
(Owner)	, ,		(street)	
(City)	, <u>N</u>	Massachusetts (State)		Zip Code)
hereinafter called "the Owner" and		. ,	(2	Exp Code)
		(Designer)		
hereinafter called the "Designer" for and Extra Services described herein a	the Designer		•	d to complete the Basic
		(name/description of Proje	ect)	
The Designer is authorized to per Phase and, pending receipt of a Design Phase. At the Owner's op design phases and/or the Construupon amendment to this Contract elects to construct the Project usi M.G.L. c. 149A, this Contract sha it may be amended from time to the Design-Bid-Build ("DBB") of shall be amended using the Authoritime by the Authority. For the performance of the service Schematic Design Phase, and excellent.	written Apportion, the Destition, the Destition Phases the will be exert as the CM at the construction with the Apportunity's Standard ces required	proval to proceed from signer may be authorised and Completion Placeted between the Coat Risk ("CM-R") coal using the Authority. If the Own delivery method pured Amendment for Dunder this Contract of	om the Owner, the ized to perform so hase, at which time owner and the Denstruction deliver by's Standard Americal and the M.G.L. and the M.G.L. are the standard by the stand	arough the Schematic ervices for subsequent me a mutually agreed esigner. If the Owner method pursuant to endment for CM-R, as truct the Project using c. 149, this Contract amended from time to a Study Phase and the
and 8.3, the Designer shall be of Payment Schedule included as Att Designer's Project Architect/Enginee The Subconsultants to provide service	compensated tachment A.	by the Owner for I	Basic Services in	accordance with the
include the following, as identified or	n the RFS:			NOT!

	Name of Firm	Name of Principal	MBE/ WBE
Civil Engineering			
Landscape Architecture			
Structural Engineering			
Fire Protection Engineering			
Plumbing Engineering			
HVAC Engineering			
Electrical/Lighting/			
Data/Communications			

Environmental Permitting		
Geotechnical Engineering		
Hazardous Materials		
Cost Estimating		
Kitchen/Food Service Consultant		
Laboratory Consultant		
Acoustical Consultant		
Specifications Consultant		
Library/Media/Audio Visual Consultant		
Technology Consultant		
Theatrical Consultant		
Sustainable/Green Design/Renewable		
Energy Consultant		
Code Consultant		
Accessibility Consultant		
Traffic Consultant		
Furniture, Fixtures and Equipment		·
Consultant		
Site Surveying		
Security Consultant		

IN WITNESS WHEREOF, the Owner and the Designer hereby agree to the terms of the Contract and have caused this Contract to be executed by their respective authorized officers or other authorized representatives.

OWNE	i.R	
	(print name)	
	(print title)	
By Date		
-	(signature)	
Date		
DESIG		
	(print name)	
	(print title)	
By		
Ву	(signature)	
Date		

TABLE OF CONTENTS

PROJECT DE	SCRIPTION, FEE AND SIGNATURES
ARTICLE 1	DEFINITIONS
ARTICLE 2	RELATIONSHIP OF THE PARTIES
ARTICLE 3	RESPONSIBILITIES OF THE OWNER
ARTICLE 4	RESPONSIBILITIES OF THE DESIGNER
ARTICLE 5	SUB-CONSULTANTS
ARTICLE 6	COMPENSATION
ARTICLE 7	BASIC SERVICES
ARTICLE 8	EXTRA SERVICES
ARTICLE 9	REIMBURSABLE EXPENSES
ARTICLE 10	COMPENSATION AND RESPONSIBILITY FOR CHANGE ORDERS
ARTICLE 11	RELEASE AND DISCHARGE
ARTICLE 12	ASSIGNMENT, SUSPENSION, TERMINATION, NO AWARD
ARTICLE 13	NOTICES
ARTICLE 14	INDEMNIFICATION
ARTICLE 15	INSURANCE
ARTICLE 16	OWNERSHIP OF DOCUMENTS
ARTICLE 17	STATUTORY REQUIREMENTS
ARTICLE 18	MISCELLANEOUS
ATTACHMEN	
ATTACHMEN	
ATTACHMEN	
ATTACHMEN	T D LETTER OF INTERNAL CONTROL (DESIGNER)
ATTACHMEN	T E LETTER FROM CPA ON INTERNAL CONTROLS
ATTACHMEN	T F FORM OF DESIGNER CONTRACT AMENDMENT

ARTICLE 1: DEFINITIONS

All terms that this Contract defines may be used with or without initial capital letters. Other terms, abbreviations and references are defined as they appear herein. Words and abbreviations that are not defined in the Contract Documents but which have recognized technical or trade meanings are used in accordance with those meanings.

APPLICABLE LAWS – All applicable laws, statutes, ordinances, by-laws, codes, rules and regulations, of the Commonwealth of Massachusetts, its political subdivisions, and the Federal Government applicable to the Project.

APPROVAL -- A written communication from the Owner approving the work of the current Phase, as identified on Attachment A, or authorizing the Designer to proceed to the next Phase or approving the scope and compensation for either Extra Services or Reimbursable Expenses.

AUTHORITY – Massachusetts School Building Authority or its authorized representative, created by St. 2004, c. 208.

BASIC SERVICES – The scope of services to be provided by the Designer under this Contract, unless the Contract is otherwise terminated pursuant to Article 12, as described in Article 7 of this Contract, and as it may be amended pursuant to Article 18.4.

CERTIFICATE OF FINAL COMPLETION – The form prescribed by the Authority which contains the certification of the Designer, OPM and the Owner that the Project has reached Final Completion.

CERTIFICATE OF SUBSTANTIAL COMPLETION – The certificate prepared by the Designer and approved by the Owner to the effect that the Work has reached Substantial Completion.

CHANGE ORDER – A written instrument prepared by the Designer and signed by the Owner, Owner's Project Manager, Contractor or CM at Risk, and Designer, stating their agreement on a change in the Construction Contract Documents, including, but not limited to, a change in the Contract Sum and/or Contract Time, and/or any other specification in the Construction Contract Documents.

COMMISSIONING CONSULTANT – A person or firm engaged by the Authority to provide building commissioning services, including advisory services during design and construction.

CONSTRUCTION CONTRACT DOCUMENTS – The Construction Contract Documents consist of the Owner-Contractor or Owner-CM at Risk Agreement, Advertisement, Instructions to Bidders, Bidding Documents, Contract Forms, Conditions of the Contract, Drawings, Plans, Technical Specifications, all addenda issued prior to execution of the Construction Contract, and other documents approved after execution of the Owner-Contractor or Owner-CM at Risk Agreement relating thereto.

CONSTRUCTION MANAGEMENT AT RISK or CONSTRUCTION MANAGEMENT AT RISK SERVICES or CONSTRUCTION MANAGEMENT AT RISK DELIVERY METHOD or CM at RISK DELIVERY METHOD - a construction method described in M.G.L. c. 149A wherein a Construction Management at Risk firm provides a range of preconstruction services and construction management services which may include cost estimation and consultation regarding the design of the building project, the preparation and coordination of bid packages, scheduling, cost control, and value engineering, acting as the general contractor during the construction, detailing the Trade Contractor scope of work, holding the trade contracts and other subcontracts, prequalifying and evaluating Trade Contractors and subcontractors, and providing management and construction services, all at a Guaranteed Maximum Price, which shall represent the maximum amount to be paid by the public agency for the building project, including the cost of the work, the general conditions and the fee payable to the Construction Management at Risk Firm.

CONSTRUCTION MANAGER AT RISK, CONSTRUCTION MANAGEMENT at RISK FIRM or CM at RISK – the individual, corporation, partnership, sole proprietorship, joint stock company, joint venture or other entity with whom the Owner has contracted pursuant to M.G.L. c. 149A, §§ 6 & 7, to provide Construction Management at Risk Services.

CONTRACT – This Contract, inclusive of all Attachments, between the Owner and the Designer; all written amendments to this Contract; and all Approvals issued pursuant to this Contract.

CONTRACTOR OR GENERAL CONTRACTOR – The person or firm with whom the Owner has contracted pursuant to M.G.L. c. 149, §§ 44A-44M to perform the construction for this Project.

CONTRACTOR APPLICATION AND CERTIFICATE FOR PAYMENT – The form prescribed by the Owner which contains the Contractor's or CM at Risk's application or requisition for periodic or final payment for Work performed in accordance with the Construction Contract Documents and the Designer's certificate for payment as approved by the OPM and the Owner.

DESIGNER – The individual, corporation, partnership, sole proprietorship, joint stock company, joint venture or other entity identified as such on page one of this Contract performing architecture, landscape architecture, and/or engineering services under this Contract and which meets the qualifications set forth in M.G.L. c. 7 § 38A 1/2.

DESIGNER SERVICES – The services to be performed by the Designer and its Subconsultants under this Contract including developing and providing all data, designs, drawings, specifications and estimates required for the Project.

DISTRICT – see "OWNER."

EXTRA SERVICES – Services requested by the Owner to be performed by the Designer but which are additional (or "extra") to the services performed as Basic Services.

FEASIBILITY STUDY AGREEMENT – The agreement between the Owner and the Authority that sets forth the terms and conditions pursuant to which the Authority will collaborate with the Owner in conducting a feasibility study, which agreement shall include the budget, scope and schedule for the feasibility study.

FEE FOR BASIC SERVICES – The fee to be paid to the Designer for satisfactorily performing the Basic Services required under this Contract, exclusive of the compensation to which the Designer may be entitled pursuant to Articles 8 (Extra Services) and 9 (Reimbursable Expenses).

FINAL COMPLETION – The Work has been completed in accordance with the Construction Contract Documents and the educational specifications, schematic plans and drawings and the Project Funding Agreement approved by the Authority.

FINAL DESIGN PROGRAM – A description of the programmatic, functional, spatial, and environmental requirements of the Project in written and graphic form indicating the scope of work and design requirements of the Project.

GENERAL LAWS – The Massachusetts General Laws as amended, including any rules, regulations and administrative procedures implementing said laws.

GUARANTEED MAXIMUM PRICE or GMP - The agreed total dollar amount for the Construction Management at Risk services, including the cost of the Work, the general conditions and the fees charged by the Construction Management at Risk firm.

GUIDELINES AND STANDARDS – Documents published by the Authority including regulations and procedures that supplement the tasks of Designers contracting with Owners for projects receiving any funding from the Authority, as they may be amended from time to time by the Authority.

MATERIALS – The designs, drawings, project manual specifications, and other materials prepared by the Designer as defined in Article 16.1.

MBE/WBE – A minority-owned business (MBE) or a women-owned business (WBE) certified by the State Office of Minority and Women Business Assistance (SOMWBA).

NOTICE TO PROCEED – The written communication issued by the Owner to the Contractor or CM at Risk authorizing him to proceed with the construction contract and establishing the date for commencement of the contract time.

OWNER – The entity identified as such on page one of this Contract, or its authorized representative, that is the owner of the property that is the site of the Project, or has or will have exclusive control over the site for at least the duration of the useful life of the school facility that is the subject of the Project, and is responsible for administering this Contract.

OWNER-CONTRACTOR AGREEMENT or OWNER – GENERAL CONTRACTOR AGREEMENT – The contract between the Owner and one or more General Contractors and/or

goods or services providers for construction of a whole or part of the Project, including approved change orders.

OWNER-CM at RISK AGREEMENT – The contract between the Owner and the CM at Risk, including, but not limited to, the GMP Amendment, for the provision of Construction Management at Risk Services for the Project.

OWNER'S PROJECT MANAGER or OPM – The individual, corporation, partnership, sole proprietorship, joint stock company, joint venture or other entity with whom the Owner has contracted to perform the Project Management Services for this Project, and who meets the qualifications of M.G.L. c. 149, § 44A ½ and has been approved by the Authority.

PHASE – A distinct portion of the work of this Contract and its associated duration, as identified on Attachment A. Prior Approval to proceed for each Phase is required from the Owner.

PRINCIPALS – The owner(s) and/or officer(s) of the Designer or Subconsultant who are in responsible charge of the Project.

PROJECT – All work that pertains to the study, planning, programming, design, construction, reconstruction, installation, demolition, maintenance and repair, if any, as described in the Project Scope and Budget Agreement and Project Funding Agreement.

PROJECT ARCHITECT AND/OR PROJECT ENGINEER – The individual designated by the Designer as its Project Architect or Project Engineer. Such Project Architect or Project Engineer shall be a registered architect, engineer or landscape architect as required by the Request For Designer Services, shall be the person who shall oversee the performance of all services provided on the Project and shall be certified in the Massachusetts Certified Public Purchasing Official Program as administered by the Inspector General of the Commonwealth of Massachusetts.

PROJECT CONSTRUCTION BUDGET – That portion of the Total Project Budget that enumerates the cost of constructing the Project inclusive of all designed construction, demolition, and renovation work, all supportive and preparatory construction work required for the Project, the General Contractor or the CM at Risk and all subcontractors, suppliers, materials, equipment, general conditions, insurance, overhead and profit and all other expenditures that are ordinarily considered as construction cost allocations. The Project Construction Budget includes the design contingency,, bidding contingency, and price escalation contingency, as appropriate to the phase of the Project.

PROJECT FUNDING AGREEMENT – the Project Funding Agreement described in the 963 CMR 2.02 and executed by the Authority and the Owner.

PROJECT SCHEDULE – A complete list of all activities, time and sequence required to complete the Project, as defined in the Project Scope and Budget Agreement or Project Funding Agreement.

PROJECT SCOPE AND BUDGET AGREEMENT – the Agreement described in 963 CMR 2.10(10) and executed by the Authority and the Owner.

RECORD DRAWINGS – The drawings prepared by the Designer and its Subconsultants pursuant to Article 7.10.5 of this Contract which incorporate the design changes made during the construction period and which incorporate information on the marked-up prints, as-built drawings and other data furnished by the General Contractor or CM at Risk and any subcontractors.

REIMBURSABLE EXPENSES – Costs and expenses incurred by the Designer that are reimbursable pursuant to the provisions of Article 9 of this Contract.

REQUEST FOR DESIGNER SERVICES or RFS – The written document appended hereto as Attachment B specifying various requirements including the project goals and general scope, project site, scope of services, submission requirements, schedule, and construction budget.

STANDARD OF CARE – The generally accepted professional standard of care ordinarily used by design professionals performing a similar scope of services in the same geographic area on projects of comparable size and complexity.

SUBCONSULTANT – The Subconsultants listed on page 1 of this Contract, together with any additional Subconsultants engaged by the Designer from time to time, which shall be an individual, company, firm, or business having a direct contractual relationship with the Designer, who provides services on the Project.

SUBCONTRACTOR – The person or entity having a direct contractual relationship with the Contractor, or CM at Risk who has the contract to perform the construction of the Project, except as otherwise specifically provided or required herein or by Law. Subcontractor when used also means "Trade Contractor" except when otherwise specified.

SUBSTANTIAL COMPLETION – The Work, as evidenced by the Certificate of Substantial Completion, is fully complete or substantially complete so that the value of the Work remaining to be done is, in the estimate of the Owner, less than one percent of the original contract price, or (2) the Contractor substantially completes the work and the Owner takes possession for occupancy, whichever occurs first.

TOTAL PROJECT BUDGET – A complete and full enumeration of all costs of the Project, as defined in the Project Scope and Budget Agreement or Project Funding Agreement.

TRADE CONTRACTOR – a subcontractor having a direct contractual relationship with a Contractor or CM at Risk to perform one or more so-called sub-bid classes of work listed in M.G.L. c.149, §44F, and anyl other sub-bid classes of work selected by the Owner for the Project in accordance with the provisions of either M.G.L. 149, §44F(1)(a) or M.G.L. c. 149A, §8(a).

WORK – The entire construction required to be furnished under the Construction Contract Documents. Work includes performing and furnishing any and all services, obligations, duties,

responsibilities, labor, materials, equipment, temporary facilities, and incidentals necessary to complete the construction assigned to, or undertaken by the Contractor or the CM at Risk pursuant to the Construction Contract Documents.

ARTICLE 2: RELATIONSHIP OF THE PARTIES

- 2.1 The Owner's Project Manager shall act as an independent contractor of the Owner in providing certain project management services required for the Project required for the project except where the OPM is an existing public employee of the Owner as described in M.G.L. c. 149, § 149A1/2.
- 2.2 The Designer is solely responsible for providing the design for the Project and for performing in accordance with this Contract.
- 2.3 The Contractor or CM at Risk, as the case may be, shall be solely responsible for construction means, methods, techniques, sequences and procedures, the Contractor's or CM at Risk's schedules, and for safety precautions and programs in connection with the Project and for performing in accordance with the Owner-Contractor or Owner CM at Risk Agreement. The Designer shall be responsible for the Designer's negligent acts or omissions but shall not have control over or charge of acts or omissions of the Contractor or CM at Risk, Subcontractors, or the agents or employees of the Contractor or CM at Risk or Subcontractors, the Owner's Project Manager, the Authority or its Commissioning Consultant or other technical consultants.
- 2.4 Nothing in this Contract shall be construed as an assumption by the Designer of the responsibilities or duties of the Contractor or CM at Risk or the Owner's Project Manager. It is the intention of the parties that the Designer's services shall be rendered in a manner compatible with and in coordination with the services provided by the Owner's Project Manager and the Commissioning Consultant. It is not intended that the services of the Designer and the Owner's Project Manager or the Commissioning Consultant be competitive or duplicative, but rather complementary. The Designer shall be entitled to rely upon the Owner's Project Manager, Commissioning Consultant and Contractor or CM at Risk for the proper performance of their obligations pursuant to their respective contracts with the Owner.

ARTICLE 3: RESPONSIBILITIES OF THE OWNER

- 3.1 The Owner shall have the right to approve the Designer's work.
- 3.2 The Owner shall designate an individual who shall have the authority to act on behalf of the Owner under this Contract and who shall be responsible for day-to-day communication between the Owner and the Designer.
- 3.3 Upon satisfactory completion of services performed, the Owner shall make payments to the Designer as provided in Articles 6, 7, 8 and 9, 10 and 11.

- 3.4 To the extent such data is available, the Owner shall furnish to the Designer existing surveys of the site, building plans, borings, test pits, structural, mechanical, chemical or other test data, tests for air and water pollution and for hazardous materials, photographs, reports and utility information. The Designer shall be entitled to reasonably rely upon the sufficiency and accuracy of the information furnished to the Designer under this Article 3.4 and under Article 4.11, provided that the Designer shall coordinate its services with the services of the Owner's consultants and shall notify the Owner in writing of any deficiencies in such data of which the Designer becomes aware.
- 3.5 Except as otherwise provided in this Contract, or when direct communications have been specially authorized, the Owner shall endeavor to communicate with the Contractor or CM at Risk and the Designer's consultants through the Designer about matters arising out of or relating to the Construction Contract Documents. The Owner shall promptly notify the Designer of any direct communications that may affect the Designer's services.
- 3.6 The Owner shall provide the Designer access to the Project site prior to commencement of the Work and shall obligate the Contractor or CM at Risk to provide the Designer access to the Work wherever it is in preparation or progress.
- 3.7 If the Owner requests the Designer to execute any certificates that are not readily available as of the effective date of this Contract, the proposed language of such certificates shall be submitted to the Designer for review at least 14 days prior to the requested dates of execution. The Designer shall not be required to execute certificates or consents that would require knowledge, services or responsibilities beyond the scope of this Contract.
- 3.8 The Owner shall deliver to the Designer in a timely manner written copies of all Approvals required by this Contract. If Approval is withheld, the Owner shall notify the Designer in a timely manner in writing why such Approval is being withheld.
- 3.9 The Owner shall not unreasonably withhold, delay, condition, or deny any approval, acceptance, or consent required under this Contract, including any Approval.

ARTICLE 4: RESPONSIBILITIES OF THE DESIGNER

- 4.1 The Designer shall perform the Designer Services in accordance with the requirements of this Contract, and in accordance with the Standard of Care. The Designer shall exercise due care and diligence in the rendition of all services under this Contract in accordance with such professional standards and shall exercise the Standard of Care to provide the services required under this Contract in conformity with all Applicable Laws.
- 4.2 The Designer shall be responsible for the Designer Services including any changes to such Services that may be required in accordance with this Contract. The Designer shall furnish appropriate competent professional services for each of the Phases in accordance with the Standard of Care. Any changes, corrections, additions or deletions requested by the Owner and the Authority shall be incorporated into the design of the Project unless detailed objections thereto are issued in writing by the Designer, subject to Article 8.2.2. Nothing

- herein shall be construed as an assumption by the Owner or the Authority of the responsibilities or duties of the Designer.
- 4.3 The Designer Services shall be performed as expeditiously as is consistent with orderly progress of the work, consistent with the agreed upon project design schedule as established under Article 7.4.2 and as it may thereafter be amended by the parties from time to time. In the event of delays due to causes outside of the Designer's control, the project design schedule may be extended as necessary, and Designer's compensation may be equitably adjusted pursuant to Article 6.6 to the extent that Designer incurs additional direct costs caused by the delay. Time is of the essence for the duration of this Contract.
- 4.4 The Designer shall provide the scope of services required by this Contract, as described in more detail in the RFS and Attachment A.
- 4.5 The Designer shall comply with the terms and conditions of all project agreements executed between the Owner and the Authority and any and all administrative directives issued by the Authority, now in effect or hereafter promulgated during the term of this Contract, without any additional compensation, that are applicable to Designer's Services under this Contract and that have been provided or are readily available to Designer prior to such Services being performed. The Owner shall reasonably compensate the Designer for complying with any term or condition of a project agreement executed between the Owner and the Authority or any administrative directive issued by the Authority, that was not provided to or was not readily available to the Designer prior to such Services being performed and that materially impacts the Designer's scope or other aspect of its Services, Fee, schedule, or any obligations and responsibilities under this Contract.
- The Designer acknowledges the importance that the Owner attributes to the abilities and qualifications of the key members of the Designer's team, including Subconsultants, and the continuity of key members' participation in the services to be provided under this Contract. This Contract has been entered into in reliance on the Designer's representation that the individuals, consultants, assignments and responsibilities will be maintained throughout the duration of this engagement. No substitution or replacement of individuals or change in the Subconsultants, listed on pages 1-2 of this Contract, shall take place without the prior written approval of the Owner and the Authority, except when necessitated by causes beyond the Designer's control (such causes shall include if an individual leaves or is no longer associated with the Designer's firm). If the Designer proposes to replace one of the members of the Designer's team, the Designer shall propose a person or consultant with qualifications at least equal to the person or firm the Designer proposes to replace. The Owner and the Authority shall have the right to approve any substitution or replacement or change in status for the persons or Subconsultants listed on page 1-2 of this Contract and such approval shall not be unreasonably withheld. At the request of the Owner, the Designer shall consult with the Owner to resolve any situation in which the Owner determines that a member of the Designer's team is failing to perform services in an acceptable manner to the Owner. The Owner shall have the right to direct the removal of any such person or consultant. The Owner shall work in good faith with the Designer to resolve any material problems identified by the Owner in writing regarding performance of the Designer's obligations under this Contract. No act or omission of the Owner or the Authority made or permitted under this Article shall relieve the Designer of its responsibility for the performance of the services specified in this Contract.

- 4.7 The Designer shall compile and distribute a job directory which includes all names, addresses, phone and fax numbers, and e-mail addresses of the representatives of the Designer and their Subconsultants. This shall be distributed upon commencement of the services, and shall be updated and redistributed as project participants and/or contact information change.
- 4.8 The Designer shall employ at all times adequate professional and support personnel with requisite expertise and adequate numbers to assure the complete, timely performance of the obligations of the Designer. The Designer shall acquaint its employees and Subconsultants with all provisions of the General Laws governing public construction projects, including but not limited to M.G.L. c. 149, M.G.L. 149A, and M.G.L. c. 30, that are relevant to the performance of Designer's obligations under this Contract. When directed by the Owner, the Designer shall fully cooperate with the Owner in obtaining the Criminal Offender Record Information (CORI) of the Designer and its employees and of any Subconsultants and their employees in accordance with the provisions of M.G.L. c. 71, § 38R, M.G.L. c. 6, §§ 167-178B (the so-called CORI Law), any other applicable law, and District policy. All contracts between the Designer and each Subconsultant shall include appropriate provisions requiring the Subconsultant to fully cooperate with the Owner in obtaining the Criminal Offender Record Information (CORI) of the Subconsultant and its employees as aforesaid.
- 4.9 The Designer shall be and shall remain liable to the Owner for all damages incurred by the Owner as a result of the failure of the Designer or its Subconsultants to perform in conformance with the terms and conditions of this Contract.

4.10 Design Within the Project Construction Budget

- 4.10.1 The Designer shall prepare cost estimates for the Project as described in Article 7 of this Contract or at more frequent intervals as required in the RFS. Unless_otherwise specified in the RFS, the cost estimates shall be considered Basic Services and the Designer is not eligible for any additional compensation for preparing the same. The format for cost estimates shall be in accordance with the requirements of the Authority.
- 4.10.2 The Designer shall produce a design for the Project meeting the requirements of the scope of work described in the RFS to be constructed within the Project Construction Budget, provided that the Designer shall be permitted to recommend to the Owner such adjustments to the Project's design, consistent with the Project Funding Agreement, as the Designer reasonably believes may be required to adhere to the Project Construction Budget. In the event the Designer's cost estimate for the Project (as reconciled in accordance with the provisions of this Contract) exceeds the Project Construction Budget, the Owner may require the Designer to revise the design, drawings and specifications to keep the cost estimate for the Project within the Project Construction Budget. The Designer shall not be entitled to extra compensation for making such revisions to contain costs within the Project Construction Budget.
- 4.10.3 In a Project constructed pursuant to M.G.L. c. 149, §§ 44A-M, if the Project Construction Budget is exceeded by the lowest bona fide, responsible bid by any

amount, the Owner shall direct the Designer to review and compare the Project Construction Budget with the bids received to identify the variances. Upon completion of this review and submission of the Designer's report to the Owner and Authority, the Owner shall, with the approval of the Authority:

- (a) direct the Designer to revise the Final Design Program, Project scope and quality as required to reduce the estimated construction costs to be within the Project Construction Budget, in accordance with Article 4.10.5 of this Contract; or
- (b) give written approval to the Designer of an increase in the Project Construction Budget; or
- (c) authorize rebidding of the Project within a reasonable time; or
- (d) terminate this Contract in accordance with Article 12.3; or
- (e) implement any other mutually accepted alternative that the Owner and the Designer may agree on.
- 4.10.4 In a Project constructed pursuant to M.G.L. c. 149A, the Designer shall be responsible for managing the design of the Project to stay within the Project Construction Budget. If the GMP proposal submitted by the CM at Risk exceeds the Project Construction Budget, the Designer shall review and compare the Project Construction Budget with the GMP proposal submitted by the CM at Risk to identify the variances. Upon completion of this review, if directed by the Owner, the Designer shall assist the Owner in negotiating a GMP within the Project Construction Budget in accordance with Article 7.7.9. If a GMP cannot be successfully negotiated between the Owner and the CM at Risk within the Project Construction Budget, the Owner shall, with the approval of the Authority:
 - (a) direct the Designer to participate with the Owner, OPM, and CM at Risk in design reviews and revise the design, including appropriate revisions to drawings and specifications, as necessary in order to reach an agreement on a GMP within the Project Construction Budget; in accordance with Article 4.10.5; or
 - (b) give written approval to the Designer of an increase in the Project Construction Budget and resume negotiating a GMP with the CM at Risk; or
 - (c) terminate this Contract in accordance with Article 12.3; or
 - (d) implement any other mutually accepted alternative that the Owner and the Designer may agree on.
- 4.10.5 (a) If the Owner chooses to proceed under Article 4.10.3(a) or 4.10.4(a), the Designer and its Subconsultants, without receiving additional compensation, except if fewer than three bona fide, responsible bids were received (in the case of a Project constructed pursuant to M.G.L. c. 149, §§ 44A-44M) or (in the case of a Project

constructed pursuant to G.L. c. 149A) if fewer than three bona fide responsible Trade Contractor or so-called non-trade contractor bids for each category of work were received, or if 4.10.5(b) and/or (c) applies, shall cooperate in revising the designs, drawings and specifications as may be required to reduce or modify the quality or scope or both, of the Project so that they will comply with the Project Construction Budget as approved at the conclusion of the Construction Documents Phase or as amended. Any changes to the educational program or the approved space summary shall be subject to the written approval of the Authority. Upon completion of these revisions, the Designer shall also be required to produce a revised cost estimate demonstrating that the estimated cost of the Project does not exceed the Project Construction Budget. Revising the designs, drawings, and specifications and updating the cost estimate shall be the sole obligation on the part of the Designer with respect to 4.10.3(a) or 4.10.4(a); (b) If the Owner elects to proceed with revisions that significantly increase the complexity either of the Construction Contract Documents themselves or the Construction Administration Phase services that the Designer will have to provide, then the Designer shall be entitled to an equitable adjustment in its Fee to reflect the impact on its services: (c) If the bid or proposal referenced in 4.10.3 or 4.10.4 above was submitted on a date that is more than three (3) months after approval of the Construction Contract Documents then such revisions shall be Extra Services.

- 4.10.6 The Designer must receive written approval of the Owner and the Authority before the Project Construction Budget shall be considered amended.
- 4.11 Additional Tests and Surveys: The Designer shall be responsible for reviewing the surveys, investigations, testing and reports completed by the Owner and as provided under Article 3.4, and determining the types of additional or expanded surveys, investigations, or testing required for the Project. Such services shall be provided by qualified specialty Subconsultants as necessary. Both the types of services and the Subconsultants shall be approved by the Owner. In the event that the Designer employs the services of a Subconsultant to provide such services, the Designer shall employ such Subconsultants who have the professional liability insurance coverage described in paragraph 15.8.1 covering such services, to the extent that such insurance coverage is generally available to Subconsultants. The Designer shall, upon the Owner's written request, assign to the Owner the Designer's contractual right to pursue a claim against such Subconsultants. Such services shall be paid for as provided in Article 8 Extra Services unless such services are specifically included as Basic Services in the RFS. Such services may include but need not be limited to:
 - 4.11.1 Site surveys;
 - 4.11.2 Structural tests and materials tests;
 - 4.11.3 Geotechnical and geoenvironmental investigations and reports, including existing buildings hazardous material reports, boring tests, test pits, observation wells, testing and chemical analysis of site substrate conditions;
 - 4.11.4 Traffic studies.

ARTICLE 5: SUBCONSULTANTS

- 5.1 The Designer may engage Subconsultants, subject to the prior written approval of the Owner and subject to Article 9.3, in order to perform services under this Contract. If Subconsultants are engaged, the person responsible for, and in control of, the Subconsultant services to be provided must be professionally registered or licensed in Massachusetts in the necessary disciplines for the services if such registration or licensing is required under the applicable General Laws. The engagement of Subconsultants shall not in any way relieve the Designer from its duties and responsibilities for its work, including, without limitation, coordinating all Designer Services furnished under this Contract by the Subconsultants.
- 5.2 Upon request, the Designer shall provide the Owner with copies of its agreements with Subconsultants, including any amendments thereto and copies of the Subconsultant's applicable certificates of insurance.
- 5.3 No substitution of Subconsultants and no use of additional Subconsultants or assignment of services shall be made without prior written approval of the Owner, which approval shall not be unreasonably withheld.

ARTICLE 6: COMPENSATION

- 6.1 For the satisfactory performance of all services required pursuant to this Contract, excluding those services specified under Articles 8 and 9, the Designer shall be compensated by the Owner in the amounts specified in Attachment A as that Fee may be amended by written amendment to this Contract.
- 6.2 When the Designer receives payment from the Owner, the Designer shall promptly make payment to each Subconsultant whose work was included in the work for which such payment was received unless payment has been theretofore made. The Owner shall have the contractual right to investigate any breach of performance of a Subconsultant and to initiate corrective measures it determines are necessary and in the best interest of the Owner. All contracts between the Designer and its Subconsultants shall include a provision in which the Owner's rights to initiate corrective action shall be stipulated.

6.3 Payment Schedule

- 6.3.1 Payments for Basic Services shall be made monthly and, where applicable, shall be in proportion to services performed within each Phase. The amount of fees attributable to each Phase shall be as set out in the schedule in Attachment A. Payment for approved Reimbursable Expenses and/or Extra Services shall be made monthly upon receipt of an approved invoice from the Designer.
- 6.3.2 The Owner shall make payments to the Designer within 30 days of the Owner's approval of an invoice from the Designer. The Owner's payment for any services provided under this Contract shall not be construed to operate as a waiver of any rights under the Contract or any cause of action arising out of performance of the Contract. The Owner shall not withhold payments to offset costs alleged to have been incurred by the Owner on account of allegedly negligent acts, errors or omissions unless the Designer agrees or has been found liable for specific amounts in a binding agreement or court judgment, or unless the Designer fails to maintain the professional

liability insurance required under paragraphs 15.7.1 and 15.7.2. The Owner may withhold approval of invoice items the Owner reasonably believes have not been performed in accordance with this Contract, including adjustments to payment amounts in instances where required submittals to the Authority may be found to be missing or incomplete. If Owner and Designer continue to disagree, the disagreement shall be immediately submitted to mediation in accordance with paragraph 18.5(b).

6.4 <u>Installment Payments During Construction</u>

- 6.4.1 During the construction Phase, the Designer shall be paid the Fee for Basic Services stipulated in Attachment A.
- 6.4.2 Payments to the Designer during the construction Phase shall be made in equal monthly installments for the duration of the construction Phase. The amount of each payment shall be determined by dividing 95% of the fee for Construction Phase/Final Completion as stipulated in Attachment A by the number of months between the Notice to Proceed and the scheduled issuance of the Certificate of Substantial Completion as indicated in the Project Schedule as approved by the Owner. The Designer shall be entitled to Extra Services in accordance with Article 8.3 should the Project be delayed beyond the 60-day period described in Article 8.3 for reasons beyond the control of the Designer.
- 6.5 <u>Final Installment:</u> The Designer shall be paid the unpaid balance of the fee for Construction Phase/Final Completion as stipulated in Attachment A (as that fee may be amended), upon compliance with the following requirements:
 - 6.5.1 Approval of the Certificate of Final Completion of construction (such Certificate to be in the form developed by the Authority). In cases where a Certificate of Partial Release of Retainage is approved, the Designer shall be paid up to an amount commensurate with the percent of retainage released until a Certificate of Final Completion is approved; and
 - 6.5.2 Delivery by the Designer to the Owner of the Record Drawings required by this Contract; and
 - 6.5.3 Verification of payment to MBE/WBE Subconsultants or Subconsultants identified on Attachment C and as required by Article 17.4; and
 - 6.5.4 A written evaluation of the General Contractor or CM at Risk by the Designer from which the Owner shall be able to complete its submission of the Contractor Evaluations as required by M.G.L. c.149 § 44D(7).
 - 6.5.5 In the event that the Designer is unable to comply with items 6.5.1 and 6.5.2 above due to reasons beyond the Designer's control, as determined by the Owner, Final Installment shall not be unreasonably withheld or delayed beyond 60 days after the date of Substantial Completion, provided that the Designer has complied with all other requirements.

6.6 Substantial Change

- 6.6.1 If there is a substantial change in the services described in the RFS to be provided by the Designer under this Contract, the Designer and the Owner will mutually agree to a written amendment describing the services and an amended Fee for Basic Services to reflect the change and reasonable cost of such change. Such changes shall be designated on Attachment F and shall be executed by the Designer and the Owner.
- 6.6.2 Should the Designer and the Owner be unable to negotiate a mutually acceptable amendment to the Fee for Basic Services when there has been a substantial change in the specified services, the Owner shall unilaterally and promptly determine, in good faith and supported by a written explanation in sufficient detail, a reasonable maximum dollar amount for the services as amended and process payments to the Designer subject to said maximum amount, until an amendment to the Fee for Basic Services for such change is set by later agreement between the parties, provided, that the Designer's acceptance of such payments shall not be considered a waiver by the Designer of its right to pursue a claim for additional compensation related to the change in services, and provided that such disagreement shall be immediately submitted to mediation in accordance with paragraph 18.5(b). In no event shall the Designer stop work under this Contract due to a disagreement with the Owner regarding an amendment in the Designer's Fee for Basic Services, provided that the Owner complies with its payment obligations under this Article 6.6.
- 6.6.3 Notwithstanding the foregoing, the amendment to this Agreement described in paragraph 7.4.8 shall be negotiated and executed by both parties prior to the start of the subsequent Phase.

ARTICLE 7: BASIC SERVICES

- 7.1 The Designer shall discuss with the Owner and the Authority the requirements for each Phase before beginning work on that Phase.
- 7.2 The Owner and the Authority will promptly review and approve the Designer's submittals. Upon completion of its review, the Owner shall promptly and in writing:
 - (a) approve the submittal as made; or
 - (b) approve that part of the submittal that is acceptable and reject the remainder; or
 - (c) reject the submittal; or
 - (d) require the Designer to submit additional information or details in support of its submittal.
 - 7.2.1 The description of Designer Services required during the various Phases as described in the RFS and hereinafter may include specification of the number of submittals the Designer will be required to make and estimates of the approximate number of meetings that the Designer will be required to prepare for and attend during each Phase.

- 7.2.2 As a part of Basic Services, the Designer shall provide six copies of each submittal to the Owner; two copies of each submittal to the Authority, and, if the Owner elects to proceed with the CM at Risk construction delivery method, one copy of each submittal to the CM at Risk. Drawings submitted to the Authority shall be reproduced at half full size. A graphic scale shall be placed upon all such drawings prior to construction documents phase submittals. If the Designer is required to make submittals in excess of the number specified or if the Designer is required to prepare for and attend meetings in excess of the number specified for a Phase, the Designer shall be entitled to compensation for Extra Services, provided, however, that the Designer shall not be entitled to such compensation if and to the extent the Owner or the Authority shall have reasonably determined that the additional submittals or the additional meetings were required due to either the Designer's lack of preparation, or other fault due to deficiencies or omissions in documents prepared by the Designer.
- 7.2.3 All document submittals shall be in the form of neatly bound printed material, and delivered to the location or locations as indicated by the Owner and Authority. One or more document submittal components may be submitted in an approved electronic format, subject to specific authorization by the Owner and/or Authority.
- 7.2.4 Electronic Submittals: In addition to all other submittals called for by this Article 7 and elsewhere in the Contract, including but not limited to hard copies and reproducibles of all submittals, the Designer shall submit two (2) electronic copies on compact disks for all required submissions of Deliverables called for by this Contract ("Electronic Submittals"). All Electronic Submittals shall be deemed to be Materials that are subject to all provisions of Article 16. The Electronic Submittals shall be provided on CD electronic format as approved by the Owner and Authority and as follows:
 - (a) All drawings shall be provided in standard AutoCAD software (release number and version to be established at time of contract execution) or in a compatible electronic CADD (.dxf) format or other industry-standard format as approved by the Owner and acceptable to the Authority. Electronic file naming convention shall be acceptable to the Owner and the Authority.
 - (b) All other documents shall be provided in pdf format, Microsoft Word, Excel, Project, or PowerPoint, as applicable to the particular submittal.
 - (c) All submittals shall be labeled identifying project name and number, file name, drawing title, software and release, and layering system.
 - (d) The Owner reserves the right to require the Designer to provide all electronic media as may be required at any time during the duration of this Contract due to technology upgrades and/or changes to the electronic systems used by the Owner or Authority, provided that if such requirement demands that the Designer

- purchase new software or train existing employees for the application of media or software such costs shall be a Reimbursable Expense but only to the extent that such purchase of new software or training of existing employees is unique or exclusive to the particular requirements of the Owner or the Authority for this particular Project.
- (e) The Designer's compliance with the terms of this Article shall be performed as part of the Basic Services under the Contract, and the Designer shall not receive any additional compensation for providing the Electronic Submittals, (including but not limited to conversions or copies of software), except as specified herein. The Designer shall not be responsible for any use of Electronic Submittals on hardware or software for which it was not intended. Creation of a Building Information Model is excluded from the definition of Electronic Submittals; if the Owner requests the Designer to create such a Model, the parties shall execute a separate agreement and Designer shall receive Extra Services for its creation.
- 7.2.5 In reviewing and preparing all documents for evaluation as part of the Feasibility Study and/or any other design phase for which the Designer may be authorized, the Designer shall determine gross area and net areas in the following manner in order to maintain uniformity in computation and consistency of both gross and net square foot areas of buildings:

Gross Area: The area included within the outside faces of the exterior walls for all stories. Custodial areas such as janitor closets, building maintenance and building employees' locker rooms, circulation areas such as corridors, lobbies, stairs, and elevators, and mechanical areas such as those designated to house mechanical and electrical equipment, utility services, and non-private toilets shall be considered as part of the gross area, but not part of the net area.

<u>Net Areas:</u> In general, those areas which have a specific assignment and functional program use as determined by the facility, including, but not limited to, areas such as cafeterias, auditoriums, libraries, administrative and classrooms. These shall be measured from the inside finish of permanent outside walls to the inside finish of corridor walls, and to the inside finish of intermediate partitions.

7.3 Feasibility Study Phase:

7.3.1 The Designer shall familiarize itself with the Authority's Guidelines and Standards for feasibility studies that further specify the work to be performed by the Designer during this Phase and shall perform its Feasibility Study Phase services in accordance with such Guidelines and Standards and the provisions of this Contract. The Designer shall meet with the Owner to arrive at a mutual understanding of the

- requirements of the Feasibility Study. The Designer shall submit a proposed work plan including anticipated tasks and submittals.
- 7.3.2 The Owner is required to ascertain the Authority's input and approval throughout the study process; therefore, the Designer shall develop and prepare the documentation required by the Feasibility Study to assist the Owner in securing the Authority's concurrence and/or approval at the following milestones before proceeding to the next milestone (Note that some of the approvals to move to the next milestone require a vote of the Authority's Board of Directors):
 - (a) Preliminary design program;
 - (b) Budget Statement for Educational Objectives, as defined by 963 CMR 2.02;
 - (c) Development of alternatives to be studied;
 - (d) Preliminary evaluation of alternatives;
 - (e) Final Evaluation of Alternatives;
 - (f) Recommendation to the Authority's Board of Directors of the preferred alternative that will be advanced to schematic design.
- 7.3.3 The Designer shall cooperate with the Owner and the Authority to define and develop a few reasonable, educationally sound, cost effective, and practical solutions for the Owner and Authority's evaluation that satisfy the Owner's educational program requirements that were provided by the Owner to the Designer. The alternatives considered shall address the following as a minimum:
 - (a) Analysis of school district student school assignment practices and available space in other schools in the district; and
 - (b) Tuition agreements with adjacent school districts (per M.G.L. c.70B §8); and
 - (c) Rental or acquisition of existing buildings that could be made available for school use. (per M.G.L. c.70B §8); and
 - (d) Renovation and/or addition to existing building(s) and related facilities or fields, if appropriate to the Project; and
 - (e) No-build or status quo option, to be used as a benchmark for comparative analysis of all other alternatives; and
 - (f) In some cases, it may also be appropriate to consider construction of new building and the evaluation of potential locations.
- 7.3.4 Feasibility Study submittals shall be provided pursuant to Article 7.2.2 and shall be subject to the written Approval of the Owner.

- 7.3.5 The Designer shall present and explain the Feasibility Study to the Owner and the Authority and at a local public meeting, if any such meeting is scheduled, or in conference.
- 7.3.6 The Designer shall meet with the Owner every other week during this Phase.

7.4 <u>Schematic Design Phase</u>

- 7.4.1 Upon receipt of an Approval to proceed to Schematic Design Phase, the Designer shall meet with the Owner to arrive at a mutual understanding of the requirements of the Final Design Program approved in writing by the Owner and the Authority.
- The Designer shall submit a proposed design work plan pursuant to this Contract including anticipated tasks and submittals. The Designer shall also submit to the Owner a proposed schedule consistent with any Project Schedule included in the RFS (Attachment B) modified as required by any subsequent schedule changes or delays outside of Designer's control. The schedule shall contain dates for submittals, deliverables, actions, milestones, design workshops, meetings and the critical path through all design service activities. It shall include time for the Owner's and the Authority's review and approval of submittals and for necessary submissions for permits in connection with the Project. The work plan shall also include a work plan schedule of values consistent with Attachment A, which shall be the basis for which payments of the Fee for Basic Services within each Phase shall be made. The work plan schedule of values shall identify deliverables within each Phase and percentages of the phase fee payable upon completion of such deliverable. When approved by the Owner as provided in Article 7.4.8, the work plan schedule of values shall govern the timing of payments of the Fee for Basic Services upon completion of deliverables within each Phase and as each Phase progresses.
- 7.4.3 The Designer shall: Prepare a preliminary evaluation of the Recommended Preferred Solution from the Feasibility Study, the Final Design Program, and Proposed Total Project Budget; collect and study all available drawings, reports, maintenance reports, and other existing data pertaining to the Project; conduct a thorough on-site review of conditions relating to the Project; assure that the "Recommended Preferred Solution" complies with all applicable codes and regulations, including any special design standards supplied by the Authority and its Commissioning Consultant; and meet with local building officials to identify and confirm applicable standards, codes and any project specific criteria.
- 7.4.4 The Designer shall develop the Recommended Preferred Solution to a full schematic design level. Schematic design level documentation shall be based on the Final Design Program, shall incorporate Owner and Authority comments and shall include each of the following, to the extent applicable to the Recommended Preferred Solution:
 - (a) Traffic Analysis analyze the impact of anticipated vehicular and pedestrian traffic, including impacts to existing infrastructure, to determine efficient and safe site access.

- (b) Environmental and Existing Building Assessment Provide additional site and building assessments as may be required to quantify presence of unsuitable materials and scope of possible remediation efforts.
- (c) Geotechnical and Geoenvironmental Analysis Provide additional geotechnical analysis as may be required to describe soil conditions, remediation requirements and appropriate foundation.
- (d) Program Analysis a space measurement analysis for the design which shall verify that the sum of all program floor areas plus all other floor areas equal the gross floor area of the Final Design Program.
- (e) Code Analysis Determine the impact of all applicable federal, state, regional and local codes, regulations and ordinances, including a listing of permitting and other regulatory filing requirements.
- (f) Utility Analysis Determine the availability and capacity of all required building utilities. Provide soils analysis and preliminary design for on-site septic/sewage treatment facilities, if required.
- (g) Massing Study an analysis of the building's integration into its surroundings and neighborhood with drawings, models, or photographs.
- (h) MA-CHPS or LEED-S Scorecard Pursuant to the Authority's Sustainable Building Design Guidelines complete a MA-CHPS or LEED-S for Schools Scorecard and describe sustainable design features and each high performance green school prerequisite and credit included in the proposed design and a plan for implementation or inclusion of any appropriate public utility energy conservation design programs.
- (i) Accessibility an analysis of the design's compliance with the Americans with Disabilities Act (ADA) and the Massachusetts Architectural Access Board requirements (MAAB).
- (j) Building Systems Descriptions Describe in narrative and on schematic plans basic information relative to:
 - 1. Building Structure a written narrative of the design approach to the structural systems including discussion of the feasible options for foundations and superstructure as well as treatment of special situations such as unusual soils conditions or long spans.
 - 2. Plumbing and HVAC written narratives of the basic systems and proposed fuel source(s) and a preliminary life cycle cost analysis pursuant to the criteria of M.G.L. c. 149 § 44(m). Provide schematic plans indicating basic distribution concepts and the location of major equipment items such as boilers, water heaters, cooling towers, chillers, air handling

- units, heat recovery units, exhaust stacks, and special systems (e.g. fume exhausts).
- 3. Fire Protection written narratives of the basic systems and design criteria. Provide schematic plans indicating basic distribution concepts and the location of major equipment items such as fire pumps, standpipes, and fire department connections.
- 4. Electrical (including power, lighting, communications, fire alarm, video/CATV, security/surveillance) written narratives of the proposed electrical and communications systems resources, needs, and proposed scope. Provide schematic plans indicating basic distribution concepts and the location of major equipment items such as switchgear, standby generator, and control centers/panels.
- 5. Information Technology written narratives of the proposed information technology system resources, needs, and proposed scope. Provide schematic plans indicating basic distribution concepts, and location of major equipment items such as switches and hubs.
- (k) Outline specifications in accordance with applicable CSI Divisions that clearly define the scope of construction, identify the sub-trades pursuant to M.G.L. c. 149 § 44F, establish the quality of materials, finishes, products, equipment and workmanship, and the special or unique conditions of construction.
- (1) Project Schedule Provide a reasonable level of design-related input to the OPM such that the OPM can prepare a draft schedule for the proposed project for the Owner in the form of a graphic representation (Gantt Chart) of the duration of all tasks, activities and phases of the design and construction processes against the progression of time up to a proposed occupancy date. Dependencies between activities and tasks will be delineated. Individual tasks and activities will be rolled up to the major project milestones. Provide input to the OPM regarding priority actions and activities that may have a major impact on the schedule. The OPM, not the Designer, is responsible for preparing and maintaining the draft and updated project schedule document, except as it pertains to the project design schedule developed under Article 7.4.2.
- (m)Construction cost estimate in Uniformat II Level 3 format with aggregated unit rates and quantities supporting each item. If independent cost estimates are prepared for the Owner by the OPM in this or subsequent phases, then the Designer shall work with the OPM to resolve such any differences in a cost reconciliation process and shall involve any relevant parties in such process.
- (n) Siting analysis, including content, traffic and access, topographic and utilities recognition.

- (o) Site Development Plan Site plan shall be at a minimum scale of 1 inch equals 40 feet and include property lines with bearings and distances, building setbacks, site acreage, wetlands information, proposed and existing topography, proposed and existing buildings and site features, floor and roof elevations for all buildings, proposed and existing utilities and utility connections, and emergency equipment access.
- (p) Schematic Building Floor Plans of all floors and roof at a minimum scale of 1/16" =1'-0" showing all elements of the building including overall dimensions, gross square footage of each floor and net square footage of each space, response to functional requirements of program, major and minor access, circulation, and room data sheets.
- (q) Schematic Exterior Building Elevations for all sides and orientations indicating all exterior finishes and fenestration.
- 7.4.5 Schematic design phase drawings, specifications, construction cost estimates and other submittals shall be subject to the written Approval of the Owner, which Approval shall not be unreasonably delayed, withheld, conditioned, or denied. Unless a lesser number is requested by the Owner, the Designer shall submit to the Owner for approval six (6) copies of schematic design drawings, specifications, cost estimates, and other submittals. Two (2) additional copies shall be submitted to the Authority by the Designer.
- 7.4.6 The Designer shall present and explain the Schematic Design to the Owner, the OPM and the Authority and at a local public meeting, if any such meeting is scheduled, or in conference.
- 7.4.7 The Designer shall meet with the Owner every other week during the Schematic Design Phase.
- 7.4.8 Prior to the issuance of an Approval to proceed to the Design Development Phase, the Designer and the Owner shall meet to finalize the design work plan, project schedule, and schedule of values described in Article 7.4.2, and they shall if necessary execute an amendment to the Contract to include all required modifications to govern the subsequent phases of the Designer's services.

7.4.9 Construction Delivery Method Evaluation and Selection

(a) The Designer shall assist the Owner in determining the appropriate construction delivery methodology for the Proposed Project. In providing such assistance, the Designer, in conjunction with the Owner's Project Manager, shall advise the Owner on the relative advantages and disadvantages associated with each of the construction delivery methods provided in M.G.L. Chapters 149 and 149A. The decision to pursue a particular construction delivery method shall be within the sole discretion of the Owner, subject to the approval of the Inspector General as provided in M.G.L. c. 149A, §4. The services provided by the Designer in assisting and advising the Owner in its determination of the

- appropriate construction delivery methodology shall be included in Basic Services.
- (b) If the Owner elects to construct the Project using the CM at Risk construction delivery method pursuant to M.G.L. c. 149A, and has obtained the approval of the Office of the Inspector General to do so, with the Approval of the Owner, this Contract shall be amended using the Authority's Standard Amendment for CM-R which includes Articles 7.5 through 7.10. If the Owner elects to construct the Project using the Design-Bid-Build ("DBB") construction delivery method pursuant to M.G.L. c. 149, with the Approval of the Owner, this Contract shall be amended using the Authority's Standard Amendment for DBB, which includes Articles 7.5 through 7.9.
- 7.5 INTENTIONALLY OMITTED
- 7.6 INTENTIONALLY OMITTED
- 7.7 <u>INTENTIONALLY OMITTED</u>
- 7.8 INTENTIONALLY OMITTED
- 7.9 <u>INTENTIONALLY OMITTED</u>
- 7.10 <u>INTENTIONALLY OMITTED</u>

ARTICLE 8: EXTRA SERVICES

8.1 General

- 8.1.1 Extra Services are those services requested by the Owner to be performed by the Designer but which are additional (or "extra") to the services performed as Basic Services. Such services are not included in the Fee for Basic Services and shall be invoiced and paid for separately. Extra services shall not be deemed authorized until a written Approval is received from the Owner, which Approvals shall not be unreasonably delayed, withheld, denied, or conditioned.
- 8.1.2 The proposed cost, scope and schedule of all Extra Services shall be presented and approved by the Owner in writing prior to the performance of any Extra Services.
- 8.1.3 Cost proposals for Extra Services shall be computed in accordance with Attachment A.
- 8.2 Unless specifically stated elsewhere and only with the prior written Approval of the Owner, the Designer shall perform any of the following services as Extra Services:
 - 8.2.1 preparing measured drawings and detailed construction investigations documentation for existing buildings when such documentation does not exist;
 - 8.2.2 substantially revising previously approved reports, drawings, specifications or other documents to address changes authorized or requested by the Owner, including

substantial changes in its size, quality, complexity, design, Budget, and/or bidding method or bid packages, and changes in Applicable Laws;

- (a) Notwithstanding the provisions of 8.2.2, revisions prepared by the Designer to keep construction costs within the Project Budget that are required pursuant to Article 4.10 of this Contract to be without additional compensation, or to correct incorrect items for which the Designer has responsibility, shall not be Extra Services;
- 8.2.3 preparing documents for bidding alternates requested by the Owner, except for a reasonable number and extent of alternates to keep construction costs within the Project Budget which shall be Basic Services;
- 8.2.4 revising Construction Contract Documents which have been initially submitted and approved in their final and complete form, if general bids (Chapter 149) or subcontractor bids (Chapter 149 or 149A) for work required thereunder are not advertised based on such Construction Contract Documents within four months after initial submission:
- 8.2.5 services in connection with rebidding if the need to rebid is not attributable to the Designer;
- 8.2.6 attending meetings with the Owner, Owner's Project Manager, the Authority, Department of Labor and Workforce Development, the Office of Attorney General, the Office of the Inspector General, or the CM at Risk (if the project is constructed pursuant to M.G.L. c. 149A) in matters of dispute if attendance is required by the Owner, provided such dispute did not arise due to the fault of the Designer;
- 8.2.7 furnishing other services in excess of Basic Services made necessary by the default or failure of performance of the General Contractor or CM at Risk or Subcontractors;
- 8.2.8 providing consultation with respect to replacement of work damaged by fire or other casualty during construction;
- 8.2.9 preparing change orders and supporting data in accordance with Article 10, or modifying the Construction Documents in response to an unreasonable amount of substitutions proposed by the Contractor or CM at Risk, or responding to unreasonable and excessive requests for information (RFIs) by the Contractor or CM at Risk, where such information is available from a careful study and review of the Construction Documents;
- 8.2.10 assisting the Owner in litigation or claims arising out of the Owner-Contractor Agreement or Owner-CM at Risk Agreement, provided such litigation or claims did not arise due to the fault of the Designer;
- 8.2.11 performing services during a construction period extended beyond the additional 60 calendar day period, specified in Article 8.3;

- 8.2.12 performing professional services which are not otherwise required under this Contract as Basic Services;
- 8.2.13 providing services in connection with partial completion or partial systems completion inspections at the time of Substantial Completion of the Work or of a project construction phase and/or separate bidding package due to delay by the Contractor or CM at Risk in completing the Work on schedule;
- 8.2.14 providing services in connection with Contractor, CM at Risk or Bidder disputes or questions arising out of the bidding process, unless such protest is a result of an act or omission of the Designer. Such services include research and preparation for and appearance at bid protest hearing and similar proceedings.
- 8.3 Construction Phase Services Provided after the Original Construction Completion Date
 - 8.3.1 If construction of the Work, or of a project construction phase and/or separate bidding package has not reached substantial completion within the original construction period (as set forth in the Owner-Contractor or Owner-CM at Risk Agreement and as agreed to by the Designer), there shall be added to said construction period a period of sixty (60) calendar days, during which period the Designer shall continue to provide construction phase services for which no extra compensation shall be paid for the services described in Article 7.9 and 7.10.1 through 7.10.4 in a CM at Risk Project or for the services described in Articles 7.8 and 7.9.1 through 7.9.4 in a DBB Project.
 - 8.3.2 If construction has not reached Substantial Completion after the 60 additional calendar days, the Designer shall thereafter be entitled to Extra Services compensation for providing the services described in Articles 7.10.3 (which are fully defined under Article 7.9.2) and 7.10.4 in a CM at Risk Project or for the services described in Articles 7.9.3 (which are fully defined under Article 7.8.2) and 7.9.4 in a DBB Project. The Designer may also be entitled to Extra Services compensation for tasks performed beyond the added sixty (60) calendar days period for tasks related to Article 7.9.1 (d) through (i) in a CM at Risk Project or 7.8.1(d) through (i) in a DBB Project. In any event, the Designer is required to identify and present the anticipated Extra Services contemplated under Article 8.3.2 in accordance with Article 8.1. In no event shall the Designer be entitled to any additional compensation on account of an extended construction period if and to the extent that a binding agreement or decision that results from a dispute resolution proceeding determines that the Designer's acts or inactions caused the construction period to be extended.
- 8.4 In the event of an emergency the Designer may proceed to perform Extra Services as required to meet the emergency after obtaining the verbal approval of the Owner. The Designer shall provide a written report to the Owner, as soon after the emergency arises as possible, and such report shall describe the emergency and the Extra Services that were performed.
- 8.5 Invoices for Extra Services shall be accompanied by a breakdown listing the name, payroll title, date, number of hours by day, hourly rate and extended amount, per specified task of

Extra Services performed. Hourly rates shall be in accordance with the Hourly Rate Schedule in Attachment A.

ARTICLE 9: REIMBURSABLE EXPENSES

- 9.1 For coordination and responsibility for the services, materials and costs described in 9.1.1 through 9.1.6, the Designer shall be reimbursed its actual costs and those of its Subconsultants, supported by invoices or receipts, plus 10%. The following are reimbursable expenses, when authorized by the Owner:
 - 9.1.1 The actual cost to the Designer for Subconsultants and for additional tests under 4.11 provided, however, that reimbursement for such costs shall not be made unless the rates of compensation, the total estimated cost of the services and the scope of work for said services shall have been previously approved in writing by the Owner.
 - 9.1.2 The cost of printing more than nine (9) sets of design submittals for a CM at Risk project, or more than eight (8) sets of design submittals for a project pursuant to G.L.c. 149, or more than two electronic versions thereof per design submission deliverable phase or sub-phase.
 - 9.1.3 The cost of printing the bid documents and the related copying, postage, and handling services during a prequalification or bid period.
 - 9.1.4 The cost of reproducing the mylar reproducibles of the construction drawings for use by the General Contractor or CM at Risk in preparing the record drawings.
 - 9.1.5 Out of pocket expenses paid by the Designer such as filing fees, testing, and permit fees if such fees would be normally paid by the Owner.
 - 9.1.6 Renderings, models, mock-ups, photographs and any other presentation materials.
 - 9.1.7 Other expenses deemed necessary or appropriate by the Owner in writing.
- 9.2 <u>Non-Reimbursable Expenses:</u> The Owner shall not reimburse the Designer or its Subconsultants for travel expenses, sustenance, telephone, copying, facsimiles, electronic mails, postage and delivery expenses or cost estimating, unless specifically required elsewhere in this Contract.
- 9.3 The Designer shall not be entitled to compensation under this Article for the services of Subconsultants hired to perform Basic Services under this Contract.

ARTICLE 10: COMPENSATION AND RESPONSIBILITY FOR CHANGE ORDERS

10.1 The Designer shall be entitled to Extra Services compensation for preparing Change Orders initiated by the Owner except as provided in Article 10.3.

- 10.2 The Designer shall not be entitled to Extra Services compensation for preparing Change Orders to adjust the scope of construction work which arises from existing conditions for which unit prices have been specified in the Construction Contract Documents.
- 10.3 The Designer shall not be entitled to Extra Services compensation for preparing Change Orders necessary to address errors or omissions by the Designer.
- 10.4 Change Orders for which the Designer is not entitled to compensation are to be referred to as "no fee change orders."
- 10.5 The fact that the Designer is not entitled to compensation for preparing a Change Order shall not limit any legal remedies which the Owner may have for recovering its additional costs necessitated by the Change Order.

ARTICLE 11: RELEASE AND DISCHARGE

11.1 The acceptance by the Designer of the last payment under the provisions of Article 6.5 or Article 12 in the event of termination of the Contract, shall in each instance, operate as and be a release to the Owner and the Authority and their employees and officers, from all claims of the Designer and its Subconsultants for payment for services performed and/or furnished, except for those written claims submitted by the Designer to the Owner with, or prior to, the last invoice.

ARTICLE 12: ASSIGNMENT, SUSPENSION, TERMINATION, NO AWARD

12.1 Assignment:

12.1.1 The Designer shall not assign or transfer any part of its services or obligations under this Contract (other than as specified in this Article 12), without the prior written approval of the Owner and the Authority. Likewise, any successor to the Designer must first be approved by the Owner and the Authority before performing any services under this Contract. Such written consent shall not in any way relieve the Designer or its assignee from its responsibilities under this Contract. The Owner shall not assign this Contract without the written consent of the Designer.

12.2 Suspension:

- 12.2.1 The Owner may, at any time, effective upon fifteen (15) business days written notice to the Designer, suspend this Contract. If the Owner provides such written notice, the Designer shall be compensated for Services satisfactorily performed in accordance with the Contract terms prior to the effective date of such suspension; invoices for such Services shall be properly submitted, but may be submitted after the date of such notice up to the effective date of suspension.
- 12.2.2 If a written notice of suspension issued pursuant to sub-paragraph 12.2.1 lasts for more than 90 consecutive calendar days, the Designer may, upon resumption of the Contract, be entitled to additional compensation for actual costs incurred due to such suspension provided that the suspension was not attributable to the Designer's fault.

12.3 Termination:

- 12.3.1 (a) By written notice to the Designer, the Owner may terminate this Contract effective on five (5) calendar days notice without cause. All compensation and reimbursement due to the Designer in accordance with the Contract terms, for services satisfactorily performed up to the date of termination, including proportionate payment for portions of the services started but incomplete at the time of termination, shall be paid to the Designer, provided no payment shall be made for services not yet performed or for anticipated profit on unperformed services. (b) Owner may terminate this Contract effective on five (5) calendar days notice for cause, and no further payment shall be due to the Designer to the extent the Owner can reasonably identify damages in specific amounts for which the Designer is liable under this Contract; Owner shall pay other amounts otherwise due and owing to the Designer.
- 12.4 <u>Suspension or Termination by Designer</u>: By written notice to the Owner and the Authority, the Designer may suspend or terminate (at Designer's sole option) this Contract:
 - 12.4.1 if the Owner, within thirty (30) days following written notice from the Designer of any material default by the Owner under the Contract (including failure to pay in accordance with the Contract), shall have failed to cure such default; or
 - 12.4.2 if, after the Designer has performed all services required during any Phase prior to construction and at least three (3) months have elapsed without receipt by the Designer of Approval to proceed with the next Phase of the Project, provided the delay was not the fault of the Designer. This provision shall not apply to a Designer who has received a notice of suspension pursuant to 12.2.
 - 12.4.3 Upon a proper termination by the Designer, the Designer shall be compensated as provided in 12.3.1 above regarding termination without cause.
- 12.5 No Award of Owner-Contractor Agreement: If the Project is constructed pursuant to M.G.L. c. 149, §§ 44A-44M, the Owner-Contractor Agreement is not awarded by the Owner within one hundred twenty (120) days after the receipt of general bids for the Project and the bids have not been rejected and the Project has not been suspended, the Designer shall be paid through the Bidding Phase as if a contract for construction were awarded according to the payment schedule provided in Attachment A. This Article 12.5 does not apply, however, if the Designer has been directed to perform design revisions pursuant to 4.10.2, for the purposes of bringing the design of the Project within the Project Construction Budget.

ARTICLE 13: NOTICES

13.1 Any	notices required or permitted to be given hereunder shall be given in writing and shall
be d	elivered (a) in person (b) by certified mail, postage prepaid, return receipt requested (c)
by fa	acsimile or (d) by a commercial overnight courier that guarantees next day delivery and
prov	vides a receipt, and such notices shall be addressed as follows:

If to_[];
If to_[
If to [1

or to such other address as the Owner, Authority and Designer may from time to time specify in writing. Any notice shall be effective only upon delivery, which for any notice given by facsimile shall mean notice that has been received by the party to whom it is sent as evidenced by confirmation slip that bears the time and date of request.

ARTICLE 14: INDEMNIFICATION

- 14.1 For claims arising out or relating to negligent errors and omissions in the performance of professional services rendered by the Designer, to the fullest extent permitted by law, the Designer shall indemnify and hold harmless the Owner and its officers and employees from and against all claims, damages, liabilities, injuries, costs, fees, expenses, or losses, including, without limitation, reasonable attorney's fees and costs of investigation and litigation, whatsoever which may be incurred by the Owner to the extent caused by the negligence of, or the breach of this Contract by, the Designer or a person employed by the Designer, or Subconsultant for whom the Designer is responsible under this Contract.
- 14.2 For all other claims, to the fullest extent permitted by law, Designer shall defend, indemnify and hold harmless the Owner and the Authority and their officers and employees from and against all claims, damages, liabilities, injuries, costs, fees, expenses, or losses, including, without limitation, reasonable attorney's fees and costs of investigation and litigation, whatsoever which may be incurred by the Owner or the Authority to the extent they result from the performance of its services provided that such claims, damages, liabilities, injuries, costs, fees, expenses, or losses are attributable to bodily injury or death or injury to or destruction of tangible property and are caused by an act or omission of the Designer or a person or Subconsultant for whom the Designer is responsible under this Contract.

ARTICLE 15: INSURANCE

- 15.1 The Designer shall obtain and maintain at its sole expense all insurance required by law and as may be required by the Owner and by the Authority under the terms of this Contract. The insurance required hereunder shall be provided at the sole expense of the Designer or its Subconsultant, as the case may be, and shall be in full force and effect for the full term of the Contract between the Owner and the Designer or for such longer period as required under this Contract.
- 15.2 All policies shall be issued by companies lawfully authorized to write that type of insurance under the laws of the Commonwealth of Massachusetts with a financial strength rating of "A" or better as assigned by A.M. Best Company, or an equivalent rating assigned by a similar rating agency acceptable to the Owner and the Authority.
- 15.3 The Designer, and any of its Subconsultants, shall submit to the Owner originals of the required certificates of insurance simultaneously with the execution of this Contract. Certificates of insurance evidencing the coverage required hereunder, together with evidence that all premiums for such insurance have been fully paid, shall be filed with the

Owner and shall be made available to the Authority upon request. Certificates shall show each type of insurance, insurance company, policy number, amount of insurance, deductibles/self-insured retentions, and policy effective and expiration dates. The Designer shall submit updated certificates to the Owner prior to the expiration of any of the policies referenced in the certificates so that the Owner shall at all times possess certificates indicating current coverage and said certificates shall be made available to the Authority upon request. Failure by the Designer to obtain and maintain the insurance required by this Article, to obtain all policy renewals, or to provide the respective insurance certificates as required shall constitute a material breach of the Contract and shall be just cause for termination of the services of the Designer under this Contract.

- 15.4 Termination, cancellation, or modification or reduction of coverage or limits by endorsement of any insurance required by this Contract, whether by the insurer or the insured, shall not be valid unless written notice thereof is given to the Owner and the Authority at least thirty days prior to the effective date thereof, which shall be expressed in said notice.
- 15.5 The Designer or its Subconsultant, as the case may be, is responsible for the payment of any and all deductibles under all of the insurance required below. Neither the Owner nor the Authority shall be responsible for the payment of deductibles, self-insured retentions or any portion thereof.
- 15.6 Workers' Compensation, Commercial General Liability, Automobile Liability, and Valuable Papers
 - 15.6.1 The Designer shall purchase and maintain at its own expense during the life of this Contract, or such other time period as provided herein, the following types and amounts of insurance, at a minimum:
 - (a) Workers' Compensation Insurance in accordance with General Laws Chapter 152. The policy shall be endorsed to waive the insurer's rights of subrogation against the Owner and the Authority.
 - (b) Commercial General Liability Insurance (including Premises/Operations; Products/Completed Operations; Contractual; Independent Contractors; Broad Form Property Damage; and Personal Injury) with a minimum limit of \$1,000,000 per occurrence, \$2,000,000 aggregate. The Designer shall maintain such insurance in full force and effect for a minimum period of one year after final payment and shall continue to provide evidence of such coverage to the Owner and the Authority. The Owner and the Authority shall be included as an additional insured in this policy. The policy shall be endorsed to waive the insurer's rights of subrogation against the Owner and the Authority.
 - (c) Automobile Liability Insurance (including owned, non-owned and hired vehicles) at limits of not less than \$1,000,000 combined single limit per accident.

(d) Valuable Papers insurance in an amount sufficient to assure the restoration of any plans, drawings, computations, field notes, or other similar data relating to the work covered by the Agreement between the Owner and the Designer in the event of loss or destruction while in the custody of the Designer until the final fee payment is made or all data is turned over to the Owner, and this coverage shall include coverage for relevant electronic media, including, but not limited to, documents stored in computer-aided design drafting (CADD) systems.

15.7 Professional Liability

- 15.7.1 The Designer shall maintain professional liability insurance covering negligent errors and omissions and negligent acts of the Designer and of any person or entity for whose performance the Designer is legally liable at all times while services are being performed under this Contract and for a period of six years thereafter (as calculated in accordance with the terms below in this 15.7.2). The minimum amount of such insurance shall be \$2,000,000 per claim/\$2,000,000 annual aggregate.
- 15.7.2 If the policy is in a "claims made" format, it shall include a retroactive date that is no later than the effective date of this Contract, and an extended reporting period of at least six years after the earlier of: (1) the date of official acceptance of the completed Project by the Owner; (2) the date of the opening of the Project to public use; (3) the date of the acceptance by the general contractor or the CM at Risk of a final pay estimate prepared by the Owner pursuant to M.G.L. chapter 30; or (4) the date of substantial completion of the Owner-Contractor Agreement or Owner-CM at Risk Agreement and the taking of possession of the Project for occupancy by the Owner, which requirement can be met by providing renewal certificates of professional liability insurance to the Owner as evidence that this coverage is being maintained.

15.8 Subconsultants

- 15.8.1 The Designer shall require by contractual obligation, and shall exercise due diligence to enforce, that any professional engineering or landscape architecture Subconsultant hired in connection with the services to be provided under this Contract shall, unless otherwise agreed in writing by the Owner, obtain and maintain all insurance required by law and as may be required by the Owner under the terms of this Contract, except that the limit of Subconsultant's professional liability insurance shall be not less than \$2,000,000 per claim/\$2,000,000 annual aggregate.
- 15.8.2 All professional liability policies obtained by Subconsultants shall be issued by companies lawfully authorized to write that type of insurance under the laws of the Commonwealth of Massachusetts with a financial strength rating of "A" or better as assigned by A.M. Best Company, or an equivalent rating assigned by a similar rating agency acceptable to the Owner and the Authority.

- 15.8.3 If the Subconsultant's insurance policy is in a "claims made" format, it shall include a retroactive date that is no later than the effective date of its contract with the Designer, and an extended reporting period of at least six years after the earlier of:

 (1) the date of official acceptance of the completed Project by the Owner; (2) the date of the opening of the Project to public use; (3) the date of the acceptance by the General Contractor or CM at Risk of a final pay estimate prepared by the Owner pursuant to M.G.L. chapter 30; or (4) the date of substantial completion of the Owner-General Contractor Agreement or the Owner-CM at Risk Agreement and the taking of possession of the Project for occupancy by the Owner, which requirement can be met by providing renewal certificates of professional liability insurance to the Owner as evidence that this coverage is being maintained.
- 15.8.4 Other nonprofessional Subconsultants shall be required to maintain insurance in the types and amounts that they routinely carry in the course of their practice.

15.9 Liability of the Designer

Insufficient insurance shall not release the Designer from any liability for breach of its obligations under this Contract. Without limitation, the Designer shall bear the risk of any loss if its valuable papers insurance coverage is insufficient to cover the loss of any work covered by this Contract.

15.10 Asbestos and Hazardous Materials

- 15.10.1 Unless otherwise provided in the RFS, the Designer shall have no responsibility for the discovery, presence, handling, removal or disposal of or for the exposure of persons to oil or hazardous materials in any form at the Project, including but not limited to asbestos-containing materials or other hazardous materials, as defined in MGL c.21E §2.
- 15.10.2 In the event that the Designer employs the services of a sub-consultant to provide services related to either the testing for asbestos-containing materials or oil or hazardous materials or related to the specification of methods and procedures for the removal or remediation of such asbestos-containing materials or oil or hazardous materials, the Designer shall employ such Subconsultants who have liability insurance coverage covering such services, to the extent that such insurance coverage is generally available to Subconsultants. Upon the Owner's written request, the Designer shall assign to the Owner the Designer's contractual right to pursue a claim against such Subconsultants. Such services shall be paid for as provided in Article 9 Reimbursable Expenses unless such services are specifically included as Basic Services in the RFS.

ARTICLE 16: OWNERSHIP OF DOCUMENTS

16.1 Unless provided otherwise by law, ownership and possession of all information, data, reports, studies, designs, drawings, specifications, materials, computer programs, documents, models, inventions, equipment, and any other documentation, product of

tangible materials to the extent authored or prepared, in whole or in part, by the Designer pursuant to this Contract (collectively, the "Materials"), other than the Designer's administrative communications, records, and files relating to this Contract, shall be the sole property of, and shall vest in, the Owner and the Authority as "works made for hire" or otherwise, provided that the Owner complies with its payment obligations under this Contract. The Owner and the Authority will own the exclusive rights, worldwide and royalty-free, to and in all Materials prepared and produced by the Designer pursuant to this Contract, including, but not limited to, United States and International patents, copyrights, trade secrets, know-how and any other intellectual property rights, and the Owner and the Authority shall have the exclusive, unlimited and unrestricted right, worldwide and royaltyfree, to publish, reproduce, distribute, transmit and publicly display all Materials prepared by the Designer. The Owner and the Authority shall provide appropriate credit to the Designer, in terms agreed upon by the Design, in any publicity about or plaque at the Project. The Designer shall have a license to publish and publicly display all Materials prepared by the Designer in its normal marketing and related professional and academic activities. The Designer shall have a license to use the typical or standard details and all other replicable elements of the Materials for this Project on other future projects. At the completion or termination of the Designer's services required pursuant to this Contract, copies of all original Materials shall be promptly turned over to the Owner and the Authority.

16.2 The Owner and the Authority agree to waive any and all claims against the Designer and, to the fullest extent permitted by law, to jointly and severally defend, indemnify and hold the Designer harmless from and against any and all claims, losses, liabilities and damages incurred by the Owner or asserted by any other entity or individual arising out of or resulting from any use of the Materials on other projects, modifications of the Materials made by the Owner or others and used on this Project, or any reuse or modification of the Materials or any of Designer's designs, drawings and specifications. The Authority shall be a party to this Contract solely for the purposes of enforcing its rights and obligations under this Article 16.

ARTICLE 17: STATUTORY REQUIREMENTS

- 17.1 Agent for Service of Process: If the Designer's principal place of business is outside of the Commonwealth of Massachusetts, the Designer shall appoint an agent for the service of process as provided in M.G.L. c.227, §5. The power of attorney reflecting such appointment shall be filed with the Secretary of State as provided in M.G.L. c.227, §5. Copies of the power shall be provided to the Owner. There shall be no lapse in such agency for as long as the Designer may have potential liability.
- 17.2 Truth-in-Negotiations Certificate (M.G.L. c.7, §38H)
 - 17.2.1 If the Designer's fee has been negotiated, the Designer must file a truth-innegotiations certificate prior to execution of this Contract by the Owner. The certificate shall contain the following certifications:

- (a) that wage rates and other costs used to support the Designer's compensation are accurate, complete, and current at the time of contracting; and
- (b) that the Contract price and any additions to the Contract may be adjusted within one year of completion of the Contract to exclude any significant amounts if the Owner determines that the fee was increased by such amounts due to inaccurate, incomplete or noncurrent wage rates or other costs.
- 17.3 Certification Pursuant to M.G.L. c.7 §38H (e): In accordance with M.G.L. c.7 §38H(e), the person signing this contract certifies, as a duly authorized signatory of the Designer, that the Designer has not given, offered or agreed to give any person, corporation, or other entity any gift, contribution or offer of employment as an inducement for, or in connection with, the award of this Contract; no Consultant to or Subconsultant for the Designer has given, offered or agreed to give any gift, contribution or offer of employment to the Designer, or to any other person, corporation, or entity as an inducement for, or in connection with, the award to the Designer or Subconsultant of a contract by the Designer; and no person, corporation or other entity, other than a bona fide full-time employee of the Designer, has been retained or hired by the Designer to solicit for or in any way assist the Designer in obtaining this Contract upon an agreement or understanding that such person, corporation or other entity be paid a fee or other consideration contingent upon the award of this Contract.
- Minority-Owned and Woman-Owned Business Participation: Pursuant to M.G.L. c. 7, § 40N, the Designer shall subcontract a minimum of eight percent (8%) of its work to SOMWBA Certified minority-owned enterprises (MBEs) and four percent (4%) to SOMWBA Certified women-owned enterprises (WBEs) certified by the State Office of Minority-and-Women-Owned Business Assistance, SOMWBA, 10 Park Plaza Suite 3740, Boston, MA 02116; such percentages shall be based on the listed services defined and required in the RFS. If the Designer is a SOMWBA certified MBE or WBE the requirements in this Article 17.4 are not applicable.
 - 17.4.1 The Designer shall complete and submit at the time of contract execution a completed Participation Schedule which is attached to this contract as Attachment C in order to be in compliance with Article 17.4 above.
- 17.5 Accounting Requirements: The Designer shall cause to be maintained complete, accurate and detailed records of all time devoted to the Project by the Designer and each Subconsultant employed by the Designer. The Owner, the Authority, and the Commonwealth's Inspector General may at all reasonable times audit such records that directly pertain to this Contract. On a Contract where the Fee for Basic Services exceeds \$100,000 the Designer shall comply with M.G.L. c.30 §39R which requires the Designer to:
 - 17.5.1 Maintain accurate and detailed accounts for a six-year period after the final payment;
 - 17.5.2 File with the Owner annual audited financial statements or statements from their accountants that their reviews are consistent with state laws.

- 17.5.3 File with the Owner a statement of management on internal accounting controls on its letterhead as prescribed in Attachment D and a statement from an independent certified public accountant (CPA) on its letterhead as prescribed in Attachment E to this Contract.
- 17.6 Revenue Enforcement and Protection Program (REAP): Pursuant to M.G.L. c. 62C §49A, the undersigned certifies under the penalties of perjury that to the best of his/her knowledge and belief that the firm and/or individuals in the firm are in compliance with all laws of the Commonwealth of Massachusetts relating to taxes, reporting of employees and contractors, and withholding and remitting child support.
- 17.7 <u>Interest of Designer</u>: The Designer hereby certifies that it is in compliance with the provisions of M.G.L. c. 268A whenever applicable. The Designer covenants that 1) neither he/she nor any member of the Designer firm presently has any financial interest and shall not acquire any such interest direct or indirect, which would conflict in any manner or degree with the services required to be performed under this Contract or which would violate M.G.L. Chapter 268A, as amended from time-to-time; 2) in the performance of this Contract, no person having any such interest shall be employed by the Designer; and 3) no partner or employee of the Designer firm is related by blood or marriage to any officer, official, or employee of the Owner.
- 17.8 Equal Opportunity: The Designer shall not discriminate in employment against any person on the basis of race, color, religion, national origin, sex, sexual orientation, age, genetics, ancestry, disability, marital status, veteran status, membership in the armed forces, presence of children or political beliefs. Each shall comply with all provisions of Title VII of the Civil Rights Act of 1964 and MGL c.151B.
- 17.9 <u>Certification of Non-Collusion</u>: The signatory certifies under penalties of perjury that the Designer's proposal has been made in and submitted in good faith and without collusion or fraud with any other person. As used in this certification, the word "person" shall mean any natural person, business, partnership, corporation, union, committee, club, or other organization, entity, or group of individuals.

ARTICLE 18: MISCELLANEOUS

- 18.1 <u>Governing Law</u>: This Contract shall be governed by the laws of the Commonwealth of Massachusetts.
- 18.2 <u>Venue</u>: Any suit by either party arising under this Contract shall be brought only in the Superior Court in the county where the Project is located. The parties hereto waive any argument that this venue is improper or that the forum is inconvenient.
- 18.3 <u>Non-Waiver</u>: Neither the Owner's review, approval, or acceptance of, nor payment for any of the services furnished under this Contract shall be construed to operate as a waiver of any rights under the Contract or any cause of action arising out of the performance of the Contract.
- 18.4 <u>Entire Agreement</u>: This Contract represents the entire and integrated agreement between the Owner and the Designer and, except as otherwise provided herein, supersedes all

- prior negotiations, representations or agreements, either written or oral. This Contract may be amended only by written agreement signed by both the Owner and the Designer, and approved by the Authority, which approval shall not unreasonably be delayed, denied, conditioned, or withheld.
- Dispute Resolution: If a dispute arises between the parties related to this Contract, the parties agree to use the following procedures to resolve the dispute: (a) Negotiation. A meeting shall be held between representatives of the parties with decision-making authority regarding the dispute to attempt in good faith to negotiate a resolution of the dispute; such meeting shall be held within fourteen calendar days of a party's written request for such a meeting; (b) Mediation. If the parties fail to negotiate a resolution of the dispute, they shall submit the dispute to mediation as a condition precedent to litigation and shall bear equally the costs of the mediation. The parties shall jointly appoint a mutually acceptable mediator; they shall seek assistance from an independent third party in such appointment if they have been unable to agree upon such appointment within 30 days of the meeting just noted in (a) above; (c) Litigation. If the parties fail to resolve the dispute through mediation, then either party may file suit in accordance with Article 18.2; and (d) This Article of dispute resolution provisions shall survive termination of this Contract.
- 18.6 Waiver of Subrogation: (a) To the extent damages are covered by property insurance, the Owner and the Designer waive all rights against each other and against the General Contractor or CM at Risk, Subcontractors, consultants, agents, and employees of the other for damages caused by fire or other causes of loss, except such rights as they may have to the proceeds of such insurance as set forth in the Owner-Contractor Agreement or Owner CM at Risk Agreement. The Owner shall require of the General Contractor or CM at Risk, Subcontractors, Owner's Project Manager, consultants, Subconsultants, and agents and employees, by appropriate agreements, written where legally required for validity, similar waivers each in favor of other parties enumerated herein. The policies shall provide such waivers of subrogation by endorsement or otherwise. A waiver of subrogation shall be effective as to a person or entity even though that person or entity would otherwise have a duty of indemnification, contractual or otherwise, did not pay the insurance premium directly or indirectly, and whether or not the person or entity had an insurable interest in the property damaged. (b) Nothing in this Contract shall create a contractual relationship with or create a cause of action in favor of a third party against the Owner or the Designer.

ATTACHMENT A

PAYMENT SCHEDULE

Payments shall be made in accordance with the provisions outlined in the Contract and with the following schedule:

Basic Services
Feasibility Study Phase
Schematic Design Phase
Design Development Phase
Construction Documents Phase
Early Bid Packages
Bidding Phase
Construction Administration Phase
Completion Phase
TOTAL
Extra Services
Extra Services provided pursuant to Article 8 shall be compensated as determined by the Owner (a) by a lump sum fee agreed upon in advance in writing by the Owner and the Designer, or (b) on an hourly basis in accordance with the lesser of \$150 per hour or the rate schedule set forth below for time expended, or (c) on an hourly basis in accordance with the lesser of \$150 per hour or a multiple of 2.5 times the direct personnel expense (without benefits) of the Designers or Subconsultants personnel including principals.
Hourly Rates:

ATTACHMENT B

REQUEST FOR DESIGNER SERVICES (RFS)

INSTRUCTIONS FOR COMPLETING THE REQUEST FOR DESIGNER SERVICES

This model Request For Designer Services ("RFS") is intended for use in the procurement of a Designer by cities, towns, and regional school districts that have been invited by the Massachusetts School Building Authority (the "MSBA") to conduct a feasibility study or that have been approved for a project by the MSBA. The MSBA Designer Selection Panel has jurisdiction over the procurement of designers, programmers and entities by cities, towns, regional school districts, and independent agricultural and technical schools seeking funding from the MSBA for public school construction projects whose estimated construction cost is anticipated to be \$5,000,000 or greater. Designer selection for public school construction projects whose estimated construction cost is less than \$5,000,000 must be conducted pursuant to Massachusetts General Laws, Chapter 7, Section 38K by the respective city, town, regional school district or independent agricultural and technical school. A copy of the MSBA Designer Selection Panel's Procedures are attached to this Model RFS as Attachment E.

Unless otherwise approved by the MSBA in writing, a city, town, or regional school district must use this model RFS in the procurement of a Designer in order to qualify for MSBA funding. Each city, town, and regional school district shall be responsible for inserting project and district specific information where indicated in the RFS. Although this model RFS is intended to be comprehensive in meeting the MSBA's requirements for the procurement of a Designer, each city, town and regional school district shall be solely responsible for ensuring that its particular RFS complies with all applicable provisions of federal, state, and local law, including, but not limited to, all procurement laws. The MSBA recommends that each city, town, and regional school district have its legal counsel review its RFS to ensure that it is in compliance with all provisions of federal, state and local law prior to its publication. No addition, deletion or revision to the model RFS of any kind shall be valid unless approved in advance by the MSBA in writing. The MSBA's approval of an RFS is solely for the purpose of determining whether the proposed RFS appears consistent with the MSBA's guidelines and requirements for designer selection and is not for the purpose of determining whether the proposed RFS meets any legal requirements imposed by federal, state or local law, including, but not limited to, public procurement laws. The MSBA shall not be responsible for any legal fees or costs of any kind that may be incurred by a city, town or regional school district in relation to its preparation or review of its RFS.

- 1) Each city, town and regional school district ("Owner") shall provide the project specific information in the areas noted by italics and bold-face lettering or as indicated by the shaded and text box areas.
- 2) The Owner should contact the MSBA prior to commencing completing the RFS model to discuss the use of MSBA documents and the Owner's procurement schedule.
- 3) Prior to placing the advertisement, the Owner must submit a red-lined version of the final RFS indicating any and all additions, deletions or revisions to the model RFS for MSBA approval.
- 4) Standard attachments included with the RFS submittal do not need to be submitted as part of the red-lined version. However, any attachments added by the Owner should be included with the Owner's red-lined version.
- 5) The Owner should allow a minimum of 10 business days for MSBA review of the RFS. Actual review time may vary.
- 6) Upon advertisement of the RFS, the Owner is responsible for sending the final RFS, all attachments and the advertisement in electronic format to the MSBA.
- 7) A copy of the final RFS and the advertisement must be submitted to the MSBA as part of the required documentation prior to the scheduled Designer Selection Panel meeting.

This model Request For Designer Services ("RFS") is intended for use in the procurement of a Designer by cities, towns, and regional school districts that have been invited by the Massachusetts School Building Authority (the "MSBA") to conduct a feasibility study or that have been approved for a project by the MSBA. Unless otherwise approved by the MSBA in writing, a city, town, or regional school district must use this model RFS in the procurement of a Designer in order to qualify for MSBA funding. Each city, town, and regional school district shall be responsible for inserting project and district specific information where indicated in the RFS. Although this model RFS is intended to be comprehensive in meeting the MSBA's requirements for the procurement of a Designer, each city, town and regional school district shall be solely responsible for ensuring that its particular RFS complies with all applicable provisions of federal, state, and local law, including, but not limited to, all procurement laws. The MSBA recommends that each city, town, and regional school district have its legal counsel review its RFS to ensure that it is in compliance with all provisions of federal, state and local law prior to its publication. No addition, deletion or revision to the model RFS of any kind shall be valid unless approved in advance by the MSBA in writing. The MSBA's approval of an RFS is solely for the purpose of determining whether the proposed RFS appears consistent with the MSBA's guidelines and requirements for designer selection and is not for the purpose of determining whether the proposed RFS meets any legal requirements imposed by federal, state or local law, including, but not limited to, public procurement laws. The MSBA shall not be responsible for any legal fees or costs of any kind that may be incurred by a city, town or regional school district in relation to its preparation or review of its RFS.

REQUEST FOR DESIGNER SERVICES (RFS)

[Town][City] of _______, MA
Public Schools

[Name of Project]
. 2008

Invitation: The (City/Town/Regional District) of ______ ("Owner") is seeking the services of a qualified "Designer" within the meaning of M.G.L. Chapter 7, Section 38A½, to provide professional design and construction administration services for the ______ School in ______, Massachusetts. Selection of a Designer will be made by the Designer Selection Panel of the Massachusetts School Building Authority ("MSBA") in accordance with the MSBA's Designer Selection Procedures.

The Owner is seeking design services to conduct a Feasibility Study which will include the development and evaluation of potential alternative solutions and continue through the Schematic Design Phase of the preferred alternative initially. Subject to the approval of a Project by the MSBA and further subject to adequate funding authorized by the Owner, the contract between the Owner and the Designer may be amended to include continued designer services through design development, construction contract documents, bidding, award of construction contract(s), construction administration, final closeout and warranty period of the potential Project. A potential Project may include a renovation of the existing school, a renovation of and addition to the existing school and/or new construction and may be comprised of multiple contract packages.

Pursuant to M.G.L. Chapter 7, Section 40N, the Designer must agree to contract with minority and women-owned businesses as certified by the State Office of Minority and Women Business Assistance (SOMWBA). The amount of participation that shall be reserved for such enterprises shall not be less than eight percent (8%) of the contract price for minority business enterprises and four percent (4%) of the

contract for women-owned business enterprises. The minority and women-owned business enterprises must be selected from those categories of work identified in Item F of this RFS.

For additional information on Designer qualifications see Sections E. and F. in this RFS.

A. Background:

(Provide background information regarding the City or Town or District, School Building Committee structure, District's grade configuration, school facility inventory and/or any other information that may be helpful to understand the context of the potential project.)

{Provide specific information regarding the identified school including, but not limited to, total square footage, site information, age of building, building conditions and problems, current grade structure and enrollment.}

B. Project Goals and General Scope:

On or about (date), the Owner submitted a Statement of Interest (Attachment A) to the MSBA for (Identify prioritized school). The MSBA is an independent public authority that administers and funds a program for grants to eligible cities, towns, and regional school districts for school construction and renovation projects. The MSBA's grant program is discretionary, and no city, town, or regional school district has any entitlement to any funds from the MSBA. At the _______, 2008 Board of Directors meeting, the MSBA Board voted to issue an invitation to the Owner to conduct a feasibility study for this Statement of Interest to identify and study possible solutions and, through a collaborative process with the MSBA, reach a mutually-agreed upon solution. The MSBA has not approved a Project and the results of this feasibility study may or may not result in a Project approved by the MSBA.

It is anticipated that the feasibility study will review the problems identified in the Statement of Interest at the _____(Identify prioritized school)

The Feasibility Study shall include a study of all alternatives and contain all information required by 963 CMR 2.10(8) and any other applicable rules, regulations, policies, guidelines and directives of the Authority, including, but not limited to, a final design program, space summary, budget statement for educational objectives, and a proposed total project budget. The Schematic Design shall include, but not be limited to, the information required by the Authority's Feasibility Study Guidelines, including, but not limited to, a site development plan, environmental assessment, geotechnical assessment, geotechnical analysis, code analysis, utility analysis, schematic building floor plans, schematic exterior building elevations, narrative building systems descriptions, MA-CHPS or LEED-S for Schools scorecard, outline specifications, cost estimates, project schedule and proposed total project budget.

Project objectives under consideration by the Owner include:

(Some examples of objectives are shown below. These may or may not apply to this Request for Services and/or the Owner may have others.)

- Identification of community concerns that may impact study options;
- Identification of specific milestone requirements and/or constraints of the District e.g. Town votes, swing space, occupancy issues;
- Life cycle costs of operating the School as it relates to future operational budgets;
- Massachusetts High Performance Green Schools Guidelines (MA-CHPS or LEED-S for Schools Guidelines);
- CM-at-Risk Delivery Method.

C. Scope of Services:

The required scope of services is set forth in the MSBA's standard Contract for Designer Services (Contract) and standard Contract amendments for both Design/Bid/Build ("DBB") and a CM at Risk projects, copies of which are attached hereto and incorporated herein by reference. If the Owner determines to use the CM-at-Risk construction delivery method, the Contract must be amended using the MSBA's standard Contract amendment for CM at Risk. If the Owner determines to use a DBB construction delivery method, the Contract must be amended using the MSBA's standard Contract amendment for DBB. Unless specifically excluded, the Designer's Basic Services consist of the tasks described in the MSBA's standard Contract for Designer Services, standard Contract amendments, and this RFS including all investigative work (to the extent provided for in the Contract), feasibility study, schematic design, and, at the Owner's option, design work, preparation of construction documents, bidding period administration, construction administration, and other related work reasonably inferred in the opinion of the Owner and the Authority as being necessary to meet the project's stated scope and goals.

This RFS will be appended to and become part of the Contract for Designer Services. Any Designer selected as a result of this RFS will be required to execute the Contract for Designer Services that is attached hereto. Designers submitting an application in response to this RFS must specify any exceptions to the Contract at the time of application. The Owner may consider any such exceptions but shall not be bound by any such exceptions. A failure to specify exceptions will be deemed an acceptance of the Contract's terms and conditions.

The MSBA standard Contract For Designer Services, Article 16, requires a minimum of \$2,000,000 of professional liability insurance. The Owner may determine that due to the complexity and risk factors associated with the project that a higher level of professional liability coverage may be required. If so, the Owner should identify these additional insurance requirements in the RFS. See suggested sentence:

In lieu of the minimum professional liability insurance specified in Article 15, the successful Respondent will be required to provide a certificate of professional liability insurance, at the time of contract execution, indicating minimum coverage in the amount of $\gamma_{\mu\nu}$ per occurrence, $\gamma_{\mu\nu}$ aggregate.

Basic Services include, but are not limited to, verification of existing record information including building dimensions, details and general existing conditions, cost estimating, architecture, civil, sanitary, mechanical, electrical, plumbing, fire protection, structural, site planning and landscape architecture, basic environmental permitting, graphics, lighting design, acoustics, data and communication, educational consultants, any specialty consultants for laboratory, library/media center and kitchen space, code consultants, accessibility, energy evaluations, detailed cost estimates; preparation of construction documents; bidding and administering the Construction Contract Documents and other design and consulting services incidental and required to fulfill the project goals. Please refer to the Contract for a complete summary of Basic Services.

Extra and reimbursable expenses are defined in Articles 8 and 9 of the Contract in Attachment B.

The Owner should identify any available studies, drawings, surveys, photographs and subsoil exploration reports of the proposed project's existing buildings and site or sites.

The Owner should identify any of the services (basic, extra or reimbursable) identified in the Contract that are $\underline{\text{NOT}}$ applicable to this Project.

D. Project Phases and Work Plan:

Work under this RFS is divided into the Project Phases as listed in Article 7 of the Contract and as may be augmented in this RFS. Each Project Phase will consist of one or more required submissions, and may include site visits, meetings with the Owner, Owner's Project Manager, the Authority and others, and other tasks as described.

The estimated total duration of the Contract for Designer Services from Feasibility Study through the approval of Schematic Design, inclusive of review and approval time, is estimated to be 40 weeks as follows:

(The District should provide the estimated schedule for the preliminary program and the schematic design phase based on the project specifics.)

Preliminary Program through Final Design Program	26	weeks
Schematic Design Phase	14	weeks
Design Development through 100% CD	TBD	
Bidding	TBD	
Construction Administration Phase	TBD	weeks
Estimated Total Duration (Exclusive of Completion Phase)	TBD	weeks

The durations for the Bidding and Construction Administration Phases are estimates only. Actual durations may vary depending upon the agreed upon solution, the extent of required document revisions, the time required for regulatory approvals, and the construction contractor's performance.

Such variances in estimated time will not, in and of themselves, constitute a justification for an increased Fee for Basic Services, nor are they a substitute for the performance time requirements shown below.

The Designer performance times listed in the table below are <u>requirements</u>, <u>not</u> estimates. The Owner, through the Owner's Project Manager will review each submission and, if acceptable, provide notice to the Designer to proceed to the next phase.

The Designer's adherence to the performance times listed below will be part of the Owner's performance evaluation of the Designer's work, which will be conducted at the end of the Project.

	Within/Weeks				
•	Attend a "Kick-Off" meeting	2	Execution of a contract with the Owner		
•	Preliminary Program	4	Execution of a contract with the Owner		
•	Development of Alternatives	6	Execution of a contract with the Owner		
•	Preliminary Evaluation of Alternatives	4	Approval of Alternatives		
•	Final Evaluation of Alternatives	4	Approval of Preliminary Evaluation		
•	Recommendation of Preferred Solution	2	Approval of Final Evaluation		
•	Final Design Program	2	Approval of Preferred Solution		
•	Schematic Design	12	Approval of the Final Design Program		
•	Design Development	TBD	Approval of the Schematic Design		
•	60% Construction Documents	TBD	Approval of Design Development		
•	100% Construction Documents	TBD	Approval of Design Development		

E. Minimum qualifications:

Selection will be made by the MSBA Designer Selection Panel in accordance with the Authority's Designer Selection Procedures, attached hereto as Attachment E. The Respondent must certify in its cover letter that it meets the following minimum requirements. Any Respondent that fails to include such certification in its response, demonstrating that these criteria have been met, will be rejected without further consideration. To be eligible for selection, the Designer must meet <u>all</u> of the following qualifications.

- 1. Be a qualified Designer within the meaning of M.G.L. Chapter 7, Section 38A½, employing a Massachusetts registered *[architect][engineer]* responsible for and being in control of the services to be provided pursuant to the Contract.
- 2. The Project Architect/Engineer for the Designer has successfully completed the Massachusetts Certified Public Purchasing Official Program seminar "Certification for School Project Designers and Owner's Project Managers" as administered by the Office of the Inspector General of the Commonwealth of Massachusetts.
- 3. Pursuant to M.G.L. Chapter 7, Section 40N, the Designer must agree to contract with minority and women-owned businesses as certified by the State Office of Minority and Women Business Assistance (SOMWBA). The amount of participation that shall be reserved for such enterprises shall not be less than eight percent (8%) of the design contract price for minority business enterprises and four percent (4%) of the design contract for women-owned business enterprises. The minority and women-owned business enterprises must be selected from those categories of work identified in Item F of this RFS.

F. Selection Criteria:

In evaluating proposals, the Owner and Designer Selection Panel will consider the members of the proposed design team. Identify those member(s) of the proposed design team who will be responsible for the following categories of work: (Firm's name, individual's name and professional registration or license number, as applicable, must be listed in the application for each category of work, as well as whether the firm is SOMWBA certified as an MBE and/or WBE).

- 1. Architecture
- 2. Environmental Permitting
- 3. Hazardous Materials
- 4. Civil Engineering
- 5. Structural Engineering
- 6. Landscape Architecture
- 7. Fire Protection Engineering
- 8. Plumbing Engineering
- 9. HVAC Engineering
- 10. Electrical Engineering
- 11. Data/Communications Consultant
- 12. Food Service Consultant
- 13. Laboratory Consultant
- 14. Acoustical Consultant
- 15. Specifications Consultant
- 16. Library/Media Consultant
- 17. Theatrical Consultant

The Owner should list only those categories of work that are important to the project, and the Applicant's response should include team members for only the categories listed. Be careful what you ask for.

Failure of an Applicant to list a team member may result in elimination of the Applicant for consideration by the DSP - even if that Applicant appears otherwise qualified.

- 18. Sustainable/Green Design/Renewable Energy Consultant
- 19. Cost Estimating
- 20. Accessibility Consultant
- 21. Traffic Consultant
- 22. Furniture, Fixtures and Equipment Consultant
- 23. Code Consultant
- 24. Security Consultant

** N.B. -

Applicants must address each category of work listed above in their application whether it is to be performed by in-house staff or by sub-consultant(s).

The members of the team for each of the categories of work listed above must be identified including the firm's name, individual's name and professional registration or license number, as applicable, as well as whether the firm is SOMWBA certified as an MBE and/or WBE).

Failure to address <u>each</u> category may result in the elimination of the applicant from consideration on this project.

Applicants should not list any consultants other than those for the categories of work listed above.

The minority and women-owned business enterprises must be selected from the categories of work listed above. Consultants other than those for the categories of work listed above may not be used for purposes of meeting M/WBE requirements.

The Owner and Designer Selection Panel will consider the following additional criteria in evaluating proposals:

- 1. Prior similar experience best illustrating current qualifications for the specific project.
- 2. Past performance of the firm, if any with regard to public, private, DOE-funded, and MSBA funded projects across the Commonwealth, with respect to:
 - a. Quality of project design.
 - b. Quality, clarity, completeness and accuracy of plans and contract documents.
 - c. Ability to meet established program requirements within allotted budget.
 - d. Ability to meet schedules including submission of design and contract documents, processing of shop drawings, contractor requisitions and change orders.
 - e. Coordination and management of consultants.
 - f. Working relationship with contractors, subcontractors, local awarding authority and MSBA staff and local officials.
- 3. Current workload and ability to undertake the contract based on the number and scope of projects for which the firm is currently under contract.
- 4. The identity and qualifications of the consultants who will work on the project.
- 5. The financial stability of the firm.
- 6. The qualifications of the personnel to be assigned to the project.
- 7. Geographical proximity of the firm to the project site or willingness of the firm to make site visits and attend local meetings as required by the client.
- 8. Additional criteria that the MSBA Designer Selection Panel considers relevant to the project.

(Provide additional preferred qualifications as desired.)

G. Proposal requirements

Persons or firms interested in applying must meet the following requirements:

1.	Applicants must have an up-to-date Master File Brochure on file at the Massachusetts School
	Building Authority.

2.	Applications shall be on "Standard Designer Application Form for Municipalities and Public
	Agencies not within DSB Jurisdiction 2005" as developed by the Designer Selection Board of the
	Commonwealth of Massachusetts (http://www.mass.gov/cam/DSB/DSB App 2005 CT.doc).
	Applications (one original and twenty (20) copies) must be received on or beforeAM/PM
	, 2008. Applications should be printed double-sided and bound in such a manner that
	the pages lie and remain flat when opened. Applications should <u>not</u> be provided with acetate
	covers

- **3.** Applications must be accompanied by a concise cover letter that is a maximum of two pages in length. A copy of the cover letter should be attached to each copy of the application. The cover letter must include the certifications as noted in Section E of this RFS. (A copy of the MCPPO certification should be attached to the cover letter as well as any SOMWBA letters.)
- 4. Applicants may supplement this proposal with graphic materials and photographs that best demonstrate design capabilities of the team proposed for this project. Limit additional information to a maximum of three, 8½"x 11" pages, double-sided.
- **5.** Proposals shall be addressed to:

Name Address Phone Number Email Fax #

6. Proposals must be clearly identified by marking the package or envelope with the following:

[Name of Project]
"Name of Applicant"

7. All questions regarding this RFS should be addressed exclusively in writing to:

Name Address Phone Number Email Fax #

H. Pre-Proposal Meeting

All interested parties should attend a briefing session at	 scheduled for
, 2008 at 10:00 AM.	

I. Withdrawal

Applicants may withdraw an application as long as the written request to withdraw is received by the Town of Marblehead prior to the time and date of the proposal opening.

J. Public Record

All responses and information submitted in response to this RFS are subject to the Massachusetts Public Records Law, M.G.L. c. 66, § 10 and c. 4, § 7(26). Any statements in submitted responses that are inconsistent with the provisions of these statutes shall be disregarded.

K. Waiver/Cure of Minor Informalities, Errors and Omissions

The Owner reserves the right to waive or permit cure of minor informalities, errors or omissions prior to the selection of a Respondent, and to conduct discussions with any qualified Respondents and to take any other measures with respect to this RFS in any manner necessary to serve the best interest of the Owner and its beneficiaries.

L. Rejection of Responses, Modification of RFS

The Owner reserves the right to reject any and all responses if the Owner determines, within its own discretion, that it is in the Owner's best interests to do so. This RFS does not commit the Owner to select any Respondent, award any contract, pay any costs in preparing a response, or procure a contract for any services. The Owner also reserves the right to cancel or modify this RFS in part or in its entirety, or to change the RFS guidelines. A Respondent may not alter the RFS or its components.

M. Additional Information

Include any additional information that is required or that may assist Respondents in responding to the RFS.

ATTACHMENTS:

Attachment A: Statement of Interest

Attachment B: Contract for Designer Services

Attachment C: Designer Application Form - DSB_App_2005 Attachment D: Certifications (*To be developed by the Owner*) Attachment E: MSBA's Designer Selection Panel's Procedures

End of Request for Designer Services

ATTACHMENT C

PARTICIPATION SCHEDULE FOR DESIGNER CONTRACTS BY SOMWBA CERTIFIED MINORITY/WOMEN BUSINESS ENTERPRISES

This form shall be submitted to the Owner by the Designer upon execution of the Contract for Designer Services attached hereto.

Owner			
Project No:			
Name of Company	Description of Work	M/WBE	<u>Dollar Value</u> Participation
1			<u> </u>
			\$
	_		\$
			\$
5	_		\$
			\$
	Dollar Value of MBE Comm	nitment: \$	
	Dollar Value of WBE Comn		
	Total Dollar Value Commit		
	Original Fee for Basic Serv	vices Amoun	t \$
DESIGNER CERTIFICAT	ΓΙΟΝ		
listed firms for the identified w	ler the penalties of perjury that (1) is rork and dollar amounts and (2) certion tract with regards to MBE/WBE part forth above.	fies that he/she	has read the terms and
Date			
		Name of Arc	hitect/Engineer
		Authorized	Signature
		Addr	ess
		City, State	& Zip Code

ATTACHMENT D

M.G.L. c.30 §39R - INTERNAL ACCOUNTING CONTROLS APPLIES TO CONTRACTS OF \$100,000 OR MORE SAMPLE LETTER TO BE PREPARED ON DESIGNER'S LETTERHEAD

Date

CEO Owner 123 Reservoir Street Enfield, MA 01234

RE: Enfield High School

Dear:

This Statement of Internal Accounting Controls is being submitted in accordance with Article 17.5.3 of the Contract for Design Services for the above captioned project. Please be advised that our firm, the Designer under the Contract, has a system of internal accounting controls which assures that:

- 1. transactions are executed in accordance with management's general and specific authorization;
- 2. transactions are recorded as necessary, to permit preparation of financial statements in conformity with generally accepted accounting principles, and to maintain accountability for assets;
- 3. access to assets is permitted only in accordance with management's general or specific authorization; and
- 4. the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action was taken with respect to any difference.

Sincerely,

<u>ATTACHMENT E</u>

MGL c.30 §39R – INTERNAL ACCOUNTING CONTROLS APPLIES TO CONTRACTS OF \$100,000 OR MORE SAMPLE LETTER TO BE PREPARED ON CPA's LETTERHEAD

CEO Owner 123 Reservoir Street Enfield, MA 01234 RE: Dear Please be advised that we have reviewed the Statement of Internal Accounting Controls prepared by the in connection with the Name of Designer above-captioned project. This statement is required under M.G.L. c.30 §39R. In our opinion, representations of management are consistent with our evaluations of the system of internal accounting controls. In addition, we believe that they are reasonable with respect to transactions and assets in the amount which would be material when measured in relation to the firm's financial statements. Sincerely, (CPA)

ATTACHMENT F

CONTRACT FOR DESIGNER SERVICES AMENDMENT NO. _____

WHEREA				("Owner") and	
		, (the "Do	esigner") (colle	ectively, the "Parties	")
entered in	to a Contract for Designer Servi	ices for the		Project (Pr	<u>oject</u>
Number	<u>) at the </u>		School on		
"Contract"	<u>; and</u>				
WHEREA	S, effective as of	, the	Parties wish	to amend the Contra	act:
this Amen	EREFORE, in consideration of t dment, and other good and valu hereby acknowledged, the Part	iable consid	leration, the re	eceipt and legal suffi	ciency of
Phase	wner hereby authorizes the Des , the Construction Phases, and t ms and conditions set forth in the	he Final Co	mpletion Phas		
	e performance of services requi e compensated by the Owner ir es:				igner
Fee fo	r Basic Services:	Original	Contract	After this Amendm	ient
Feasib	ility Study Phase	\$		\$	
Schem	natic Design Phase	\$		\$	-
Desigr	Development Phase	\$		\$	
Constr	ruction Document Phase	Φ		\$	_
Biddin	g Phase	\$		\$	_
	ruction Phase			\$	-
Compl	etion Phase	Φ.		\$	
	Total Fee	\$		\$	
This Amer	ndment is a result of:				

3.	The Construction Budget shall be as follows:	lows:	
	Original Budget:	\$	
	Amended Budget	\$	
4.	The Project Schedule shall be as follows	s:	
	Original Schedule:	\$	
	Amended Schedule	\$	
5.	5. This Amendment contains all of the terms and conditions agreed upon by the Parties as amendments to the original Contract. No other understandings or representations, oral or otherwise, regarding amendments to the original Contract shall be deemed to exist or bind the Parties, and all other terms and conditions of the Contract remain in full force and effect		
		the prior approval of the Authority, and the Designer ed by their respective authorized officers.	
OW	VNER		
	(print name)		
	(print title)	_	
By	(signature)	_	
Dai	te	_	
DE	SIGNER		
	(print name)		
	(print title)		
•	(signature)		

```
THINONALLY LEFT BLANK
```

ATTACHMENT B -1:

Designer Services Contract

Amendment for Design / Bid / Build



```
THINONALLY LEFT BLANK
```

7.5 <u>Design Development Phase</u>

- 7.5.1 Upon receipt of an Approval to proceed to the Design Development Phase, the Designer shall meet regularly and as necessary with the Owner, the OPM, and the Authority. This shall include meeting at least once every other week with the Owner and the OPM during this Phase.
- 7.5.2 The Designer shall update and refine items submitted during the Schematic Design Phase, and shall submit to the Owner and the Authority, on or before the date specified in the Project Schedule, and on the basis of the approved Schematic Design Phase Documents, the following deliverables as they are defined in this Article 7.5.2 and as they are further defined in Articles 7.5.3, 7.5.4, 7.5.5, 7.5.6 and 7.5.7:
 - (a) a list of all filings and permits within Designer's scope of services and professional expertise required to implement the design and a schedule of target dates for the procurement of such permits, which list and schedule shall be regularly updated during the term of this Contract;
 - (b) information and documentation within the technical expertise of the Designer and that is necessary for the Owner to file local basic zoning and environmental permits. The Designer, as Extra Services, shall provide information and documentation for the Owner to file Environmental Notification Forms, Environmental Impact Reports, and any other filings for permits that must be filed during the design development phase;
 - (c) soils exploration data, geotechnical and geoenvironmental reports, showing exploratory locations relative to siting of proposed structures;
 - (d) complete design development drawings; outline specifications indicating any filed sub-bid sections and sub-sub trades based on the cost of the work and other documents necessary to specify the size and character of the Project as to siting, landscape, architectural, structural, fire protection, plumbing, heating, ventilating and air conditioning, electrical, ADA/MAAB, product requirements and other features:
 - (e) quality control documentation demonstrating, without limitation, coordination of: ceiling clearances, mechanical room size, and shaft sizes; specifications and drawings; filed sub-bid work or sections; scheduling; equipment and power; existing and new construction; and phasing;
 - (f) design development drawings which the Designer shall submit for review to the local building official;
 - (g) a life cycle cost analysis to determine which design decisions related to all energy and water consuming devices and overall building operation and maintenance are the most cost effective [M.G.L. c. 149, s. 44M];
 - (h) a construction cost estimate for the design in Uniformat II Level 3 format, with unit rates and quantities supporting each item and reconciled with the detailed construction cost estimate and any updated cost estimates in accordance with Article 7.5.6. The estimate cost shall be projected, to the mid point of the construction period;

- (i) a space measurement analysis for the design verifying that the sum of all program areas in the Project plus all other floor areas in the Project equals the gross floor area of the Project;
- (j) a written summary or summaries comparing the project design, as represented in the design development drawings, specifications and cost estimates with the Final Design Program requirements, and explaining any deviations in writing.
- 7.5.3 Design Development Drawing Requirements: The Design Development drawings shall illustrate and describe the refinement of the design of the Project to a level of detail that is customary and standard, establishing the scope, relationships, forms, size and appearance of the Project by means of plans, sections and elevations, typical construction details, and equipment layouts. Drawings shall delineate locations and elements of Work which may be proposed to be assigned to project construction phases and/or separate bidding packages. Documents shall include, but not be limited to, the following:
 - (a) Site and utility drawings showing;
 - 1. Existing and proposed contours and locations of the proposed building or addition(s). Show entry level elevation and key exterior grades at perimeter. Indicate all retaining walls. Include benchmarks of site if survey is available.
 - 2. All utilities existing and proposed, indicating location, elevation, composition and size e.g., manholes, sewers, hydrants, light standards. Include work by others, e.g., gas and electric utility providers.
 - 3. Roads, laid out parking areas, walks, recreation areas, terraces and other site improvements.
 - 4. Building locations fixed and referenced from main survey baseline, if available.
 - 5. Plant materials with preliminary schedule.
 - (b) Building drawings and other graphic and written requirements with floor plans showing: (minimum scale 1/8" = 1'0");
 - 1. building perimeter with exterior wall thicknesses and overall dimensions;
 - 2. structural grid;
 - 3. plan requirements of mechanical and electrical systems;
 - 4. building core; elevators, stairs, shafts, toilet rooms;
 - 5. interior partitions; appropriate thicknesses and dimensions to fix basic organizations; indicate fire separations, ratings;
 - 6. door swings;
 - 7. floor elevations;
 - 8. built-in furniture and equipment; and
 - 9. furniture layout concept drawings.
 - (c) Roof plans showing;

- 1. proposed systems type;
- 2. pitch and drainage patterns;
- 3. roof drains, gutters and scuppers;
- 4. skylights, stairs through roof, penthouses, major equipment, chimneys.
- (d) Building sections: One transverse and one longitudinal section. Indicate floor to ceiling heights and floor-to-floor heights. Label all spaces;
- (e) Building elevations showing;
 - 1. full height elevations including roof structures, e.g., mechanical equipment, chimneys, and penthouses;
 - 2. floor elevations, floor-to-floor heights, and overall height related to benchmarks on site plans;
 - 3. all fenestration:
 - 4. column centerlines;
 - 5. principal finish materials indicating major control and expansion joints, and divisions of materials where required;
 - 6. louver and equipment enclosure systems; and
 - 7. exterior grades and topographical features in context.
- (f) Full height wall sections for main elevations and at special conditions. Show foundation and perimeter treatment, wall construction including insulation and supporting structure, fenestration and mechanical penetrations, and floor construction;
- (g) Interior elevations: Major spaces, e.g. library, lobby, and all typical spaces, e.g. classrooms;
- (h) Reflected ceiling plans: show prototypical structural, fire protection, mechanical and electrical information for classrooms and major spaces, including lighting layouts with ceiling heights and material changes;
- (i) Colored interior elevations and perspectives of major and typical spaces
- (i) Schedules:
 - 1. finish schedule by room types;
 - 2. door schedule by room;
 - 3. window schedule;
 - 4. equipment schedules, e.g., food service, instructional media.
- (k) Structural Concepts;
 - 1. Foundation plan showing sizes and locations of typical components.
 - 2. Framing plans: typical floor framing, roof framing, special framing, show framing at major openings and sizes of members.
 - 3. Column locations.

- 4. Preliminary details including floor and roof deck, statements as to methods of lateral bracing and how requirements of earthquake code will be met.
- 5. Details for special and/or incidental structural features, e.g. tunnels, connecting bridges and unique architectural features.
- 6. Connection to existing buildings at foundation and at key points at existing structure if applicable.
- (l) Fire Protection: floor plans indicating wet or dry type systems, hose racks or cabinets and fire department tie-ins. Indicate whether a fire pump will be required and, if so, show location within the building. Show typical sprinkler head layout;
- (m)Plumbing and sanitary systems: floor plans indicating locations of all plumbing fixtures and special features, and approximate location and size of all piping systems and principal items of equipment;
- (n) Heating, Ventilating and Air Conditioning Systems;
 - 1. Show locations and approximate sizes of piping systems, air handling systems and principal items of equipment such as compressors or cooling towers.
 - 2. Indicate space requirements of major equipment and their location in mechanical rooms and fan rooms. Major shafts.

(o) Electrical Systems;

- 1. All services including those for special purposes shall be located and indicated.
- 2. Lighting shall be indicated as to type, location and intensities in foot-candles for each special and typical space.
- 3. Switchgear and emergency generator.
- 4. Fire alarm system drawings showing all initiation and signaling devices, control panels, annunciator panels, etc.
- 5. Security system drawings.
- 6. Communications drawings showing chases, major equipment locations and any special distribution requirements.
- 7. CATV/CCTV drawings showing chases, major equipment locations and any special distribution requirements.
- 8. Information Technology drawings showing chases, major equipment locations and any special distribution requirements.
- 7.5.4 Other Consultant's Drawings and Other Graphic and Written Requirements: For special consultants, e.g., kitchen, elevator, library, media room, equipment where appropriate, provide drawings that locate and define the scope of the work. Coordinate with other disciplines. Provide cuts of all major pieces of equipment.
- 7.5.5 Project Manual Requirements (Specifications):
 - (a) Outline Specifications that are to accompany Design Development Drawings shall be prepared to a level of detail that is standard and customary and shall include,

but not be limited to, a comprehensive description of the Project and the materials proposed for use in the work. No detailed specifications of materials or workmanship procedures need be included; however, the general scope shall be indicated by CSI MasterFormat as applicable to proposed construction.

- The Design Development Outline Specification shall also include a
 comprehensive "BASIS OF DESIGN." The "BASIS OF DESIGN" shall be a
 narrative description of the Project and shall include all applicable
 architectural, civil, structural, mechanical and electrical programs and/or
 systems. Identify all proposed filed sub-bid categories.
- 2. Project Manual shall include a statement to define Work which is proposed to be included in separate construction phases and/or bid packages.
- (b) The following is a list of items that shall at a minimum be identified or outlined in this Phase:
 - 1. Site work; clearing, drives, walks, parking areas, fences, excavation, backfill, planting.
 - 2. Footings; on earth, rock, piles, caissons, proposed bearing pressures, boring logs.
 - 3. Foundation walls; type of concrete, reinforcing, type and extent of waterproofing.
 - 4. Footing drains; type, disposal of drainage.
 - 5. Exterior walls: superstructure, type, materials, brick type, alternate cladding, back-up materials, dampproofing material and extent, special features.
 - 6. Roofs; types, vapor barrier, insulation, flashings, all materials.
 - 7. Flashings; general types, all materials, weights, where each type is to be used.
 - 8. Sheet metal; gutters, leaders, others uses, except flashings.
 - 9. Windows; general types, materials, sub-frames, finish, glazing, screens.
 - 10. Doors, exterior and interior; types.
 - 11. Steps, exterior; including platforms and landings' materials.
 - 12. Stairs, interior; including platforms, landings, walls, materials and finishes.
 - 13. Framing; wood, concrete or metal systems in accordance with general design.
 - 14. Partition construction related to room type.
 - 15. Cabinet and casework; types and materials.
 - 16. Food Service Equipment; types and materials.
 - 17. Furring; lathing, plastering, materials and locations.
 - 18. Insulation thermal; types, thicknesses, methods of application and locations.
 - 19. Acoustical treatments; types, thicknesses, methods of application and location.
 - 20. Interior finishes; materials for floors, walls, bases, wainscots, trim, ceilings, ceiling heights.

- 21. Fire Protection; standpipe systems, sprinkler systems, fire pumps and accessories.
- 22. Water supply; source; location of main to which connection will be made; type of pipe for service main; load requirements; load factors and pressures.
- 23. Sanitary sewers; sewage disposal system, pipe and other materials.
- 24. Storm sewers; storm drainage disposal system (institution or local facility), pipe and other materials.
- 25. Gas main; material, size, location. Interface with utility company.
- 26. Plumbing; systems such as wastes, vents, hot water, cold water, gas, air, oxygen, vacuum, main source of supply, materials for each, water heaters, pumps, thermal insulation fixture quality, all special features.
- 27. Heating, ventilating and air conditioning; type of heating and refrigeration plants, type and capacity of boilers and cooling equipment, fuel, type of burners, fuel storage, heaters, feed water pumps and heaters, thermal insulation, type of heating medium, supply and return piping, radiation, unit heaters, radiant heating, principal air conditioning equipment types, special features, supply, return and exhaust ductwork.
- 28. Electric work; service connection, location, institution or public utility, overhead or underground, transformers including type and location, types of conduit and wiring, types of fixtures, location of main switchboard, radio, fire alarm, telephone, public address, emergency lighting and wiring, emergency or other generators, special features, including Master TV, information retrieval and/or data processing system.
- 29. Elevators, dumbwaiters and platform lifts; capacities, speed, travel in feet, landings, operation, controls, platform sizes, machine type and location, car and entrance finishes, signals.
- 30. Other built-in equipment, types and materials.
- 31. Special features.
- 7.5.6 Construction Cost Estimate Requirements The Designer shall provide a construction cost estimate in Uniformat II Level 3 format with aggregated unit rates and quantities supporting each item referenced in Article 7.5.5(b). The estimate cost shall be projected, to the mid point of the construction period.
 - (a) The Designer shall review its construction cost estimate in comparison with the detailed construction cost estimate, and any updated cost estimates, provided by the OPM and shall work in good faith and in cooperation and coordination with the OPM to reconcile any differences between the construction cost estimates, to clarify assumptions upon which the cost estimates are based and to address any concerns or questions with the cost estimates that are raised by the Owner, the OPM, or the Authority. If the Designer is unable to reconcile all differences between the two construction cost estimates with the OPM, then the Designer shall provide a detailed explanation of the differences to the Owner. If, in any case, the agreed-upon, reconciled construction cost estimate exceeds the Project

Construction Budget, the Designer shall cooperate with the Owner and the OPM in identifying, specifying and recommending changes in, or additional specification of materials, equipment, component systems and types of construction, or other adjustments in the scope or quality of the Project, including contingencies or alternative bid items, so as to facilitate revision of the design of the Project to reduce the cost of construction so as to comply with the authorized Project Construction Budget.

- (b) Cost estimate data shall be organized to identify elements of project work which may be proposed to be advanced under separate construction phases and/or separate bidding packages. When so proposed, estimates shall develop cost data relative to corresponding bidding and work execution dates established in project schedules.
- 7.5.7 Reports, drawings, specifications, cost estimates and other design development submittals shall be subject to the written approval of the Owner and the Authority. Unless a lesser number is requested by the Owner, the Designer shall submit to the Owner for approval six (6) copies of Design Development drawings, specifications, cost estimates, and other submittals. Two (2) copies shall be submitted to the Authority by the Designer.
- 7.5.8 The Designer shall present and explain the Design Development submittal to the Owner and the Authority and at a local public meeting scheduled by the Owner, if any such meeting is scheduled or in conference.
- 7.5.9 The Designer and its Subconsultants shall collaborate with the Authority's Commissioning Consultant to develop design criteria which will support the purposes of building commissioning and energy/resources conservation concepts as commonly understood and as prescribed by the Commissioning Consultant.
- 7.6 <u>Construction Documents Phase:</u> In addition to the requirements specified in the RFS (Attachment B), upon receipt of an Approval to proceed with the Construction Documents Phase of the Project from the Owner, the Designer shall do the following:
 - 7.6.1 The Designer shall meet regularly and as necessary with the Owner, the Authority, the OPM, and the Commissioning Consultant. This shall include meeting at least twice per month (or more frequently if needed) with the Owner and the OPM during this Phase.
 - 7.6.2 Based on the submittals approved in the Design Development Phase of the Project, the Designer shall update and refine the items previously submitted and shall submit the following on or before the date and time specified in the Project Schedule:
 - (a) Construction documents progress submittals as follows:
 - 1. a 60% Construction Documents Submittal, with deliverables as defined in Article 7.6.3;
 - 2. a 90% Construction Documents Submittal, with deliverables as defined in Article 7.6.4:

- 3. a Final Construction Documents Submittal, with deliverables as defined in Article 7.6.5:
- 4. a Bid Documents Submittal, with deliverables as defined in Article 7.6.6
- (b) As a part of each of the submittals required under Articles 7.6.3, 7.6.4, and 7.6.5, an updated work plan and recommended updates for incorporation into the Project Schedule by the OPM;
- (c) As a part of each of the submittals required under Articles 7.6.3, 7.6.4, and 7.6.5, a report on the status of environmental, zoning, planning, building code, and ADA/MAAB approvals and permitting processes and a certified list of all required testing and all required permits identified in 7.5.2(a).
- (d) All submittals by the Designer shall be subject to the written approval of the Owner, which approval shall not be unreasonably delayed, withheld, conditioned, or denied. Unless a lesser number is requested by the Owner or is specifically provided hereinafter, the Designer shall furnish to the Owner for approval six (6) sets of the drawings, specifications, construction cost estimates and all other submittals. Unless a lesser number is specifically provided hereinafter, the Designer shall furnish two (2) sets of said drawings, specifications, construction cost estimates and all other submittals to the Authority. The Designer shall also furnish to the Owner and the Authority electronic media copies of the foregoing drawings and documents in such form as may be required by the Authority.

7.6.3 60 Percent Construction Documents Submittal:

- (a) The Designer shall provide, on or before the date and time specified in the Project Schedule, a 60 % Construction Documents Submittal (60% CD Submittal), which shall include:
 - 1. Construction Documents and other deliverables, as defined in this Article 7.6.3 and as further defined in Articles 7.6.2, 7.6.7, 7.6.8, and 7.6.9, advanced to a level of intermediate (60 percent) completion, and incorporating corrections to indicate compliance with Owner and Authority review comments related to prior submittals.
 - 2. In instances where the Designer takes exception to the Authority's previous review comments on the Design Development submittal, a written statement explaining its position.
 - 3. The Basis of Design that accompanied the Outline Specifications in the Design Development Phase shall be updated and expanded to include all proposed architectural, structural, fire protection, plumbing, mechanical, electrical, civil, and landscape design concepts for the Project.
 - 4. A space summary, in the form and format prescribed by the Authority, that sets forth the current space calculations and totals and certifies that said space calculations and totals are in compliance with those previously authorized by the Authority in the Project Funding Agreement.

- 5. Keying of graphics shall be sufficient to allow a reviewer to make his or her way through the set.
- 6. A list of all drawings related to the Project.
- 7. A materials selection statement identifying typical interior and exterior surfaces and their materials.
- 8. A color theory statement indicating proposed paint colors and material selections for typical and special spaces and why they have been selected and how these selections relate to surrounding materials and colors.
- 9. Large scale plans of all mechanical and electrical spaces with major equipment indicated.
- 10. Project Manual, including all sections to be included in final technical specifications, developed to include a list of all materials in the building with their manufacturers. Identify all specifications sections which need to be filed sub-bid.
- 11. Identify all proposed bid alternates by inclusion in a project manual section to be titled "Alternates." Alternates shall be listed in sequence as approved by the Owner. Work required under bid alternates shall be described and/or drawn, as appropriate, to clearly define the design criteria and extent of work involved for implementation of the bid alternate. In each instance, the existing conditions and/or new design criteria for base bid work shall also be described and indicated in documents.
- 12. Code analysis: Provide a building code analysis. Any deviation from methods of compliance described in earlier submittals shall be indicated. Code analysis shall identify its preparer, code edition referenced, and include a comprehensive description of operative building code provisions, with floor plans showing fire separation types, area calculations, egress capacity for exits and exitways, and any special features required to comply.
- (b) As a requirement of the 60% CD Submittal, and in accordance with the provisions of this paragraph and Article 7.6.9, the Designer shall provide a construction cost estimate prepared using the Uniformat II Classification to Level 3, the CSI MasterFormat 6-digit format to Level 3 and MGL c.149 §44F (filed sub-bid) format including quantities of all materials and unit prices of labor, equipment, and materials as well as a cost estimate for each item of work, for review by the Owner and the Authority. The Designer shall submit said construction cost estimate separately, as a supplement to the 60% CD Submittal, no later than twenty-one days after the submission of the 60% CD Submittal described in Article 7.6.3(a). The development of said construction cost estimate shall under no circumstances delay the timely submission of the remainder of the 60% CD Submittal.

7.6.4 90 Percent Construction Documents Submittal:

- (a) The Designer shall provide, on or before the date and time specified in the Project Schedule, a 90 % Construction Documents Submittal (90% CD Submittal), which shall include:
 - 1. Construction documents and other deliverables as defined in this Article 7.6.4 and as further defined in Articles 7.6.2, 7.6.7, 7.6.8, and 7.6.9, advanced to a level of substantial (90 percent) completion, and incorporating corrections to indicate compliance with Owner and Authority review comments related to prior submittals.
 - 2. A space summary, in the form and format prescribed by the Authority, that sets forth the current space calculations and totals and certifies that said space calculations and totals are in compliance with those authorized by the Authority in the Project Funding Agreement.
 - 3. Interior Materials Color Boards, including samples of principal interior materials, labeled and mounted to indicate locations.
 - 4. Final structural and energy design calculations.
 - 5. A statement confirming that the Owner has been provided with structural design drawings, specifications, and calculations sufficient to enable execution of an independent structural peer review process, as defined in the Massachusetts Building Code, as amended (this requirement is applicable, to satisfy Authority requirements for all school construction projects having a floor area in excess of 10,000 square feet). The Designer shall have advised the Owner of this requirement in writing not less than sixty (60) days prior to delivery of the 90% CD Submittal in order for the Owner to arrange for the services of an Independent Structural Peer Reviewer. Upon reaching 90 percent completion of construction documents, Designer's structural engineering consultant shall have reached a level of 100 percent completion of its construction documents to enable advancement of the independent structural peer review.
 - 6. The Designer and its consultants shall fully cooperate with the Independent Structural Peer Reviewer in the process. The Designer shall obtain a copy of the Independent Structural Engineering Review report and submit same to the Owner and the Authority at the time of completion of the remainder of the construction documents at the level of final completion.
 - 7. In instances where the Designer takes exception to any of the Authority's 60% CD Submittal review comments, a written position statement explaining the Designer's position on its exceptions to said review comments.

7.6.5 Final Construction Documents Submittal:

- (a) The Designer shall provide, on or before the date and time specified in the Project Schedule, a Final Construction Documents Submittal, which shall include:
 - 1. construction documents and other deliverables as defined in this Article 7.6.5 and as further defined in Articles 7.6.2, 7.6.7., 7.6.8, and 7.6.9, advanced to a level of final (100 percent) completion, and incorporating corrections to indicate compliance with Owner and Authority review comments related to prior submittals.
 - 2. a final construction cost estimate, in accordance with the provisions of this paragraph and Article 7.6.9, based on 90% Construction Documents, including cost estimates for general conditions, overhead and profit, insurance, bonds, and all other items expressed as percentage rates for design contingencies and construction contingencies and escalation to the bid date; and other mutually agreed upon contingencies. The final construction cost estimate shall be prepared in Uniformat II Elemental Classification to Level 3 (Sections A-G inclusive), the CSI MasterFormat to Level 3 and M.G.L. c.149, §44F (filed sub-bid) format and shall be complete with a single line description for each item with the detailed unit rate or item cost buildup provided in each case.
 - 3. complete construction drawings and specifications, certified by the Designer as having satisfied the firm's quality control review process as previously confirmed with the Owner, in sufficient detail to permit fixed-price bids in open competition for construction of the Project when documents have been approved for issuance for bidding.
 - 4. no later than at the 100% stage of completion of the final drawings and specifications, two sets of the final drawings and specifications that shall be provided to the local building official to be signed and stamped "Approved" by the local building official; two sets of plumbing drawings and specifications that shall be provided to the local plumbing inspector to be signed and stamped "Approved" by the local plumbing inspector; two sets of the fire protection, HVAC, and electrical construction documents that shall be provided to the local fire official to be signed and stamped "Approved" by the local fire official; two sets of the electrical construction documents that shall be provided to the local electrical inspector to be signed and stamped "Approved" by the local electrical inspector. Notwithstanding the foregoing, the Owner acknowledges that building officials, department inspectors, and fire officials have varying policies on approvals and submittal procedures, and the only obligation of the Designer in this regard is to promptly make the submittals described herein and assist the Owner in receiving the approvals to the extent available.
 - 5. at the 100 percent stage of completion of final drawings and specifications, a written summary comparing the final construction drawings and specifications and final estimated construction cost with the Final Design Program requirements and submittals made during the Design Development Phase and

- earlier in the Construction Documents Phase, explaining any significant deviations.
- 6. In instances where the Designer takes exception to any of the Authority's 90% CD Submittal review comments, a written position statement explaining the Designer's position on its exceptions to said review comments.
- 7. The Independent Structural Engineering Peer Review Report obtained from the Independent Structural Engineering Peer Reviewer referenced in Article 7.6.4(e). The Designer shall include a certification statement from the project structural engineer designer of record to acknowledge receipt of the Report and to indicate response actions pursuant thereto. The Designer shall also forward a copy of said Report to the Building Inspector.
- 8. A certification that all applicable local, state and utility officials have been contacted by the Designer regarding each utility connection and that the persons responsible for permits or connection approval have agreed to the systems' use.

7.6.6 <u>Bid Documents Submittal</u>:

- (a) The Designer shall provide, on or before the date and time specified in the Project Schedule, a Bid Documents Submittal which shall include:
 - 1. Construction documents and other deliverables as defined in this Article 7.6.6 and as further defined in Articles 7.6.2, 7.6.7, and 7.6.8, incorporating corrections to indicate compliance with Owner and Authority review comments related to prior submittals.
 - 2. From the construction drawings and specifications approved by the Owner, incorporating such changes as the Owner or the Authority requires, a set of reproducible black and white drawings and original specifications on high quality white bond paper, single-sided, properly packaged, suitable for reproduction, stamped and signed by all disciplines, that shall be prepared by the Designer and transmitted to the Owner; which documents shall become the property of the Owner as provided under Article 16. Other suitable reproducible media, having the same content shall be substituted, when so directed or authorized by the Owner.
 - 3. Upon receipt of Owner authorization to advance to reproduction the approved documents for distribution to bidders and, upon reproduction thereof, the Designer shall promptly submit complete sets of bid documents to the Owner (two sets) and the Authority (one set half size for Drawings). Any subsequent addenda shall be promptly submitted to the Owner and the Authority.

7.6.7 <u>Drawing Requirements</u>:

- (a) The documents prepared during the Construction Documents Phase shall set forth the requirements for construction of the Project to a level of detail that is customary and standard and shall include, but not be limited to:
 - 1. General information showing drawing index, symbols, abbreviations, notes, locations map.
 - 2. Site drawings shall be complete to define the extent and detail of site work. Show the following:
 - a. Layout and location of all proposed work including buildings, structures, retaining walls, parking, walls and all other site improvements, with details.
 - b. Existing and proposed grades and contours including floor elevations, existing structures and topography, survey base line, bench marks and boring locations.
 - c. Landscaping and planting.
 - d. All utility service lines, systems and structures for electricity, gas, oil, water, steam, telephone, CATV, fire alarm, sanitary and storm drainage including size, composition, grades and directions of flow.
 - e. Contract Limit Line and Storage Area for construction materials.
 - f. All existing foundations, obstructions and other physical characteristics of the site which may affect the construction work.
 - g. Site survey.
 - h. Cuts of benches, light standards.
 - 3. Demolition drawings and temporary work required.
 - 4. Architectural drawings shall include at a minimum:
 - a. Floor plans of each floor, including basement and lofts or attic with room and corridor dimensions, wall thicknesses, column locations, floor elevations, mechanical and electrical openings, door and window designations, partition types, floor materials, built in furniture and equipment, keyed to other architectural drawings. All rooms numbered.
 - b. Large scale floor plans where required to illustrate detailed requirements of rooms.
 - c. Large scale plans showing key areas e.g. lobby, special spaces. Indicate surface materials. (minimum scale $\frac{1}{4}$ " = 1' 0")
 - d. Roof plans showing openings, drainage, slopes, expansion joints and all projections, including equipment.
 - e. Key plans on all floor plans and section drawings, where appropriate.
 - f. Building Sections as required to show spatial organization of building but no less than one longitudinal and one transverse.
 - g. Building elevations. All building elevations shall be fully developed, and hidden elevations shall be shown. Elevations shall be shown in a sequence as unfolded from a certain point.

- h. Full height wall sections indicating dimensions, flashing, anchorage, reinforcing, coursing, cladding, and all other conditions at wall, roof, foundation, interior floors.
- i. Exterior details, for roofing, flashing, expansion control and construction joints, waterstops and other details showing all conditions both vertical and horizontal, including schedules.
- j. Door, window, entrance, and storefront, schedules, and details.
- k. Vertical circulation plans, sections and details including stairs, elevators, conveyors, dumbwaiters.
- 1. Interior elevations of all significant and typical spaces.
- m. Interior details including casework, paneling surfacing and acoustical treatment.
- n. Reflected ceiling plans coordinated with fire protection, mechanical and electrical drawings, and ceiling details.
- o. Schedules (clearly define new or existing)
 - i. Doors
 - ii. Equipment, e.g. for services
 - iii. Partitions
 - iv. Finishes

5. Structural drawings shall indicate the following:

- a. Indicate or refer to location of geotechnical exploration data and reports related thereto.
- b. Foundation plans with bottom grades showing layout of all footings, walls, slabs on grade including reinforcing, grade beams, and columns; include design soil bearing pressures and live loads.
- c. Floor and roof plans of structural systems including framing, grades of finished floors and depressed areas, with locations and dimensions for all openings. Also indicate design floor loads.
- d. Complete foundation wall elevation and typical sections, with reinforcing indicating location, dimensions and grades for all footings, steps and wall openings.
- e. Complete details and sections with dimensions for all construction including expansion and construction joints, reinforcing and other embedded items.
- f. Schedules (with dimensions) for all lintels, beams, joists, and columns.
- g. Unless detailed on the Drawings, the following information shall appear in the general notes: class and 28 day strength of concrete for each portion, structural steel and concrete reinforcing design stresses for each type of structural member, concrete cover for each type of structural member, shrinkage and temperature steel requirements, reinforcing laps for main reinforcing and temperature steel; bendpoint, cutoff, and hook locations for all members, minimum beam and lintel bearing. Reinforcing steel fabrication shall be in accordance with most recent ACI, "Manual of Standard Practice for Detailing Reinforced Concrete." Structural steel fabrication shall be in accordance with the AISC "Manual of Steel Construction."

- 6. Fire protection drawings shall indicate standpipe systems, sprinkler systems, suppression systems, access panels, fire pumps, accessories, and piping. All piping, equipment, fixtures and devices shall be located and sized. Design criteria shall be provided on the drawings in accordance with NFPA requirements.
 - a. Fire protection work, other than site work, shall not be combined on the same sheets with the Plumbing, HVAC, Electrical, or other drawings except with the prior approval of the Owner.

7. Plumbing drawings shall indicate the following:

- a. All work done by the Plumbing Subcontractor, which includes all water, gas, air, vacuum, medical gases, sanitary and storm wastes, and accessories. Include foundation drain lines unless established as the work of the General Contractor and shall not be indicated on the Plumbing Drawings. Site utilities shall be indicated on the utility drawings.
- b. Plumbing work, other than site work, shall not be combined on the same sheets with the Fire Protection, HVAC, Electrical, or other drawings except with the prior approval of the Owner.
- c. Trapping and venting of all plumbing fixtures including floor drains.
- d. Water and gas supply sources, storm and sanitary discharge mains.
- e. All piping shall be carefully sized and all sizes shall be indicated on drawings and riser diagrams. Indicate all directions of flow and pitch on piping.
- f. All accessories, valves, fixtures including all drinking fountains, grease traps for kitchen waste and all necessary panels, identified as to type and size.
- g. All piping and connections required for other trades (e.g., kitchen equipment, HVAC make-up water, etc.).
- h. Acid waste, vents and neutralization systems for laboratories.
- i. Plumbing Legend and/or graphical symbols on the first sheet of the Plumbing Drawings in accordance with the American National Standards Institute (ANSI).
- j. Plumbing riser diagrams for structures two or more stories in height above the ground level.
- k. Domestic water booster pumps, boiler feed water, meter location, hose bibbs, and wall hydrants.
- 1. Domestic hot water: storage tanks, piping material, hanger details.
- m. All required access panels shall be indicated.
- n. Backflow preventors and cleanouts. Verify that access and clearance provisions for periodically inspected devices, including backflow prevention, are adequate to satisfy requirements of inspecting agencies.
- 8. Heating, Ventilating and Air Conditioning Drawings shall indicate the following:

- a. HVAC work, other than site work, shall not be combined on the same sheets with Fire Protection, Plumbing, Electrical, or other drawings except with the prior approval of the Owner.
- b. All piping and ductwork systems shall be located and sized. All ductwork shall be shown double line.
- c. All systems shall be sized at all reductions and riser diagrams of piping and duct systems shall be indicated.
- d. All directions of flow and pitch on piping, and direction of flow, volumes for duct systems shall be indicated.
- e. All equipment shall have sufficient servicing and/or replacement space indicated on drawings.
- f. All equipment, accessories, valves and dampers with all necessary access panels, identified as to type and size. Access panels, where required for access to valves and dampers shall be indicated on drawings.
- g. Cooling system pumps, chillers, cooling towers, air handling units, ductwork system and dampers, fan details, temperature control system, air and hydronic balancing equipment, and schedules shall be indicated.
- h. Cooling tower design shall be indicated on the drawings showing site location, elevations and floor plan of equipment layout and typical flow diagram as related to the total HVAC system.
- i. All fire and smoke dampers, access panels and doors.
- j. Mechanical room designs:
 - i. Vent pipes for safety valves, relief valves, back pressure valves and tanks shall be extended above flat roofs in accordance with all governing authorities.
 - ii. In all designs for boiler and refrigeration plants, include a complete floor plan indicating location of all major mechanical equipment and sufficient service space.
 - iii. In designs of new and/or replacement boiler and refrigeration plants, provide a flow diagram detailing steam or hot water distribution systems, return systems, including all existing equipment and their function, as well as any proposed expansions with all necessary instrumentation and controls.

9. Electrical Drawings shall indicate the following:

- a. Site utilities shall be indicated on separate electrical site drawings, unless ample space is available on common site for utility drawings.
- b. Electrical work, other than site work, shall not be combined on the same sheets with Fire Protection, Plumbing, HVAC, or other drawings except with the prior approval of the Owner.
- c. General arrangement: Outline layout of each floor. Typical sections through the structure shall be indicated when necessary to define requirements, floor and ceiling heights, elevations, and type construction, including concrete pads shall be indicated. Indicate interface with other systems. Identify any work by general contractor or other trades.
- d. Interior lighting system: Light fixture schedules, circuiting location and mounting heights of all fixtures, receptacle and switch outlets, sizes and

- types of all lamps, conduits, all other accessories and riser diagrams shall be indicated on drawings. Indicate details and method of supporting electrical fixtures and conduits. Designer shall specify that all electrical lighting fixtures be supported from the building structure, and shall be independent of ducts, pipes, ceilings and their supporting members. Comply with seismic design criteria.
- e. Power system: Locations, types and method of control for all motors, heaters, appliances, controllers, starters, branch circuits, feeder conductors and conduits. Indicate riser diagrams. Show details and indicate method of supporting electrical conduit. For larger projects, thermostats and control wiring are normally covered under the HVAC sub-contract, assure coordination.
- f. Fire Alarm, Data, Communications, CATV/CCTV Systems: Locations and types of all devices, outlets and equipment, service connections, wiring diagrams, all other essential details.
- g. Services: Location and details of all services, whether overhead or underground, feeder sizes, plans and elevations of switchgear and transformers, metering and service switchboard arrangements, wiring and ground fault diagram and bus ducts.
- h. General and sub-stations: Location, size, method of connection and protection of all generators, transformers, exciters, motor generators, switch gear, and associated equipment, current characteristics and equipment capacities. Indicate equipment connections by means of one line and/on wiring diagrams and schedule all major items of equipment and all instruments.
- i. Underground work: The size and locations of manholes and types of cables, number, size, and location of ducts, locations, sizes and types of cable supports, fireproofing, duct line profile, and one line diagram of connections. All underground chambers, including manholes and pullboxes, shall be constructed of cast in place or one piece pre-cast concrete.
- j. Pole line work: if required as contract work, indicate location, length, treatment and class of poles, guying, cross arms, insulators, circuiting, transformers, protective and switching devices, lightning arresters, special structures, diagrams, current characteristics and grounding.
- k. Exterior lighting: Location, size, and type of transformers, luminary, poles, light standards, cables, ducts, and manholes, details of control equipment and connection diagrams.
- 1. Emergency system details including transfer switch, type of fuel.
- m. One line diagram indicating load KVA, and available short circuit amperes at each transformer, switchboard, distribution panel board, branch circuit panel board, and at major pieces of equipment.
- n. Riser diagrams for all systems.

7.6.8 Project Manual Requirements:

(a) The format for the Project Manual, including its technical specifications, shall be in accordance with the current CSI MasterFormat with separate sections for each of class of work required by M.G.L. c. 149 §44F.

- (b) The following general information applies to the development of final Specifications:
 - 1. Describe the extent of the work, the materials and workmanship, and include the work under the proper section. If any portion of the work included in a section of the Specifications is to be performed by a trade covered by another section, there shall be clear and distinct cross-referencing between the sections. Merely to state "by others" is not acceptable.
 - 2. For each item of material or equipment, the specifications shall provide for a minimum of three named brands of material or equipment and the words "or equal" or a description of material or equipment which can be met by a minimum of three manufacturers or producers, and the words "or equal." Proprietary products shall not be specified except as provided by M.G.L. c. 30, § 39M; however, when they are specified, proprietary specifications are subject to the "or equal" provisions of M.G.L. c.30, § 39M.
 - 3. Specify materials mined or manufactured in Massachusetts first and the United States of America second whenever possible.
 - 4. Do not use general clauses intended to be all-inclusive in lieu of complete descriptions.
 - 5. Do not duplicate standard requirements that are contained in the contract form.
 - 6. Use consistency throughout. The word "will" shall be used to designate what the Owner, Authority, Owner's Project Manager, Commissioning Consultant, or the Designer can be expected to do, and the word "shall" shall be used to designate what is mandatory for the Contractor or subcontractors to do.
 - 7. Use the same term throughout for the same subject and the term shall be the same as that used on the drawings.
 - 8. Do not use the term "etc."
 - 9. Avoid such terms as "to the satisfaction of the Designer," "as directed by the Designer," "as approved," and "as required".
 - 10. Specify work in appropriate Sections according to local trade jurisdiction.
 - 11. Avoid the use of the following symbols:

Symbol	<u>Use Instead</u>
#	number, no., or pounds percent
"	inch or in.
X	by
•	feet or ft.
0	degree
/	per or at

12. In sections for which filed sub-bids are required, refrain from using such terms as "the Contractor," the "Heating Contractor," or "the Plumbing Contractor," but where necessary for clarity refer to the "HVAC Subcontractor," the "General Contractor" and the like.

- 13. Do not give numbers both in words and figures. Numbers less than 10 shall be written in words, 10 and higher numbers shall be written in figures. In expressing dimensions, figures such as 2 in., 16 in., 7 ft., 6 in., shall be used.
- 14. Each filed sub-bid section shall detail all labor and materials required by the particular sub-trade and list, by number, those drawings (and only those drawings) indicating work of that sub-trade. In addition, list drawings indicating work of a particular trade that appears on drawings that are not customarily included in the work of the trade, when applicable.
- 15. Do not specify that a product or system shall require prequalification or advance approval for use prior to bidding.
- 16. Established unit price items shall be used for work categories which cannot be ascertained for exact quantities in bid documents (e.g. earthwork removal and/or replacement items). In such cases, the Designer shall establish ranges of quantities with associated unit price values for each range. Unit price values shall be established for added work, for deleted work, for base bid quantities when conditions so-suggest. Unit price values shall be ascertained through consultation with cost estimators, be current, equitable, and well defined as to elements of work, overhead, like issues to be encompassed. Established unit prices shall be published within the applicable technical specification sections, and referenced from general conditions as being operative as the basis for determining values to be used for payment or recovery for change order work.
- 17. Staging, scaffolding, cutting and patching, refuse collection and disposal, demolition work and cleaning task, allocation policy and proposed language shall be carefully assigned to avoid duplication or omission.
- 18. A final draft of Project Advertisement, Notice to Bidders, Instructions to Bidders, Contract Forms, General Conditions, Supplementary General Conditions, and other "front end" documents shall be included in the 90% construction documents submittal, along with a final version of all text to appear in Division 1, General Requirements. The Designer may defer insertion of final advertising / bid dates and wage rates, understanding that they are to be established and inserted immediately prior to release of documents for bidding.

7.6.9 Construction Cost Estimate Requirements

The Designer shall provide the construction cost estimates described in Articles 7.6.3 and 7.6.5 in accordance with the following provisions:

(a) The Designer shall review its construction cost estimate in comparison with the detailed construction cost estimate, and any update cost estimates, provided by the OPM and shall work in good faith and in cooperation and coordination with the OPM to reconcile any differences between the cost estimates, to clarify assumptions upon which the cost estimates are based and to address any concerns or questions with the cost estimates that are raised by the Owner, the OPM, or the Authority. If the Designer is unable to reconcile all differences between the two construction cost estimates with the OPM, then the Designer shall provide a detailed explanation of the differences to the Owner and the Authority. If, in any

case, the agreed-upon, reconciled construction cost estimate exceeds the Project Construction Budget, the Designer shall cooperate with the Owner and the OPM in identifying, specifying and recommending changes in materials, equipment, component systems and types of construction, or other adjustments in the scope or materials selections for the Project, including contingencies or alternative bid items, so as to facilitate revision of the design of the Project to reduce the cost of construction so as to comply with the Project Construction Budget.

- (b) Cost estimate data shall be organized to identify elements of project work which may be proposed to be advanced under separate construction phases and/or separate bidding packages. When so proposed, estimates shall develop cost data relative to corresponding bidding and work execution dates established in project schedules.
- (c) Cost estimates shall be projected to the mid point of the construction period.
- (d) The summary sheets shall contain the following:
 - 1. The date that the estimate was prepared. (Value Date).
 - 2. The anticipated bid date.
 - 3. The project and contract number.
 - 4. The title and location of the project.
 - 5. The name of the Designer.
 - 6. The name of the Estimator.
 - 7. The site work cost (including all utilities).
 - 8. The building cost (including fixed equipment).
 - 9. The estimated construction cost of each Phase of the work, totaled.
- 7.6.10 The Designer shall participate in a final review of the Construction Documents with the Owner, the OPM, and the Commissioning Consultant, and the Designer shall incorporate such changes as are necessary to satisfy the Owner's review comments.

7.7 Bidding Phase

7.7.1 The Designer shall, when authorized by the Owner, prepare for reproduction and distribution the construction bid documents, including advertisements, for receipt of proposals from construction contractors, and for execution of the Owner-Contractor Agreement. The Designer shall prepare all addenda (to include bidder questions and Designer responses), subject to the Approval of the Owner and the Authority. The Designer shall attend the pre-bid conference if one is scheduled, taking note of all questions asked. Relevant questions submitted in writing shall be answered by the

Designer by means of written addenda to the bid documents as required. The Designer shall attend each bid opening and, with the assistance of the Owner's Project Manager, conduct a review of the qualifications of the low filed sub-bidders and general bidder (and of other bidders if necessary) and shall, within five working days of the respective bid opening dates, advise the Owner in writing of the Designer's opinions as to the sub-bidders' bids and as to which general bidder is the responsible and eligible bidder that has submitted the lowest bid.

- 7.7.2 The Designer shall assist the Owner in the prequalification of prime contractors and subcontractors in the filed sub-bidder or trade contractor scopes of work pursuant to M.G.L. c. 149, §§44D½ and 44D¾ including participation as a member of the Owner's Prequalification Committee.
- 7.7.3 The Designer shall receive all inquiries relating to the bid documents and, when necessary, answer questions by preparing and issuing written addenda. The Owner shall review and approve all such addenda prior to issuance to bidders.
- 7.7.4 When sub-bids are required:
 - (a) Attend sub-bid openings.
 - (b) Assist in reviewing sub-bids with the Owner for completeness and accuracy.
 - (c) Assess sub-bid amounts relative to cost estimates.
 - (d) Assist in checking references of sub-bidders and make written recommendations as to their qualifications, only required for projects in which pre-qualification has not occurred.
 - (e) Issue a letter of recommendation to Owner upon acceptance of sub-bids, identify any categories to be re-bid and reason(s) therefor.
 - (f) Prepare and distribute the filed sub-bid tabulation to all prospective bidders. The tabulation shall be reviewed and approved by the Owner prior to its issuance to bidders.
- 7.7.5 Unless otherwise directed by the Owner, attend and conduct the general bid opening.
- 7.7.6 Review with the Owner and the Owner's Project Manager general bids for completeness and accuracy.
- 7.7.7 Review bidder responses for alternates and make written recommendations as to their acceptance.
- 7.7.8 If the Project has to be re-bid because of a defect in the bid documents prepared by the Designer or in procedures proposed by the Designer, the Designer shall correct the defect and take the necessary actions for re-bidding the Project on proper bid documents without any additional compensation to the Designer.
- 7.7.9 If within three (3) months after approval of Construction Contract Documents, in final form, the bids of the lowest responsible and eligible bidders or negotiated proposals exceed the approved Project Construction Budget, the provisions of Article 4.10 shall apply.

- 7.7.10 If the Owner awards a construction contract for an amount that exceeds the amount established in the Project Construction Budget, such an award will not affect the Fee for Basic Services.
- 7.8 Construction Administration Phase Obligations During Construction: Following the execution of the Owner-Contractor Agreement, the Designer shall undertake certain of the obligations of administering the Owner-Contractor Agreement on behalf of the Owner, provided that Designer shall not be subject to provisions of the Owner-Contractor Agreement that would have the effect of expanding Designer's responsibilities or liabilities under this Contract without Designer's written consent. Services during this phase include, but are not necessarily limited to:
 - 7.8.1 Upon commencement of construction activities for the Work or early bid packages or at times established in Project schedules, the Designer shall:
 - (a) Furnish the General Contractor with information for establishing lines and grades and such supplemental drawings as are reasonably needed to implement the intent of the Construction Contract Documents;
 - (b) With reasonable promptness and in accordance with schedules agreed upon by the Designer and Contractor, observe testing when required under this Contract, and review and act upon samples, schedules, shop drawings and other submissions from the General Contractor;
 - (c) Prepare, maintain and update logs for all submittals;
 - (d) Visit the site at intervals appropriate to the stage of construction, weekly or as otherwise agreed to by the parties, and observe the progress of the Work, issue written progress reports, and attend job meetings, and review and respond to meeting minutes prepared by the Owner's Project Manager, and to determine in general if the Work observed is being built in a manner indicating the Work when completed will be in accordance with approved Construction Contract Documents;
 - (e) Collaborate with the on-site Project Representative of the OPM to identify and monitor issues of concern relative to the progress of the Work, and establish communications processes to help assure that matters of mutual concern are exchanged on a timely basis with one another, the OPM, Commissioning Consultant, and Owner;
 - (f) On a weekly basis, make specific recommendations on rejection of any Work observed by the Designer that fails to conform to the Construction Contract Documents, and observe corrected Work;
 - (g) Require each Subconsultant engaged in accordance with Article 5 to make visits weekly or as otherwise agreed to by the parties during the progress of any work to which that Subconsultant 's services relate, and to report upon it in writing to the Designer;
 - (h) Recommend actions to be taken which may include condemnation or rejection of any work that the Designer determines fails to conform to the Owner-Contractor Agreement;

- (i) Review and recommend appropriate action for proposed requests for changes and where required by the Owner, prepare documents associated with requests for a change in any Construction Contract Documents. Compensation for change order work by the Designer shall be determined in accordance with Article 10;
- (j) Conduct semi-final and final inspections of the Project and report the results of such inspections in writing to the Owner;
- (k) In association with the Commissioning Consultant, review the report by such Commissioning Consultant on the balancing of air and water circulation systems;
- (l) In association with the Commissioning Consultant, review the report by such Commissioning Consultant on the setting and adjustment of automatic controls;
- (m)In a timely manner, decide all questions regarding interpretation of, or compliance with, the Construction Contract Documents, except as the Owner may in writing otherwise determine;
- (n) In association with the Commissioning Consultant, review the recommendations of such Consultant for requirements upon operating and maintenance documents and building user training events and instructional media as established in the Construction Contract Documents; such Consultant or OPM shall coordinate involvement of contracting parties, the Designer, and Owner;
- (o) Furnish the Record Drawings as submitted by the General Contractor in accordance with 7.8.3, and other required documents;
- (p) Assist the Owner in providing the written Contractor Evaluations required of the Owner pursuant to M.G.L. c.149 §44D(7) at the completion of approximately 50% of the Construction Phase on forms prescribed by M.G.L. c.149 §44D(16);
- (q) Perform inspections of the work as necessary to prepare a punch list identifying each incomplete or deficient Work item and performing re-inspections to authorize removal of satisfactorily completed Work items from the punch list, or to determine that the Project is complete. In association with the OPM, a cost shall be assigned to each incomplete or deficient Work item when it has been determined that the Project has reached Substantial Completion; and
- (r) Receive from the General Contractor all maintenance and operating manuals, occupancy permits, guarantees and other similar relevant materials.
- 7.8.2 The Designer shall submit to the Owner's Project Manager within 48 hours all requisitions for payment submitted by the General Contractor in the form required by the Owner. The Designer may establish procedures with the Contractor for advance notification of requisition and/or draft version processing. With respect to each such requisition, the Designer shall certify to the best of its knowledge that the percentage of Work included in the requisition is accurate and that the work performed is in accordance with the Construction Contract Documents. In the event the Designer does not approve the requisition exactly as submitted by the General Contractor, the Designer shall forward it for payment to the Owner's Project Manager dated and signed with corrections and with an accompanying letter of explanation setting forth the Designer's objections and recommended changes. The Designer shall coordinate

- the required visits of its own staff and those of its Subconsultants, to the construction site so as to enable it to submit to the Owner's Project Manager the General Contractor's monthly requisition for payment. Timely payments to the Contractor are required by M.G.L. c. 30, § 39K. Therefore, the Designer shall establish procedures to help assure either immediate mail or messenger delivery of the requisition for payment to the Owner's Project Manager, and shall process requisitions for payment within five business days after receipt of the same, provided the Contractor has submitted a full and complete requisition for payment in the correct form.
- 7.8.3 Prior to issuance of the Certificate of Substantial Completion, the Designer shall obtain from the General Contractor as-built drawings, including drawings showing the actual installation of the site utilities, plumbing, heating, ventilating and electrical work under the Owner-Contractor Agreement, and recording all changes. The Designer shall ascertain that changes authorized by change orders are shown on the General Contractor's as-built drawings, but Designer shall be entitled to rely upon the accuracy and completeness of the Contractor's as-built information, and shall forward such to the Owner as Record Drawings.
- 7.8.4 Issue the Certificate of Substantial Completion of Construction.
- 7.8.5 The Designer shall meet with the Owner monthly during this Phase.
- 7.9 <u>Completion Phase:</u> Upon acceptance of the Certificate of Substantial Completion of Construction by the Owner, the Designer shall thereafter provide the following services:
 - 7.9.1 With respect to a completed Project, preparing a Certificate of Final Completion.
 - 7.9.2 With respect to a punch list, re-inspecting the work up to three times in order to determine that the punch list work is satisfactorily completed.
 - 7.9.3 Reviewing and certifying the Contractor's Application(s) and Certificate(s) for Payment as necessary.
 - 7.9.4 Attending meetings as reasonably necessary in the opinion of the Owner's Project Manager, unless such meetings involve continued discussions of incomplete or deficient work and the Basic Services punch list site visits have been expended. In such instance, the meetings shall be paid for as Extra Services.
 - 7.9.5 Using the as-built information maintained by the General Contractor during construction referred to in Article 7.8.3, and revising the applicable original reproducible drawings and electronic media drawings on the basis of the as-built drawings, provided that Designer shall be entitled to rely upon the accuracy and completeness of the Contractor's as-built information. Upon completion of the required drafting and editing, provide one set of mylar reproducibles, two sets of prints and two (2) electronic version copies to the Owner which shall become the property of the Owner. The cost for printing the mylar reproducibles and two sets of prints are Reimbursable Expenses.
 - 7.9.6 Ten (10) months after the date of substantial completion, performing one (1) site inspection and preparing a list of construction warranty deficiencies. The Designer shall consult with the Commissioning Consultant upon the acceptability of warranty compliance requirements and response actions.

- 7.9.7 Informing the Owner in writing, through the Owner's Project Manager, of all such warranty deficiencies that should be addressed.
- 7.9.8 Performing one (1) site inspection within a further sixty (60) days to see that all such warranty deficiencies have been corrected.
- 7.9.9 Evaluation of Contractor: The Designer shall assist the Owner with providing the written Contractor Evaluations required of the Owner pursuant to M.G.L. c.149 § 44D(7) within 70 days of the date of Substantial Completion for construction, on forms prescribed by M.G.L. c.149 § 44D(16).
- 7.9.10 Two (2) suitably bound legible copies of all original design and quantity calculations including those pertinent to change orders and shop drawings if applicable shall be furnished by the Designer to the Owner at the conclusion of the Owner-Contractor Agreement.

```
THINONALLY LEFT BLANK
```

ATTACHMENT B-2:

Designer Services Contract

Amendment for CM-at-Risk



```
THINONALLY LEFT BLANK
```

7.5 CM at Risk Construction Delivery Method

7.5.1 CM at Risk Prequalification & Selection

- (a) The Designer shall participate as a member of the Owner's CM at Risk Prequalification Committee and CM at Risk Selection Committee pursuant to M.G.L. c. 149A, §§ 5 & 6.
- (b) The Designer shall, when authorized by the Owner, prepare for reproduction and distribution all project design documents, that are required for the solicitation and receipt of qualifications and proposals from CM at Risk firms pursuant to M.G.L. c. 149A, §§ 5(b) & 6(a). The Designer shall prepare all addenda (to include questions from CM at Risk firms and Designer responses), subject to the approval of the Owner. The Designer shall attend a pre-proposal conference, and existing site and building tour if either or both are to be scheduled, taking note of all questions asked. Relevant questions submitted in writing shall be answered by the Designer in conjunction with the OPM by means of written addenda to the RFQ or RFP described below, as required.
- (c) As a member of the Owner's CM at Risk prequalification committee, the Designer shall review and evaluate in conjunction with the Prequalification Committee, the Statements of Qualifications received from CM at Risk firms on the basis of the evaluation criteria established in the RFQ and shall make appropriate recommendations regarding the selection of qualified CM at Risk firms to receive a request for proposals from the Owner in accordance with the provisions of M.G.L. c. 149A, § 5(f).
- (d) As a member of the Owner's CM at Risk selection committee, the Designer shall review and evaluate the RFP's received from prequalified CM at Risk firms on the basis of the evaluation criteria included in the RFP. The Designer shall make appropriate recommendations regarding the evaluation and ranking of RFP's and the conducting of interviews, if any, in accordance with the provisions of M.G.L. c. 149A, § 6(d), and the applicable regulations and procedures promulgated by the Inspector General. If the Selection Committee elects to conduct interviews of the CM at Risk firms, the Designer shall participate in conducting interviews.
- (e) As member of the Owner's CM at Risk Selection Committee, the Designer shall assist the CM at Risk Selection Committee in non-fee negotiations with the CM at Risk until the Selection Committee has reached an acceptable contract with one of the prequalified CM at Risk firms in accordance with M.G.L. c. 149A § 6(e).
- (f) If, at any time, the Owner terminates the Owner-CM at Risk contract, the Designer shall continue to provide the Designer Services required under this Contract with

any substitute CM at Risk procured by the Owner. If, as provided by law, the Owner elects to proceed with the Project pursuant to the provisions of M.G.L. c. 149 (design-bid-build), the Designer may continue to provide Designer Services pursuant to a mutually agreeable amendment to this Contract subject to the approval of the Authority.

7.5.2 <u>Design Review for the CM at Risk Construction Delivery Method</u>

- (a) The Designer shall provide Designer Services in a manner consistent with the CM at Risk Delivery Method, as defined herein, in all Phases of the Project and shall work cooperatively with the CM at Risk, as well as the Owner, OPM, Commissioning Consultant and the Authority to achieve timely completion of the Project within the Project Construction Budget.
- (b) Upon execution of the Owner-CM at Risk Agreement, the Designer shall:
 - 1. meet with the Owner, the OPM and the CM at Risk to discuss issues and to establish procedures for efficient interaction in a cooperative and mutually supportive manner that will permit all parties to perform their contractual obligations. These procedures shall include, but not be limited to: arrangements for the collaboration and coordination between the Designer and the CM at Risk in the preparation and submission of all design phase documents to the Owner; arrangements for discussions concerning all design phase document submittals among the Owner, OPM, CM at Risk and Designer; and arrangements for frequent and productive interactions between the Owner, OPM, CM at Risk and Designer during all the design phases.
 - 2. provide copies of the schematic design drawings, specifications, cost estimates and other submittals to the CM at Risk, to assist the CM at Risk in fulfilling its responsibilities to the Owner. The Designer shall consult with the CM at Risk and provide the CM at Risk with an opportunity to review and comment upon deliverables developed by the Designer during the Schematic Design Phase.
- (c) The Designer shall attend and participate in meetings as necessary with the CM at Risk, the Owner and the OPM to resolve all issues.
- (d) The Designer shall consult with the Owner, the OPM, and the CM at Risk regarding the sequence of delivery of design services; the selection of materials, building systems and equipment; alternative solutions recommended by the CM at Risk when design details affect construction feasibility, schedules, cost or quality; other value engineering comments and recommendations made by the CM at Risk; comments and recommendations concerning the design documents with respect to clarity, consistency, constructability,

- maintainability/operability and coordination among the trades, coordination between the specifications and drawings, compliance with M.G.L. c. 149A for procurement, installation and construction, and sequence of construction, including recommendations designed to minimize adverse effects of labor or material shortages.
- (e) The Designer may be required, as a part of Basic Services if previously agreed with the Owner, to prepare plans and specifications for discrete portions of the Work that can be incorporated into separate bid packages for the various Subcontractors who will construct the Project. Such contracts may be awarded concurrently with other contracts or individually, or at different points in time, which may result in the Designer completing portions of the design after commencement of construction of the Project and/or providing Construction Phase services before completion of all design phase services. The design for each separate bid package shall separately be subject to all requirements applicable to the various phases set forth in this Contract and shall be performed in a manner consistent with the provisions of the Project Funding Agreement, including, but not limited to, the Project Construction Budget and Project Schedule.
- (f) The Designer shall consult with the CM at Risk concerning the ordering and delivery of products and assemblies and shall identify and describe any long lead products or assemblies that need to be priced and pre-ordered to meet the Project Schedule.
- (g) The Designer shall identify and describe any multiple bid packages or fast-tracked construction that will be used and any separate bid packages that will be required.

7.6 Design Development Phase

- 7.6.1 The Designer shall provide the CM at Risk with an opportunity to review and comment upon design documents developed by the Designer during the Design Development Phase. The Designer shall work cooperatively with the CM at Risk throughout the Design Development Phase of the Project to obtain the benefit of the knowledge and experience of the CM at Risk with respect to design review, value engineering, constructability analysis, cost estimating, cost control, scheduling, coordination of bid packages, phasing, and other services and, with the approval of the Owner, the Designer shall thereupon incorporate recommended and mutually accepted changes into its design documents.
- 7.6.2 Upon receipt of an Approval to proceed to the Design Development Phase, the Designer shall meet regularly and as necessary with the Owner, the OPM, the CM at Risk and the Authority. This shall include meeting at least once every other week with the Owner, the OPM and the CM at Risk during this Phase.

- 7.6.3 Upon receipt of an Approval to proceed to the Design Development Phase, the Designer shall update and refine items submitted during the Schematic Design Phase, and shall submit to the Owner, CM at Risk, and the Authority, on or before the date specified in the Project Schedule, and on the basis of the approved Schematic Design Phase Documents, the following deliverables as they are defined in this Article 7.6.3 and as they are further defined in Articles 7.6.4, 7.6.5, 7.6.6, 7.6.7, and 7.6.8:
 - (a) a list of all filings and permits within Designer's scope of services and professional expertise required to implement the design and a schedule of target dates for the procurement of such permits, which list and schedule shall be regularly updated during the term of this Contract;
 - (b) information and documentation within the technical expertise of the Designer and that is necessary for the Owner to file local basic zoning and environmental permits. The Designer, as Extra Services, shall provide information and documentation for the Owner to file Environmental Notification Forms, Environmental Impact Reports, and any other filings for permits that must be filed during the design development phase;
 - (c) soils exploration data, geotechnical and geoenvironmental reports, showing exploratory locations relative to siting of proposed structures;
 - (d) complete design development drawings; outline specifications indicating any filed sub-bid sections and sub-sub trades based on the cost of the work and other documents necessary to specify the size and character of the Project as to siting, landscape, architectural, structural, fire protection, plumbing, heating, ventilating and air conditioning, electrical, ADA/MAAB, product requirements, and other features;
 - (e) quality control documentation demonstrating, without limitation, coordination of: ceiling clearances, mechanical room size, and shaft sizes; specifications and drawings; filed sub-bid work or sections; scheduling; equipment and power; existing and new construction; and phasing;
 - (f) design development drawings which the Designer shall submit for review to the local building official;
 - (g) a life cycle cost analysis to determine which design decisions related to all energy and water consuming devices and overall building operation and maintenance are the most cost effective [M.G.L. c. 149, s. 44M];
 - (h) a construction cost estimate for the design in Uniformat II Level 3 format, with unit rates and quantities supporting each item and reconciled with the detailed construction cost estimate and any updated cost estimates in accordance with Article 7.6.7. The estimate cost shall be projected, to the mid point of the construction period;

- (i) a space measurement analysis for the design verifying that the sum of all program areas in the Project plus all other floor areas in the Project equals the gross floor area of the Project;
- (j) a written summary or summaries comparing the project design, as represented in the design development drawings, specifications and cost estimates with the Final Design Program requirements, and explaining any deviations in writing.
- 7.6.4 Design Development Drawing Requirements: The Design Development drawings shall illustrate and describe the refinement of the design of the Project to a level of detail that is customary and standard, establishing the scope, relationships, forms, size and appearance of the Project by means of plans, sections and elevations, typical construction details, and equipment layouts. Drawings shall delineate locations and elements of Work which may be proposed to be assigned to project construction phases and/or separate bidding packages. Documents shall include, but not be limited to, the following:
 - (a) Site and utility drawings showing;
 - 1. Existing and proposed contours and locations of the proposed building or addition(s). Show entry level elevation and key exterior grades at perimeter. Indicate all retaining walls. Include benchmarks of site if survey is available.
 - 2. All utilities existing and proposed, indicating location, elevation, composition and size e.g., manholes, sewers, hydrants, light standards. Include work by others, e.g., gas and electric utility providers.
 - 3. Roads, laid out parking areas, walks, recreation areas, terraces and other site improvements.
 - 4. Building locations fixed and referenced from main survey baseline, if available.
 - 5. Plant materials with preliminary schedule.
 - (b) Building drawings and other graphic and written requirements with floor plans showing: (minimum scale 1/8" = 1'0");
 - 1. building perimeter with exterior wall thicknesses and overall dimensions;
 - 2. structural grid;
 - 3. plan requirements of mechanical and electrical systems;
 - 4. building core; elevators, stairs, shafts, toilet rooms;
 - 5. interior partitions; appropriate thicknesses and dimensions to fix basic organizations; indicate fire separations, ratings;
 - 6. door swings;
 - 7. floor elevations;

- 8. built-in furniture and equipment; and
- 9. furniture layout concept drawings.
- (c) Roof plans showing:
 - 1. proposed systems type;
 - 2. pitch and drainage patterns;
 - 3. roof drains, gutters and scuppers;
 - 4. skylights, stairs through roof, penthouses, major equipment, chimneys.
- (d) Building sections: One transverse and one longitudinal section. Indicate floor to ceiling heights and floor-to-floor heights. Label all spaces;
- (e) Building elevations showing:
 - 1. full height elevations including roof structures, e.g., mechanical equipment, chimneys, and penthouses;
 - 2. floor elevations, floor-to-floor heights, and overall height related to benchmarks on site plans;
 - 3. all fenestration:
 - 4. column centerlines;
 - 5. principal finish materials indicating major control and expansion joints, and divisions of materials where required;
 - 6. louver and equipment enclosure systems; and
 - 7. exterior grades and topographical features in context.
- (f) Full height wall sections for main elevations and at special conditions. Show foundation and perimeter treatment, wall construction including insulation and supporting structure, fenestration and mechanical penetrations, and floor construction;
- (g) Interior elevations: Major spaces, e.g. library, lobby; and all typical spaces, e.g. classrooms;
- (h) Reflected ceiling plans: show prototypical structural, fire protection, mechanical and electrical information for classrooms and major spaces, including lighting layouts with ceiling heights and material changes;
- (i) Colored interior elevations and perspectives of major and typical spaces;
- (i) Schedules:
 - 1. finish schedule by room types;
 - 2. door schedule by room;
 - 3. window schedule;

4. equipment schedules, e.g., food service, instructional media.

(k) Structural Concepts:

- 1. Foundation plan showing sizes and locations of typical components.
- 2. Framing plans: typical floor framing, roof framing, special framing, show framing at major openings and sizes of members.
- 3. Column locations.
- 4. Preliminary details including floor and roof deck, statements as to methods of lateral bracing and how requirements of earthquake code will be met.
- 5. Details for special and/or incidental structural features, e.g. tunnels, connecting bridges and unique architectural features.
- 6. Connection to existing buildings at foundation and at key points at existing structure if applicable.
- (1) Fire Protection: floor plans indicating wet or dry type systems, hose racks or cabinets and fire department tie-ins. Indicate whether a fire pump will be required and, if so, show location within the building. Show typical sprinkler head layout;
- (m)Plumbing and sanitary systems: floor plans indicating locations of all plumbing fixtures and special features, and approximate location and size of all piping systems and principal items of equipment;
- (n) Heating, Ventilating and Air Conditioning Systems;
 - 1. Show locations and approximate sizes of piping systems, air handling systems and principal items of equipment such as compressors or cooling towers.
 - 2. Indicate space requirements of major equipment and their location in mechanical rooms and fan rooms. Major shafts.

(o) Electrical Systems;

- 1. Calculations showing total electrical load.
- 2. All services including those for special purposes shall be located and indicated.
- 3. Lighting shall be indicated as to type, location and intensities in foot-candles for each special and typical space.
- 4. Switchgear and emergency generator.
- 5. Fire alarm system drawings showing all initiation and signaling devices, control panels, annunciator panels, etc.
- 6. Security system drawings.
- 7. Communications drawings showing chases, major equipment locations and any special distribution requirements.

- 8. CATV/CCTV drawings showing chases, major equipment locations and any special distribution requirements.
- 9. Information Technology drawings showing chases, major equipment locations and any special distribution requirements.
- 7.6.5 Other Consultant's Drawings and Other Graphic and Written Requirements: For special consultants, e.g., kitchen, elevator, library, media room, equipment where appropriate, provide drawings that locate and define the scope of the work. Coordinate with other disciplines. Provide cuts of all major pieces of equipment.
- 7.6.6 Project Manual Requirements (Specifications):
 - (a) Outline Specifications that are to accompany Design Development Drawings shall be prepared to a level of detail that is standard and customary and shall include, but not be limited to, a comprehensive description of the Project and the materials proposed for use in the work. No detailed specifications of materials or workmanship procedures need be included; however, the general scope shall be indicated by CSI MasterFormat as applicable to proposed construction.
 - 1. The Design Development Outline Specification shall also include a comprehensive "BASIS OF DESIGN." The "BASIS OF DESIGN" shall be a narrative description of the Project and shall include all applicable architectural, civil, structural, mechanical and electrical programs and/or systems. Identify all proposed filed sub-bid categories.
 - 2. Project Manual shall include a statement to define Work which is proposed to be included in separate construction phases and/or bid packages.
 - (b) The following is a list of items that shall at a minimum be identified or outlined in this Phase.
 - 1. Site work; clearing, drives, walks, parking areas, fences, excavation, backfill, planting.
 - 2. Footings; on earth, rock, piles, caissons, proposed bearing pressures, boring logs.
 - 3. Foundation walls; type of concrete, reinforcing, type and extent of waterproofing.
 - 4. Footing drains; type, disposal of drainage.
 - 5. Exterior walls: superstructure, type, materials, brick type, alternate cladding, back-up materials, dampproofing material and extent, special features.
 - 6. Roofs; types, vapor barrier, insulation, flashings, all materials.
 - 7. Flashings; general types, all materials, weights, where each type is to be used.
 - 8. Sheet metal; gutters, leaders, others uses, except flashings.
 - 9. Windows; general types, materials, sub-frames, finish, glazing, screens.

- 10. Doors, exterior and interior; types.
- 11. Steps, exterior; including platforms and landings' materials.
- 12. Stairs, interior; including platforms, landings, walls, materials and finishes.
- 13. Framing; wood, concrete or metal systems in accordance with general design.
- 14. Partition construction related to room type;
- 15. Cabinet and casework; types and materials.
- 16. Food Service Equipment; types and materials.
- 17. Furring; lathing, plastering, materials and locations.
- 18. Insulation thermal; types, thicknesses, methods of application and locations.
- 19. Acoustical treatments; types, thicknesses, methods of application and location.
- 20. Interior finishes; materials for floors, walls, bases, wainscots, trim, ceilings, ceiling heights.
- 21. Fire Protection; standpipe systems, sprinkler systems, fire pumps and accessories.
- 22. Water supply; source; location of main to which connection will be made; type of pipe for service main; load requirements; load factors and pressures.
- 23. Sanitary sewers; sewage disposal system, pipe and other materials.
- 24. Storm sewers; storm drainage disposal system (institution or local facility), pipe and other materials.
- 25. Gas main; material, size, location. Interface with utility company.
- 26. Plumbing; systems such as wastes, vents, hot water, cold water, gas, air, oxygen, vacuum, main source of supply, materials for each, water heaters, pumps, thermal insulation fixture quality, all special features.
- 27. Heating, ventilating and air conditioning; type of heating and refrigeration plants, type and capacity of boilers and cooling equipment, fuel, type of burners, fuel storage, heaters, feed water pumps and heaters, thermal insulation, type of heating medium, supply and return piping, radiation, unit heaters, radiant heating, principal air conditioning equipment types, special features, supply, return and exhaust ductwork.
- 28. Electric work; service connection, location, institution or public utility, overhead or underground, transformers including type and location, types of conduit and wiring, types of fixtures, location of main switchboard, radio, fire alarm, telephone, public address, emergency lighting and wiring, emergency or other generators, special features, including Master TV, information retrieval and/or data processing system.

- 29. Elevators, dumbwaiters and platform lifts; capacities, speed, travel in feet, landings, operation, controls, platform sizes, machine type and location, car and entrance finishes, signals.
- 30. Other built-in equipment, types and materials.
- 31. Special features.
- 7.6.7 Construction Cost Estimate Requirements The Designer shall provide a construction cost estimate in Uniformat II Level 3 format with aggregated unit rates and quantities supporting each item referenced in Article 7.6.6(b). The estimate cost shall be projected, to the mid point of the construction period.

The Designer shall review its construction cost estimate in comparison with the detailed construction cost estimate, and any updated cost estimates, provided by the CM at Risk and/or OPM and shall work in good faith and in cooperation and coordination with the CM at Risk and/or OPM to reconcile any differences between the construction cost estimates, to clarify assumptions upon which the cost estimates are based and to address any concerns or questions with the cost estimates that are raised by the Owner, the OPM, the CM at Risk, or the Authority. If the Designer is unable to reconcile all differences between the two construction cost estimates with the CM at Risk, then the Designer shall provide a detailed explanation of the differences to the Owner. If, in any case, the agreed-upon, reconciled construction cost estimate exceeds the Project Construction Budget, the Designer shall cooperate with the Owner, the OPM, and the CM at Risk in identifying, specifying and recommending changes in materials, equipment, component systems and types of construction, or other adjustments in the scope or materials selections for the Project, including contingencies or alternative bid items, so as to facilitate revision of the design of the Project to reduce the cost of construction so as to comply with the authorized Project Construction Budget.

Cost estimate data shall be organized to identify elements of project work which may be proposed to be advanced under separate construction phases and/or separate bidding packages. When so proposed, estimates shall develop cost data relative to corresponding bidding and work execution dates established in project schedules.

7.6.8 Reports, drawings, specifications, cost estimates and other design development submittals shall be subject to the written approval of the Owner and the Authority. Unless a lesser number is requested by the Owner, the Designer shall submit to the Owner for approval six (6) copies of Design Development drawings, specifications, cost estimates, and other submittals. Two (2) copies shall be submitted to the Authority by the Designer. The Designer submit to the CM at Risk one copy (1) of Design Development drawings, specifications, cost estimates and other submittals to assist the CM at Risk in fulfilling its responsibilities to the Owner.

- 7.6.9 The Designer shall present and explain the Design Development submittal to the Owner and the Authority and at a local public meeting scheduled by the Owner, if any such meeting is scheduled or in conference.
- 7.6.10 The Designer and its Subconsultants shall collaborate with the Authority's Commissioning Consultant to develop design criteria which will support the purposes of building commissioning and energy/resources conservation concepts as commonly understood and as prescribed by the Commissioning Consultant.

7.7 Construction Documents Phase:

In addition to the requirements specified in the RFS (Attachment B), upon receipt of an Approval to proceed with the Construction Documents Phase of the Project from the Owner, the Designer shall do the following:

- 7.7.1 The Designer shall provide the CM at Risk with an opportunity to review and comment upon design documents developed by the Designer during the Construction Documents Phase. The Designer shall work cooperatively with the CM at Risk throughout the Construction Documents Phase of the Project to obtain the benefit of the knowledge and experience of the CM at Risk with respect to design review, value engineering, constructability analysis, cost estimating, cost control, scheduling, coordination of bid packages, phasing, and other services and, with the approval of the Owner, the Designer shall thereupon incorporate recommended and mutually accepted changes into its design documents.
- 7.7.2 The Designer shall meet regularly and as necessary with the Owner, the Authority, the OPM, the CM at Risk and the Commissioning Consultant. This shall include meeting with the Owner at least twice per month (or more frequently if needed) during this Phase.
- 7.7.3 Based on the submittals approved in the Design Development Phase of the Project, the Designer shall update and refine the items previously submitted and shall submit the following to the Owner, the CM at Risk, and the Authority on or before the date and time specified in the Project Schedule:
 - (a) Construction documents progress submittals as follows:
 - 1. a 60% Construction Documents Submittal, with deliverables as defined in Article 7.7.4:
 - 2. a 90% Construction Documents Submittal, with deliverables as defined in Article 7.7.5;
 - 3. a Final Construction Documents Submittal, with deliverables as defined in Article 7.7.6;
 - 4. a Bid Documents Submittal, with deliverables as defined in Article 7.7.7

- (b) As a part of each of the submittals required under Articles 7.7.4, 7.7.5, and 7.7.6, an updated work plan and recommended updates for incorporation into the Project Schedule by the OPM;
- (c) As a part of each of the submittals required under Articles 7.7.4, 7.7.5, and 7.7.6, a report on the status of environmental, zoning, planning, building code, and ADA/MAAB approvals and permitting processes and a certified list of all required testing and all required permits identified in 7.6.3 (a).
- (d) All submittals by the Designer shall be subject to the written approval of the Owner, which approval shall not be unreasonably delayed, withheld, conditioned, or denied. Unless a lesser number is requested by the Owner or is specifically provided hereinafter, the Designer shall furnish to the Owner for approval six (6) sets of the drawings, specifications, construction cost estimates and all other submittals. Unless a lesser number is specifically provided hereinafter, the Designer shall furnish two (2) sets of said drawings, specifications, construction cost estimates and all other submittals to the Authority and shall furnish one (1) set thereof to the CM at Risk. The Designer shall also furnish to the Owner, the Authority, and the CM at Risk electronic media copies of the foregoing drawings and documents in such form as may be required by the Authority.

7.7.4 The 60 Percent Construction Documents Submittal:

- (a) The Designer shall provide, on or before the date and time specified in the Project Schedule, a 60 % Construction Documents Submittal (60% CD Submittal), which shall include:
 - 1. Construction Documents and other deliverables, as defined in this Article 7.7.4 and as further defined in Articles 7.7.3, 7.7.8, 7.7.9, and 7.7.10, advanced to a level of intermediate (60 percent) completion, and incorporating corrections to indicate compliance with Owner and Authority review comments related to prior submittals.
 - 2. In instances where the Designer takes exception to the Authority's previous review comments on the Design Development submittal, a written statement explaining its position.
 - 3. The Basis of Design that accompanied the Outline Specifications in the Design Development Phase shall be updated and expanded to include all proposed architectural, structural, fire protection, plumbing, mechanical, electrical, civil, and landscape design concepts for the Project.
 - 4. A space summary, in the form and format prescribed by the Authority, that sets forth the current space calculations and totals and certifies that said space calculations and totals are in compliance with those previously authorized by the Authority in the Project Funding Agreement.

- 5. Keying of graphics shall be sufficient to allow a reviewer to make his or her way through the set.
- 6. A list of all drawings related to the Project.
- 7. A materials selection statement identifying typical interior and exterior surfaces and their materials.
- 8. A color theory statement indicating proposed paint colors and material selections for typical and special spaces and why they have been selected and how these selections relate to surrounding materials and colors.
- 9. Large scale plans of all mechanical and electrical spaces with major equipment indicated.
- 10. Project Manual, including all sections to be included in final technical specifications, developed to include a list of all materials in the building with their manufacturers. Identify all specifications sections which need to be filed sub-bid.
- 11. Identify all proposed bid alternates by inclusion in a project manual section to be titled "Alternates." Alternates shall be listed in sequence as approved by the Owner. Work required under bid alternates shall be described and/or drawn, as appropriate, to clearly define the design criteria and extent of work involved for implementation of the bid alternate. In each instance, the existing conditions and/or new design criteria for base bid work shall also be described and indicated in documents.
- 12. Code analysis: Provide a building code analysis. Any deviation from methods of compliance described in earlier submittals shall be indicated. Code analysis shall identify its preparer, code edition referenced, and include a comprehensive description of operative building code provisions, with floor plans showing fire separation types, area calculations, egress capacity for exits and exitways, and any special features required to comply.
- (b) As a requirement of the 60% CD Submittal, and in accordance with the provisions of this paragraph and Article 7.7.10, the Designer shall provide a construction cost estimate prepared using the Uniformat II Classification to Level 3, the CSI MasterFormat 6-digit format to Level 3 and MGL c.149 §44F (filed sub-bid) format including quantities of all materials and unit prices of labor, equipment, and materials as well as a cost estimate for each item of work, for review by the Owner, the CM at Risk and Authority. The Designer shall submit said construction cost estimate separately, as a supplement to the 60% CD

Submittal, no later than twenty-one days after the submission of the 60% CD Submittal described in Article 7.7.4(a). The development of said construction cost estimate shall under no circumstances delay the timely submission of the remainder of the 60% CD Submittal.

7.7.5 The 90 Percent Construction Documents Submittal:

- (a) The Designer shall provide, on or before the date and time specified in the Project Schedule, a 90 % Construction Documents Submittal (90% CD Submittal), which shall include:
 - 1. Construction documents and other deliverables as defined in this Article 7.7.5 and as further defined in Articles 7.7.3, 7.7.8, 7.7.9, and 7.7.10, advanced to a level of substantial (90 percent) completion, and incorporating corrections to indicate compliance with Owner and Authority review comments related to prior submittals.
 - 2. A space summary, in the form and format prescribed by the Authority, that sets forth the current space calculations and totals and certifies that said space calculations and totals are in compliance with those authorized by the Authority in the Project Funding Agreement.
 - 3. Interior Materials Color Boards, including samples of principal interior materials, labeled and mounted to indicate locations.
 - 4. Final structural and energy design calculations.
 - 5. A statement confirming that the Owner has been provided with structural design drawings, specifications, and calculations sufficient to enable execution of an independent structural peer review process, as defined in the Massachusetts Building Code, as amended (this requirement is applicable, to satisfy Authority requirements for all school construction projects having a floor area in excess of 10,000 square feet). The Designer shall have advised the Owner of this requirement in writing not less than sixty (60) days prior to delivery of the 90% CD Submittal in order for the Owner to arrange for the services of an Independent Structural Peer Reviewer. Upon reaching 90 percent completion of construction documents, Designer's structural engineering consultant shall have reached a level of 100 percent completion of its construction documents to enable advancement of the independent structural peer review.
 - 6. The Designer and its consultants shall fully cooperate with the Independent Structural Peer Reviewer in the process. The Designer shall obtain a copy of the Independent Structural Engineering Review report and submit same to the

- Owner and the Authority at the time of completion of the remainder of the construction documents at the level of final completion.
- 7. In instances where the Designer takes exception to any of the Authority's 60% CD Submittal review comments, a written position statement explaining the Designer's position on its exceptions to said review comments.

7.7.6 Final Construction Documents Submittal:

- (a) The Designer shall provide, on or before the date and time specified in the Project Schedule, a Final Construction Documents Submittal, which shall include:
 - 1. Construction documents and other deliverables as defined in this Article 7.7.6 and as further defined in Articles 7.7.3, 7.7.8, 7.7.9, and 7.7.10, advanced to a level of final (100 percent) completion, and incorporating corrections to indicate compliance with Owner and Authority review comments related to prior submittals.
 - 2. a final construction cost estimate, in accordance with the provisions of this paragraph and Article 7.7.10, based on 90% Construction Documents, including cost estimates for general conditions, overhead and profit, insurance, bonds, and all other items; and allowances expressed as percentage rates for design contingencies and construction contingencies and escalation to the bid date; and other mutually agreed upon contingencies. The final construction cost estimate shall be prepared in Uniformat II Elemental Classification to Level 3 (Sections A-G inclusive), the CSI MasterFormat to Level 3 and M.G.L. c.149, §44F (filed sub-bid) format and shall be complete with a single line description for each item with the detailed unit rate or item cost buildup provided in each case.
 - 3. complete construction drawings and specifications, certified by the Designer as having satisfied the firm's quality control review process as previously confirmed with the Owner, in sufficient detail to permit fixed-price bids in open competition for construction of the Project when documents have been approved for issuance for bidding.
 - 4. no later than at the 100% stage of completion of the final drawings and specifications, two sets of the final drawings and specifications that shall be provided to the local building official to be signed and stamped "Approved" by the local building official; two sets of plumbing drawings and specifications that shall be provided to the local plumbing inspector to be signed and stamped "Approved" by the local plumbing inspector; two sets of the fire protection, HVAC, and electrical construction documents that shall be provided to the local fire official to be signed and stamped "Approved" by the local fire official; two sets of the electrical construction documents that shall

- be provided to the local electrical inspector to be signed and stamped "Approved" by the local electrical inspector. Notwithstanding the foregoing, the Owner acknowledges that building officials, department inspectors, and fire officials have varying policies on approvals and submittal procedures, and the only obligation of the Designer in this regard is to promptly make the submittals described herein and assist the Owner or CM at Risk in receiving the approvals to the extent available.
- 5. at the 100 percent stage of completion of final drawings and specifications, a written summary comparing the final construction drawings and specifications and final estimated construction cost with the Final Design Program requirements and submittals made during the Design Development Phase and earlier in the Construction Documents Phase, explaining any significant deviations.
- 6. In instances where the Designer takes exception to any of the Authority's 90% CD Submittal review comments, a written position statement explaining the Designer's position on its exceptions to said review comments.
- 7. The Independent Structural Engineering Peer Review Report obtained from the Independent Structural Engineering Peer Reviewer referenced in Article 7.7.5(a)5. The Designer shall include a certification statement from the project structural engineer designer of record to acknowledge receipt of the Report and to indicate response actions pursuant thereto. The Designer shall also forward a copy of said Report to the Building Inspector
- 8. A certification that all applicable local, state and utility officials have been contacted by the Designer regarding each utility connection and that the persons responsible for permits or connection approval have agreed to the systems' use.

7.7.7 Bid Documents Submittal:

- (a) The Designer shall provide, on or before the date and time specified in the Project Schedule, a Bid Documents Submittal which shall include:
 - 1. Construction documents and other deliverables as defined in this Article 7.7.7 and as further defined in Articles 7.7.3, 7.7.8, and 7.7.9, incorporating corrections to indicate compliance with Owner and Authority review comments related to prior submittals.
 - 2. From the construction drawings and specifications approved by the Owner, incorporating such changes as the Owner or the Authority requires, a set of reproducible black and white drawings and original specifications on high quality white bond paper, single-sided, properly packaged, suitable for

reproduction, stamped and signed by all disciplines, that shall be prepared by the Designer and transmitted to the Owner; which documents shall become the property of the Owner as provided under Article 16. Other suitable reproducible media, having the same content shall be substituted, when so directed or authorized by the Owner.

3. Upon receipt of Owner authorization to advance to reproduction the approved documents for distribution to bidders and, upon reproduction thereof, the Designer shall promptly submit complete sets of bid documents to the Owner (two sets), the CM at Risk (one set) and the Authority (one set - half size for Drawings). Any subsequent addenda shall be promptly submitted to the Owner, the CM at Risk, and the Authority.

7.7.8 <u>Drawing Requirements</u>:

The documents prepared during the Construction Documents Phase shall set forth the requirements for construction of the Project to a level of detail that is customary and standard and shall include, but not be limited to:

- (a) General information showing drawing index, symbols, abbreviations, notes, location map.
- (b) Site drawings shall be complete to define the extent and detail of site work. Show the following:
 - 1. Layout and location of all proposed work including buildings, structures, retaining walls, parking, walls and all other site improvements, with details.
 - 2. Existing and proposed grades and contours including floor elevations, existing structures and topography, survey base line, bench marks and boring locations.
 - 3. Landscaping and planting.
 - 4. All utility service lines, systems and structures for electricity, gas, oil, water, steam, telephone, CATV, fire alarm, sanitary and storm drainage including size, composition, grades and directions of flow.
 - 5. Contract Limit Line and Storage Area for construction materials.
 - 6. All existing foundations, obstructions and other physical characteristics of the site which may affect the construction work.
 - 7. Site survey.
 - 8. Cuts of benches, light standards.
- (c) Demolition drawings and temporary work required.
- (d) Architectural drawings shall include at a minimum:

- 1. Floor plans of each floor, including basement and lofts or attic with room and corridor dimensions, wall thicknesses, column locations, floor elevations, mechanical and electrical openings, door and window designations, partition types, floor materials, built in furniture and equipment, keyed to other architectural drawings. All rooms numbered.
- 2. Large scale floor plans where required to illustrate detailed requirements of rooms.
- 3. Large scale plans showing key areas e.g. lobby, special spaces. Indicate surface materials. (minimum scale $\frac{1}{4}$ " = 1'-0")
- 4. Roof plans showing openings, drainage, slopes, expansion joints and all projections, including equipment.
- 5. Key plans on all floor plans and section drawings, where appropriate.
- 6. Building Sections as required to show spatial organization of building but no less than one longitudinal and one transverse.
- 7. Building elevations. All building elevations shall be fully developed, and hidden elevations shall be shown. Elevations shall be shown in a sequence as unfolded from a certain point.
- 8. Full height wall sections indicating dimensions, flashing, anchorage, reinforcing, coursing, cladding, and all other conditions at wall, roof, foundation, interior floors.
- 9. Exterior details, for roofing, flashing, expansion control and construction joints, waterstops and other details showing all conditions both vertical and horizontal, including schedules.
- 10. Door, window, entrance, and storefront, schedules, and details.
- 11. Vertical circulation plans, sections and details including stairs, elevators, conveyors, dumbwaiters.
- 12. Interior elevations of all significant and typical spaces.
- 13. Interior details including casework, paneling surfacing and acoustical treatment.
- 14. Reflected ceiling plans coordinated with fire protection, mechanical and electrical drawings, and ceiling details.
- 15. Schedules (clearly define new or existing)
 - a. Doors
 - b. Equipment, e.g. for services
 - c. Partitions
 - d. Finishes
- (e) Structural drawings shall indicate the following:
 - 1. Indicate or refer to location of geotechnical exploration data and reports related thereto.
 - 2. Foundation plans with bottom grades showing layout of all footings, walls, slabs on grade including reinforcing, grade beams, and columns; include design soil bearing pressures and live loads.

- 3. Floor and roof plans of structural systems including framing, grades of finished floors and depressed areas, with locations and dimensions for all openings. Also indicate design floor loads.
- 4. Complete foundation wall elevation and typical sections, with reinforcing indicating location, dimensions and grades for all footings, steps and wall openings.
- 5. Complete details and sections with dimensions for all construction including expansion and construction joints, reinforcing and other embedded items.
- 6. Schedules (with dimensions) for all lintels, beams, joists, and columns.
- 7. Unless detailed on the Drawings, the following information shall appear in the general notes: class and 28 day strength of concrete for each portion, structural steel and concrete reinforcing design stresses for each type of structural member, concrete cover for each type of structural member, shrinkage and temperature steel requirements, reinforcing laps for main reinforcing and temperature steel; bendpoint, cutoff, and hook locations for all members, minimum beam and lintel bearing. Reinforcing steel fabrication shall be in accordance with most recent ACI, "Manual of Standard Practice for Detailing Reinforced Concrete." Structural steel fabrication shall be in accordance with the AISC "Manual of Steel Construction."
- (f) Fire protection drawings shall indicate standpipe systems, sprinkler systems, suppression systems, access panels, fire pumps, accessories, and piping. All piping, equipment, fixtures and devices shall be located and sized. Design criteria shall be provided on the drawings in accordance with NFPA requirements.
 - 1. Fire protection work, other than site work, shall not be combined on the same sheets with the Plumbing, HVAC, Electrical, or other drawings except with the prior approval of the Owner.
- (g) Plumbing drawings shall indicate the following:
 - 1. All work done by the Plumbing Subcontractor, which includes all water, gas, air, vacuum, medical gases, sanitary and storm wastes, and accessories. Include foundation drain lines unless established as the work of the CM at Risk and shall not be indicated on the Plumbing Drawings. Site utilities shall be indicated on the utility drawings.
 - 2. Plumbing work, other than site work, shall not be combined on the same sheets with the Fire Protection, HVAC, Electrical, or other drawings except with the prior approval of the Owner.
 - 3. Trapping and venting of all plumbing fixtures including floor drains.
 - 4. Water and gas supply sources, storm and sanitary discharge mains.
 - 5. All piping shall be carefully sized and all sizes shall be indicated on drawings and riser diagrams. Indicate all directions of flow and pitch on piping.

- 6. All accessories, valves, fixtures including all drinking fountains, grease traps for kitchen waste and all necessary panels, identified as to type and size.
- 7. All piping and connections required for other trades (e.g., kitchen equipment, HVAC make-up water, etc.).
- 8. Acid waste, vents and neutralization systems for laboratories.
- 9. Plumbing Legend and/or graphical symbols on the first sheet of the Plumbing Drawings in accordance with the American National Standards Institute (ANSI).
- 10. Plumbing riser diagrams for structures two or more stories in height above the ground level.
- 11. Domestic water booster pumps, boiler feed water, meter location, hose bibbs, and wall hydrants.
- 12. Domestic hot water: storage tanks, piping material, hanger details.
- 13. All required access panels shall be indicated.
- 14. Backflow preventors and cleanouts. Verify that access and clearance provisions for periodically inspected devices, including backflow prevention, are adequate to satisfy requirements of inspecting agencies.

(h) Heating, Ventilating and Air Conditioning Drawings shall indicate the following:

- 1. HVAC work, other than site work, shall not be combined on the same sheets with Fire Protection, Plumbing, Electrical, or other drawings except with the prior approval of the Owner.
- 2. All piping and ductwork systems shall be located and sized. All ductwork shall be shown double line.
- 3. All systems shall be sized at all reductions and riser diagrams of piping and duct systems shall be indicated.
- 4. All directions of flow and pitch on piping, and direction of flow, volumes for duct systems shall be indicated.
- 5. All equipment shall have sufficient servicing and/or replacement space indicated on drawings.
- 6. All equipment, accessories, valves and dampers with all necessary access panels, identified as to type and size. Access panels, where required for access to valves and dampers shall be indicated on drawings.
- 7. Cooling system pumps, chillers, cooling towers, air handling units, ductwork system and dampers, fan details, temperature control system, air and hydronic balancing equipment, and schedules shall be indicated.
- 8. Cooling tower design shall be indicated on the drawings showing site location, elevations and floor plan of equipment layout and typical flow diagram as related to the total HVAC system.
- 9. All fire and smoke dampers, access panels and doors.
- 10. Mechanical room designs:

- a. Vent pipes for safety valves, relief valves, back pressure valves and tanks shall be extended above flat roofs in accordance with all governing authorities.
- b. In all designs for boiler and refrigeration plants, include a complete floor plan indicating location of all major mechanical equipment and sufficient service space.
- c. In designs of new and/or replacement boiler and refrigeration plants, provide a flow diagram detailing steam or hot water distribution systems, return systems, including all existing equipment and their function, as well as any proposed expansions with all necessary instrumentation and controls.

(i). Electrical Drawings shall indicate the following:

- 1. Site utilities shall be indicated on separate electrical site drawings, unless ample space is available on common site for utility drawings.
- 2. Electrical work, other than site work, shall not be combined on the same sheets with Fire Protection, Plumbing, HVAC, or other drawings except with the prior approval of the Owner.
- 3. General arrangement: Outline layout of each floor. Typical sections through the structure shall be indicated when necessary to define requirements, floor and ceiling heights, elevations, and type construction, including concrete pads shall be indicated. Indicate interface with other systems. Identify any work by other trades.
- 4. Interior lighting system: Light fixture schedules, circuiting location and mounting heights of all fixtures, receptacle and switch outlets, sizes and types of all lamps, conduits, all other accessories and riser diagrams shall be indicated on drawings. Indicate details and method of supporting electrical fixtures and conduits. Designer shall specify that all electrical lighting fixtures be supported from the building structure, and shall be independent of ducts, pipes, ceilings and their supporting members. Comply with seismic design criteria.
- 5. Power system: Locations, types and method of control for all motors, heaters, appliances, controllers, starters, branch circuits, feeder conductors and conduits. Indicate riser diagrams. Show details and indicate method of supporting electrical conduit. For larger projects, thermostats and control wiring are normally covered under the HVAC sub-contract, assure coordination.
- 6. Fire Alarm, Data, Communications, CATV/CCTV Systems: Locations and types of all devices, outlets and equipment, service connections, wiring diagrams, all other essential details.
- 7. Services: Location and details of all services, whether overhead or underground, feeder sizes, plans and elevations of switchgear and

- transformers, metering and service switchboard arrangements, wiring and ground fault diagram and bus ducts.
- 8. General and sub-stations: Location, size, method of connection and protection of all generators, transformers, exciters, motor generators, switch gear, and associated equipment, current characteristics and equipment capacities. Indicate equipment connections by means of one line and/on wiring diagrams and schedule all major items of equipment and all instruments.
- 9. Underground work: The size and locations of manholes and types of cables, number, size, and location of ducts, locations, sizes and types of cable supports, fireproofing, duct line profile, and one line diagram of connections. All underground chambers, including manholes and pull-boxes, shall be constructed of cast in place or one piece pre-cast concrete.
- 10. Pole line work: if required as contract work, indicate location, length, treatment and class of poles, guying, cross arms, insulators, circuiting, transformers, protective and switching devices, lightning arresters, special structures, diagrams, current characteristics and grounding.
- 11. Exterior lighting: Location, size, and type of transformers, luminary, poles, light standards, cables, ducts, and manholes, details of control equipment and connection diagrams.
- 12. Emergency system details including transfer switch, type of fuel.
- 13. One line diagram indicating load KVA, and available short circuit amperes at each transformer, switchboard, distribution panel board, branch circuit panel board, and at major pieces of equipment.
- 14. Riser diagrams for all systems.

7.7.9 Project Manual Requirements:

- (a) The format for the Project Manual, including its technical specifications shall be in accordance with the current CSI MasterFormat with separate sections for each of class of work required by M.G.L. c. 149 §44F.
- (b) The following general information applies to the development of final Specifications:
 - 1. Describe the extent of the work, the materials and workmanship, and include the work under the proper section. If any portion of the work included in a section of the Specifications is to be performed by a trade covered by another section, there shall be clear and distinct cross-referencing between the sections. Merely to state "by others" is not acceptable.
 - 2. For each item of material or equipment, the specifications shall provide for a minimum of three named brands of material or equipment and the words "or equal" or a description of material or equipment which can be met by a minimum of three manufacturers or producers, and the words "or equal." Proprietary products shall not be specified except as provided by M.G.L. c.

- 30, § 39M; however, when they are specified, proprietary specifications are subject to the "or equal" provisions of M.G.L. c.30, § 39M.
- 3. Specify materials mined or manufactured in Massachusetts first and the United States of America second whenever possible.
- 4. Do not use general clauses intended to be all-inclusive in lieu of complete descriptions.
- 5. Do not duplicate standard requirements that are contained in the contract form.
- 6. Use consistency throughout. The word "will" shall be used to designate what the Owner, Authority, Owner's Project Manager, Commissioning Consultant, or the Designer can be expected to do, and the word "shall" shall be used to designate what is mandatory for the CM at Risk or subcontractors to do.
- 7. Use the same term throughout for the same subject and the term shall be the same as that used on the drawings.
- 8. Do not use the term "etc."
- 9. Avoid such terms as "to the satisfaction of the Designer," "as directed by the Designer," "as approved" and "as required."
- 10. Specify work in appropriate Sections according to local trade jurisdiction.
- 11. Avoid the use of the following symbols:

Symbol	<u>Use Instead</u>
#	number, no., or pounds
%	percent
"	inch or in.
X	by
•	feet or ft.
O	degree
/	per or at

- 12. In sections for which filed sub-bids are required, refrain from using such terms as "the Contractor," the "Heating Contractor," or "the Plumbing Contractor," but where necessary for clarity refer to the "HVAC Subcontractor," the "CM at Risk" and the like.
- 13. Do not give numbers both in words and figures. Numbers less than 10 shall be written in words, 10 and higher numbers shall be written in figures. In expressing dimensions, figures such as 2 in., 16 in., 7 ft., 6 in., shall be used.
- 14. Each filed sub-bid section shall detail all labor and materials required by the particular sub-trade and list, by number, those drawings (and only those drawings) indicating work of that sub-trade. In addition, list drawings indicating work of a particular trade that appears on drawings that are not customarily included in the work of the trade, when applicable.
- 15. Do not specify that a product or system shall require prequalification or advance approval prior to bidding.

- 16. Established unit price items shall be used for work categories which cannot be ascertained for exact quantities in bid documents (e.g. earthwork removal and/or replacement items). In such cases, the Designer shall establish ranges of quantities with associated unit price values for each range. Unit price values shall be established for added work, for deleted work, for base bid quantities when conditions so-suggest. Unit price values shall be ascertained through consultation with cost estimators and the CM at Risk, be current, equitable, and well defined as to elements of work, overhead, like issues to be encompassed. Established unit prices shall be published within the applicable technical specification sections, and referenced from general conditions as being operative as the basis for determining values to be used for payment or recovery for change order work.
- 17. Staging, scaffolding, cutting and patching, refuse collection and disposal, demolition work and cleaning task, allocation policy and proposed language shall be carefully assigned to avoid duplication or omission.
- 18. A final draft of Project Advertisement, Notice to Bidders, Instructions to Bidders, Contract Forms, General Conditions, Supplementary General Conditions, and other "front end" documents shall be included in the 90% construction documents submittal, along with a final version of all text to appear in Division 1, General Requirements. The Designer may defer insertion of final advertising / bid dates and wage rates, understanding that they are to be established and inserted immediately prior to release of documents for bidding.

7.7.10 <u>Construction Cost Estimate Requirements</u>

- (a) The Designer shall provide the construction cost estimates described in Articles 7.7.4 and 7.7.6 in accordance with the following provisions:
 - 1. The Designer shall review its construction cost estimate in comparison with the detailed construction cost estimate, and any update cost estimates, provided by the CM at Risk and shall work in good faith and in cooperation and coordination with the CM at Risk to reconcile any differences between the cost estimates, to clarify assumptions upon which the cost estimates are based and to address any concerns or questions with the cost estimates that are raised by the Owner, the OPM, the CM at Risk or the Authority. If the Designer is unable to reconcile all differences between the two construction cost estimates with the CM at Risk, then the Designer shall provide a detailed explanation of the differences to the Owner and the Authority. If, in any case, the agreed-upon, reconciled construction cost estimate exceeds the Project Construction Budget, the Designer shall cooperate with the Owner, the OPM, and the CM at Risk in identifying, specifying and recommending changes in materials, equipment, component systems and types of construction, or other adjustments in the scope or materials selections for the Project, including

- contingencies or alternative bid items, so as to facilitate revision of the design of the Project to reduce the cost of construction so as to comply with the Project Construction Budget.
- 2. Cost estimate data shall be organized to identify elements of project work which may be proposed to be advanced under separate construction phases and/or separate bidding packages. When so proposed, estimates shall develop cost data relative to corresponding bidding and work execution dates established in project schedules.
- 3. Cost estimates shall be projected to the mid point of the construction period.
- 4. The summary sheets shall contain the following:
 - a. The date that the estimate was prepared. (Value Date).
 - b. The anticipated bid date.
 - c. The project and contract number.
 - d. The title and location of the project.
 - e. The name of the Designer.
 - f. The name of the Estimator.
 - g. The site work cost (including all utilities).
 - h. The building cost (including fixed equipment).
 - i. The estimated construction cost of each Phase of the work, totaled.
- 7.7.11 The Designer shall participate in a final review of the Construction Documents with the Owner, the Owner's Project Manager, the Commissioning Consultant, and the CM at Risk, and the Designer shall incorporate such changes as are necessary to satisfy the Owner's review comments.
- 7.7.12 Guaranteed Maximum Price ("GMP")
 - (a) When the Construction documents are 60% complete as determined by the Owner, or at such later time as may be designated by the Owner, the Designer shall prepare a fully coordinated set of the then-current Construction Documents, which shall be delivered to the CM at Risk and shall be the basis of the CM's GMP proposal.

- (b) The Designer shall provide technical assistance to the Owner and the OPM in the negotiation and development of a GMP with a CM at Risk in accordance with M.G.L. c. 149A, §7, that is acceptable to the Owner. The Designer shall meet with the Owner, OPM, and the CM at Risk to review the GMP proposal and the written statement of its basis. If the GMP proposal submitted by the CM at Risk exceeds the Construction Budget, the provisions of Articles 4.10.4 and 4.10.5 shall apply.
- (c) The Designer shall provide technical assistance to the Owner and the Owner's Project Manager in the negotiation, preparation and execution of any amendments to the Owner-CM at Risk contract, including, but not limited to, the Guaranteed Maximum Price ("GMP") amendment pursuant to M.G.L. c.149A, § 7 and any separate amendment for any construction work commenced before execution of the GMP amendment pursuant to M.G.L. c.149A, §7(b)(3).

7.8 Bidding Phase

- 7.8.1 The Designer shall, when authorized by the Owner, prepare for reproduction and distribution the construction bid documents required for the solicitation and receipt of statements of qualifications and bids from Trade Contractors. The Designer shall prepare all addenda (to include bidder questions and Designer responses), subject to the Approval of the Owner. The Designer shall attend the pre-bid conference if one is scheduled, taking note of all questions asked. Relevant questions submitted in writing shall be answered by the Designer by means of written addenda to the bid documents as required. The Designer shall attend each bid opening of the Trade Contractors (and of other bidders if necessary) and shall, within five working days of the respective bid opening dates, advise the Owner in writing of the Designer's opinions as to the bids of Trade Contractors (and of other bidders if necessary).
- 7.8.2 The Designer shall receive all inquiries relating to the bid documents and, when necessary, answer questions by preparing and issuing written addenda. The Owner shall review and approve all such addenda prior to issuance to bidders.
- 7.8.3 There may be multiple bid packages for the Project. Multiple bid packages may be assembled and bid concurrently or consecutively as a portion of the Project. Portions of the Project may be bid separately from other portions. The Designer shall appropriately staff and structure its design and construction phase performance to assist the Owner in the preparation, issuance, bidding and negotiation, if any, of so-called early bid packages as provided in G.L. c. 149A, § 7(b)(3).
- 7.8.4 If the Project has to be re-bid, or the GMP Amendment must be re-negotiated and amended because of a defect in the bid documents prepared by the Designer or in procedures proposed by the Designer, the Designer shall correct the defect and take the necessary actions for re-bidding the Project on proper bid documents without any additional compensation to the Designer.

- 7.8.5 The Designer shall review alternates and make written recommendations to the Owner as to their acceptance.
- 7.8.6 If the Owner executes a GMP Amendment for an amount that exceeds the amount established in the Project Construction Budget, such an award will not affect the Fee for Basic Services.

7.8.7 <u>Trade Contractor Selection Process</u>

- (a) Trade Contractor Prequalification pursuant to M.G.L. c. 149A, §8(c)
 - 1. The Designer shall participate as a member of the Owner's Trade Contractor Prequalification Committee established by the Owner pursuant to M.G.L. c.149A, § 8(b).
 - 2. The Designer shall review the information provided by the CM at Risk describing the work to be required of each Trade Contractor and shall assist the Owner in the preparation of the Request for Qualifications for Trade Contractors to be used to solicit responses from eligible Trade Contractors and to prequalify Trade Contractors for participation in the Project.
- (b) Request for Bids for Trade Contractor Services pursuant to M.G.L. c. 149A, §8(g)
 - 1. The Designer shall assist and advise the Owner in the preparation of the Invitation for Bids for Trade Contractor services in accordance with the provisions of M.G.L. c. 149A, §8.
 - 2. The Designer shall attend all pre-bid conferences and meetings.

(c) Trade Contractor Bid Review

1. The Designer shall attend all bid openings and shall review all Trade Contractor bids in conjunction with the Owner's Project Manager and CM at Risk to determine responsiveness, completeness, accuracy, price and conformance to the requirements of M.G.L. c.149A, § 8(g)-(i), and to provide technical guidance to the Owner regarding the acceptance or rejection of any Trade Contractor bid. Within five business days after the respective bid opening dates, the Designer shall advise the Owner in writing of the Designer's opinions as to the bids of Trade Contractors (and of other bidders if necessary).

7.8.8 Selection of Subcontractors Who Are Not Trade Contractors pursuant to M.G.L. c.149A, § 8(j) ("Non-Trade Contractors")

(a) Non-Trade Contractor Bidding

1. The Designer shall review the detailed bidding information developed by the CM at Risk in accordance with M.G.L. c. 149A, § 8(j) for accuracy, completeness, coordination of scope and conformance with the construction documents.

(b) Non-Trade Contractor Bid Review and Award

- 1. The Designer shall attend all bid openings and scoping meetings if permitted or otherwise allowed by law, and, in conjunction with the Owner's Project Manager and CM at Risk, the Designer shall review all Non-Trade Contractor bids for responsiveness and completeness and advise the Owner on the acceptance or rejection of any Non-Trade Contractor bids by the CM at Risk. The Designer shall, in conjunction with the OPM, attend all final scope and negotiation meetings conducted by the CM at Risk. The Designer shall, within five working days of the respective bid opening dates, advise the Owner in writing of the Designer's opinions as to the bids of Non-Trade Contractors.
- 7.9 Construction Administration Phase Obligations During Construction: Following the execution of the Owner-CM at Risk Agreement, the Designer shall undertake certain of the obligations of administering the Owner-CM at Risk Agreement on behalf of the Owner, provided that Designer shall not be subject to provisions of the Owner-CM at Risk Agreement that would have the effect of expanding Designer's responsibilities or liabilities under this Contract without Designer's written consent. Services during this phase include, but are not necessarily limited to:
 - 7.9.1 Upon commencement of construction activities for the Work or early bid packages or at times established in Project schedules, the Designer shall:
 - (a) Furnish the CM at Risk with information for establishing lines and grades and such supplemental drawings as are reasonably needed to implement the intent of the Construction Contract Documents;
 - (b) With reasonable promptness and in accordance with schedules agreed upon by the Designer and CM at Risk, observe testing when required under this Contract, and review and act upon samples, schedules, shop drawings and other submissions from the CM at Risk;
 - (c) Prepare, maintain and update logs for all submittals;
 - (d) Visit the site at intervals appropriate to the stage of construction, weekly or as otherwise agreed to by the parties, and observe the progress of the Work, issue written progress reports, and attend job meetings, and review and respond to meeting minutes prepared by the Owner's Project Manager, and to determine in general if the Work observed is being built in a manner indicating the Work when completed will be in accordance with approved Construction Contract Documents;

- (e) Collaborate with the on-site Project Representative of the OPM to identify and monitor issues of concern relative to the progress of the Work, and establish communications processes to help assure that matters of mutual concern are exchanged on a timely basis with one another, the OPM, CM at Risk, Commissioning Consultant, and Owner;
- (f) On a weekly basis, make specific recommendations on rejection of any Work observed by the Designer that fails to conform to the Construction Contract Documents, and observe corrected Work;
- (g) Require each Subconsultant engaged in accordance with Article 5 to make visits weekly or as otherwise agreed to by the parties during the progress of any work to which that Subconsultant 's services relate, and to report upon it in writing to the Designer;
- (h) Recommend actions to be taken which may include condemnation or rejection of any work that the Designer determines fails to conform to the Owner- CM at Risk Agreement;
- (i) Review and recommend appropriate action for proposed requests for changes and where required by the Owner, prepare documents associated with requests for a change in any Construction Contract Documents. Compensation for change order work by the Designer shall be determined in accordance with Article 10;
- (j) Conduct semi-final and final inspections of the Project and report the results of such inspections in writing to the Owner;
- (k) In association with the Commissioning Consultant, review the report by such Commissioning Consultant on the balancing of air and water circulation systems;
- (l) In association with the Commissioning Consultant, review the report by such Commissioning Consultant on the setting and adjustment of automatic controls;
- (m)In a timely manner, decide all questions regarding interpretation of, or compliance with, the Construction Contract Documents, except as the Owner may in writing otherwise determine;
- (n) In association with the Commissioning Consultant, review the recommendations of such Commissioning Consultant for requirements upon operating and maintenance documents and building user training events and instructional media as established in the Construction Contract Documents; such Commissioning Consultant or OPM shall coordinate involvement of contracting parties, the Designer, and Owner;
- (o) Furnish the Record Drawings as submitted by the CM at Risk in accordance with 7.9.3, and other required documents;

- (p) Assist the Owner in providing the written CM at Risk Evaluations required of the Owner pursuant to M.G.L. c.149 §44D(7) at the completion of approximately 50% of the Construction Phase on forms prescribed by M.G.L. c.149 §44D(16);
- (q) Perform inspections of the work as necessary to prepare a punch list identifying each incomplete or deficient Work item and performing reinspections to authorize removal of satisfactorily completed Work items from the punch list, or to determine that the Project is complete. In association with the OPM, a cost shall be assigned to each incomplete or deficient Work item when it has been determined that the Project has reached Substantial Completion; and
- (r) Receive from the CM at Risk all maintenance and operating manuals, occupancy permits, guarantees and other similar relevant materials.
- 7.9.2 The Designer shall submit to the Owner's Project Manager within 48 hours all requisitions for payment submitted by the CM at Risk in the form required by the Owner. The Designer may establish procedures with the CM at Risk for advance notification of requisition and/or draft version processing. With respect to each such requisition, the Designer shall certify to the best of its knowledge that the percentage of Work included in the requisition is accurate and that the work performed is in accordance with the Construction Contract Documents. In the event the Designer does not approve the requisition exactly as submitted by the CM at Risk, the Designer shall forward it for payment to the Owner's Project Manager dated and signed with corrections and with an accompanying letter of explanation setting forth the Designer's objections and recommended changes. The Designer shall coordinate the required visits of its own staff and those of its Subconsultants, to the construction site so as to enable it to submit to the Owner's Project Manager the CM at Risk's monthly requisition for payment. Timely payments to the CM at Risk are required by M.G.L. c. 30, § 39K. Therefore, the Designer shall establish procedures to help assure either immediate mail or messenger delivery of the requisition for payment to the Owner's Project Manager, and shall process requisitions for payment within five business days after receipt of the same, provided the CM at Risk has submitted a full and complete requisition for payment in the correct form.
- 7.9.3 Prior to issuance of the Certificate of Substantial Completion, the Designer shall obtain from the CM at Risk as-built drawings, including drawings showing the actual installation of the site utilities, plumbing, heating, ventilating and electrical work under the Owner-CM at Risk Agreement, and recording all changes. The Designer shall ascertain that changes authorized by change orders are shown on the CM at Risk's as-built drawings, but Designer shall be entitled to rely upon the accuracy and completeness of the CM at Risk's as-built information, and shall forward such to the Owner as Record Drawings.
- 7.9.4 Issue the Certificate of Substantial Completion of Construction.

- 7.9.5 The Designer shall meet with the Owner monthly during this Phase.
- 7.10 <u>Completion Phase</u>: Upon acceptance of the Certificate of Substantial Completion of Construction by the Owner, the Designer shall thereafter provide the following services:
 - 7.10.1 With respect to a completed Project, preparing a Certificate of Final Completion.
 - 7.10.2 With respect to a punch list, re-inspecting the work up to three times in order to determine that the punch list work is satisfactorily completed.
 - 7.10.3 Reviewing and certifying the CM at Risk's Application(s) and Certificate(s) for Payment as necessary.
 - 7.10.4 Attending meetings as reasonably necessary in the opinion of the Owner or Owner's Project Manager, unless such meetings involve continued discussions of incomplete or deficient work and the Basic Services punch list site visits have been expended. In such instance, the meetings shall be paid for as Extra Services.
 - 7.10.5 Using the as-built information maintained by the CM at Risk during construction referred to in Article 7.9.3, and revising the applicable original reproducible drawings and electronic media drawings on the basis of the as-built drawings, provided that Designer shall be entitled to rely upon the accuracy and completeness of the CM at Risk's as-built information. Upon completion of the required drafting and editing, provide one set of mylar reproducibles, two sets of prints and two (2) electronic version copies to the Owner which shall become the property of the Owner. The cost for printing the mylar reproducibles and two sets of prints are Reimbursable Expenses.
 - 7.10.6 Ten (10) months after the date of substantial completion, performing one (1) site inspection and preparing a list of construction warranty deficiencies. The Designer shall consult with the Commissioning Consultant upon the acceptability of warranty compliance requirements and response actions.
 - 7.10.7 Informing the Owner in writing, through the Owner's Project Manager, of all such warranty deficiencies that should be addressed.
 - 7.10.8 Performing one (1) site inspection within a further sixty (60) days to see that all such warranty deficiencies have been corrected.
 - 7.10.9 Evaluation of CM at Risk: The Designer shall assist the Owner with providing the written CM at Risk Evaluations required of the Owner pursuant to M.G.L. c.149 § 44D(7) within 70 days of the date of Substantial Completion for construction, on forms prescribed by M.G.L. c.149 § 44D(16).
 - 7.11.10 The Designer shall assist the Owner in providing the written summary report on the Project to the Office of the Inspector General as required by the provisions of 945 CMR 2.09
 - 7.10.11 Two (2) suitably bound, legible copies of all original design and quantity calculations including those pertinent to change orders and shop drawings, if applicable, shall be

ATTACHMENT C:

Standard Designer Application Form for Municipalities and Public Agencies not within DSB Jurisdiction (Updated July 2016)



```
THINONALLY LEFT BLANK
```

Municipalities & Other Public Agencies Form Page 1 Updated July 2016

5

		I	Τ	I		
rertisement (List Up To	Thousands) Fee for Work for Which Firm Was Responsible					
eas Listed In The Adv	e. Project Cost (In Thousands) Construction Costs (Actual, Or Estimated If Not Completed)					
ualifications In The Ar	d. Completion Date (Actual Or Estimated)					
Include ONLY Work Which Best Illustrates Current Qualifications In The Areas Listed In The Advertisement (List Up To	C. Client's Name, Address And Phone Number (Include Name Of Contact Person)					
	b. Brief Description Of Project And Services (Include Reference To Relevant Experience)					
8a.	a.	(1)	(2)	(3)	(4)	(2)

Municipalities & Other Public Agencies Form Page 5

8b.		List Current and Relevant Work By Sub-Consultants Which Best Illustrates Current Qualifications In The Areas Listed In The A Consultant). Use Additional Sheets Only As Required For The Number Of Sub-Consultants Requested In The Advertisement.	List Current and Relevant Work By Sub-Consultants Which Best Illustrates Current Qualifications In The Areas Listed In The Advertisement (Up To But Not More Than 5 Projects For Each Sub-Consultants Requested In The Advertisement.	isement (Up To But N	Not More Than 5 Proj	ects For Each Sub-
Sub	Sub-Consultant Name:					
a.	Project Name and Location	b. Brief Description Of Project and	c. Client's Name, Address And Phone	d. Completion	e. Project Cost (In Thousands)	Thousands)
	Principal-In-Charge	Services (include Reference Lo Relevant Experience	Number. Include Name Of Contact Person	Date (Actual Or Estimated)	Construction Costs (Actual, Or Estimated If Not Completed)	Fee For Work For Which Firm Was/Is Responsible
<u> </u>						
(5)						
(3)						
(4)						
(2)						

```
THINONALLY LEFT BLANK
```

ATTACHMENT D:

Certifications: None Included

```
THINONALLY LEFT BLANK
```

ATTACHMENT E:

MSBA's Designer Selection Panel's Procedures

```
THINONALLY LEFT BLANK
```

Massachusetts School Building Authority Designer Selection Procedures

Section 1: Introduction

The following designer selection process has been adopted by the Massachusetts School Building Authority (MSBA) pursuant to Massachusetts General Laws, Chapter 7C, Sections 44 through 58 for the procurement of designers, and programmers by cities, towns, regional school districts, and independent agricultural and technical schools seeking funding from the MSBA for public school construction projects where the estimated construction cost is equal to or greater than \$5,000,000.00 (or other such amount as may be determined from time to time by the Executive Director of the MSBA), except for the MSBA's model schools program. Designer selection for public school construction projects where the estimated construction cost is less than \$5,000,000.00 (or other such amount as may be determined from time to time by the Executive Director of the MSBA) shall be conducted pursuant to Massachusetts General Laws, Chapter 7C, Section 54, by the respective city, town, regional school district or independent agricultural and technical school and in accordance with the MSBA's Designer Selection Guidelines.

Section 2: Designer Selection Panel

- A. The MSBA Designer Selection Panel (DSP) shall be composed of the following individuals who shall be appointed to the DSP by the MSBA's Executive Director ("Executive Director") in accordance with following procedures:
 - 1. The Executive Director, ex officio, or his/her designee;
 - 2. Three (3) MSBA staff members associated with project management, design and/or construction oversight selected by the Executive Director;
 - 3. One (1) public member selected by the Executive Director;
 - 4. One (1) member who is a Massachusetts registered architect or architect emeritus as recommended by the Boston Society of Architects;
 - 5. Two (2) members who are Massachusetts registered architects or architect emeritus selected by the Executive Director;
 - 6. One (1) member who is a Massachusetts registered engineer as recommended by the American Council of Engineering Companies of Massachusetts;
 - 7. Two (2) members who are Massachusetts registered professional engineers selected by the Executive Director;
 - 8. One (1) member who is a representative of the construction industry as recommended by Associated General Contractors of Massachusetts;

- 9. One (1) member who is a representative of the construction industry as recommended by the Massachusetts Building Trades Council;
- 10. Three (3) members who are proposed by the respective city, town, regional school district, independent agricultural and technical school or other public agency that is the Eligible Applicant, as defined in M.G.L. Chapter 70B, Section 2 for the specific project under consideration, one (1) of whom shall be designated by the school committee, district school committee, or board of trustees of the Eligible Applicant, as the case may be; one (1) of whom shall be the superintendent of schools of the Eligible Applicant, ex officio, or his/her designee; and one (1) of whom shall be the chief executive officer of the city or town that is the Eligible Applicant, ex officio, or his/her/its designee or, in all other cases, a member of the School Building Committee designated by the School Building Committee. The appointment of members pursuant to this Section 2(A)(10) shall be subject to the execution of a certification by each such member that the member has read and understands these procedures and the Designer Selection Guidelines.
- B. Members proposed or recommended by the societies or associations pursuant to subsections 2(A)(4), 2(A)(6), 2(A)(8), and 2(A)(9) above and the members proposed by the Eligible Applicant pursuant to subsection 2(A)(10) above shall be subject to appointment by the Executive Director who reserves the right, within his/her discretion, not to appoint or to disapprove the appointment of said proposed or recommended members. In considering the appointment of members proposed by the Eligible Applicant pursuant to subsection 2(A)(10), the Executive Director may consider, among other things, the extent to which the three (3) proposed members, as a whole, represent the interests of the Eligible Applicant.
- C. The Executive Director shall appoint a chairperson from one of the members appointed to the DSP pursuant to subsections 2(A)(3) through 2(A)(9) above, who is a registered architect, architect emeritus or registered professional engineer and who shall also serve as chairperson of any subcommittee of the DSP.
- D. All meetings of the DSP shall be open to the public unless the DSP votes to go into executive session by a roll call vote and announces the purpose of the executive session and whether the DSP will convene in open session at the conclusion of the executive session. Any action taken by the DSP in executive session shall be by a roll call vote.
- E. The presence of nine (9) members, no less than four (4) of whom shall be registered architects, architects emeritus or registered professional engineers, shall constitute a quorum. The DSP shall not conduct any business without the presence of a quorum. The affirmative vote of a simple majority of the members present and voting shall be necessary and sufficient for any action taken by the DSP. No vacancy in the membership of the DSP shall impair the right of a quorum to exercise all the rights and duties of the DSP. In the absence of a quorum, the Chairperson may recess a meeting to some other time or until a quorum is obtained.
- F. Subject to the discretion of the Executive Director, each member appointed pursuant to subsections 2(A)(3) through 2(A)(9) shall serve for a two-year term provided that every member that is appointed by the Executive Director shall continue to serve until a successor has been appointed to the DSP by the Executive Director. Members representing the Eligible Applicant who are appointed pursuant to subsection 2(A)(10) shall serve only while the DSP

conducts business directly related to the selection of a designer for the project being proposed by that particular Eligible Applicant.

G. No member of the DSP shall participate in the selection of a designer as a finalist for any project if the member's participation would constitute a conflict of interest or an appearance of conflict in violation of M.G.L. Chapter 268A.

Section 3: Public Notice

- A. Each contract for designer services for a project subject to these procedures shall be publicly advertised in a newspaper of general circulation in the area in which the project is located or is to be located and, in the Massachusetts Central Register at least two weeks before the deadline for filing applications. The public notice shall contain:
 - 1. A description of the project, including the specific designer services sought, the time period within which the project is to be completed, and, if available, the estimated construction cost;
 - 2. If there is a program for the project, a statement of when and where the program will be available for inspection by applicants, and when and where a briefing session will be held for applicants and if there is not a program for the project, a statement to the effect:
 - 3. The qualifications required of applicants for the projects;
 - 4. The categories of designers' consultants, if any, for which applicants must list the names of consultants which the applicant may choose to use;
 - 5. Whether the fee has been set or will be negotiated, and if the fee has been set, the amount of the fee;
 - 6. The deadline for submission of applications;
 - 7. The person and address from which application forms may be obtained and, when completed, to whom they may be delivered;
 - 8. Any other pertinent information that may be required by law or deemed appropriate by the MSBA.
- B. The individual designated by the Eligible Applicant to be in charge of procurement for a project who holds the Massachusetts Certified Public Purchasing Official Program certification shall certify that the public notice and all other documents issued pursuant to the selection of a designer, including, but not limited to, program descriptions and request for services, have been prepared and issued in conformance with these procedures and Massachusetts General Laws, Chapter 7C, Sections 44 through 58.

Section 4: Master File Brochure and Application

- A. Prior to filing an application for any project, designers shall first file a Master File Brochure with the DSP containing the following information:
 - 1. Certification that the applicant, if applying to perform design services other than preparation of studies, surveys, soil testing, cost estimates or programs, is a designer as defined in M.G.L. Chapter 7C, Section 44 paragraph (b);
 - 2. The names and addresses of all partners, if a partnership, of all officers, directors and all persons with an ownership interest of more than five per cent in the applicant if not a partnership;
 - 3. The registration number and status of each such person in every jurisdiction in which such person has ever been registered as an architect, landscape architect or engineer;
 - 4. A list of all projects for all public agencies within the Commonwealth for which the applicant has performed or has entered into a contract to perform design services within the five-year period immediately preceding the filing of the information required in this section;
 - 5. A list of all current projects for which the applicant is performing or is under contract to perform any design services; and
 - 6. If the applicant is a joint venture, the information required in this section shall be required for each joint venturer, as well as for the joint venture itself.
- B. The DSP shall keep a permanent record of the Master File Brochures. Each designer shall update its Master File Brochure on an annual basis and shall make current the lists of projects required under Section 4(A)(4)-(6) with each application filed.
- C. An applicant to perform design, programming or feasibility study services on a project must file, in addition to the Master File Brochure, a written application prescribed by the DSP relating to the applicant's experience, ability, and qualifications.

Every application or Master File Brochure filed shall be sworn to under penalties of perjury. Any applicant who has been determined by the DSP to have filed materially false information shall be disqualified by the DSP from further consideration for any project for such time as the DSP determines is appropriate.

Section 5: Selection Criteria

- A. Minimum qualifications shall include:
 - 1. Must be a qualified Designer within the meaning of M.G.L. Chapter 7C, Section 44 employing a Massachusetts registered architect or engineer responsible for and being in control of the services to be provided.
 - The Massachusetts registered architect or engineer responsible for and being in control
 of the services to be provided for the Designer must have successfully completed the
 Massachusetts Certified Public Purchasing Official Program seminar "Certification for

School Project Designers and Owner's Project Managers," as administered by the Office of the Inspector General of the Commonwealth of Massachusetts, and must maintain certification by completing the "Recertification for School Project Designers and Owner's Project Managers" seminar every three years thereafter. Proof of recertification or registration in the next recertification seminar for which space is available must be provided.

3. Pursuant to M.G.L. Chapter 7C, Section 6, the Designer must agree to contract with minority and women-owned businesses as certified by the Supplier Diversity Office (SDO). The amount of participation that shall be reserved for such enterprises shall not be less than seventeen and nine tenths percent (17.9%) of the contract price for combined minority business enterprises (MBE) and women-owned business enterprises (WBE). Applicants must include a reasonable representation of both MBE and WBE firms that meets or exceeds the combined goal.

B. Other criteria for selection of finalists shall include:

- 1. Prior similar experience best illustrating current qualifications for the specific project.
- 2. Past performance of the firm, if any, with regard to public, private, DOE-funded, and MSBA-funded projects across the Commonwealth, with respect to:
 - a) Quality of project design.
 - b) Quality, clarity, completeness and accuracy of plans and contract documents.
 - c) Ability to meet established program requirements within allotted budget.
 - d) Ability to meet schedules including submission of design and contract documents, processing of shop drawings, contractor requisitions and change orders.
 - e) Coordination and management of consultants.
 - f) Working relationship with contractors, subcontractors, local awarding authority and MSBA staff and local officials.
- 3. Current workload and ability to undertake the contract based on the number and scope of projects for which the firm is currently under contract.
- 4. The identity and qualifications of the consultants who will work on the project.
- 5. The financial stability of the firm.
- 6. The qualifications of the personnel to be assigned to the project.
- 7. Geographical proximity of the firm to the project site or willingness of the firm to make site visits and attend local meetings as required by the client.
- 8. Any other criteria that may be required by law or that the DSP considers relevant to the project.

Section 6: Selection Process

- A. Cities, towns, regional school districts, and independent agricultural and technical schools subject to these procedures shall not rank or pre-rank applicants. Rankings shall occur only by vote of the DSP in accordance with these procedures and shall occur only after interviews, if allowed by vote of the DSP, have been concluded by the DSP.
- B. In the event that, upon reaching the deadline for submission of applications, three or fewer designer applications are received for a project, the Eligible Applicant may choose to modify the project description, estimated construction cost, program, desired designer qualifications, fee information, or other project information as necessary to attract interested designer applicants and begin the selection process again, starting with re-advertisement pursuant to Section 3: Public Notice. Should the Eligible Applicant choose to proceed with three or fewer designer applications and not re-advertise, the following procedure shall be followed:
 - 1. The Eligible Applicant designee shall submit a statement that explains why the Eligible Applicant may have received three or less applications for the proposed project, The explanation should include but not necessarily be limited to:
 - a. A description of the public advertisement including the names of the publications in which the advertisement was placed and the date(s) in which the advertisement was published.
 - b. A description of the pre-proposal conference, if any, including the date, time, and location of the conference and names of attendees and the firms they represent.
 - 2. The Eligible Applicant designee and/or the OPM shall contact those design firms that attended the pre-proposal conference/walkthrough but did not submit an application and summarize why an application was not submitted for the proposed project.
 - 3. Legal counsel for the Eligible Applicant (i.e. town counsel or city solicitor) and the individual designated by the Eligible Applicant to be in charge of procurement for a project who holds the Massachusetts Certified Public Purchasing Official Program certification shall certify as to the adequacy and completeness of the procurement activity undertaken by the Eligible Applicant.
 - 4. At the discretion of the chairperson and with the concurrence of the three DSP members representing the Eligible Applicant, the DSP may forego the initial application review and invite all the designer applicants to appear for an interview before the DSP.
- C. The DSP may require any number of applicants to:
 - 1. Appear for an interview before the DSP;
 - 2. Present a written proposal to the DSP through the Eligible Applicant; or
 - 3. Participate in a design competition held by the DSP through the Eligible Applicant.
- D. The DSP shall use the following procedures to rank three (3) finalists in order of qualifications from among the applicants for a particular project:

- 1. Prior to a DSP meeting at which the selection of finalists will be made or discussed, each member of the DSP shall be given a copy of each designer's application for his or her review.
- 2. At the DSP meeting, the DSP shall consider each application alphabetically or by some other method that may be determined by the chairperson from time to time.
- 3. When recognized by the chairperson, members of the DSP may comment or ask questions related to the selection process or the applications before the DSP.
- 4. Any potentially disqualifying deficiencies in an application should be noted in the record of the meeting.
- 5. After each member of the DSP has been given an opportunity to comment or ask questions, at the direction of the chairperson, each member of the DSP who is present shall utilize a ballot form provided by the MSBA to assign points to his or her top three (3) choices in order of qualifications so that each number one choice shall receive three (3) points, each number two choice shall receive two (2) points, and each number three choice shall receive one (1) point. The completed ballot forms shall be signed by each member and submitted to the DSP Administrator who shall tally the total points awarded to each applicant. The chairperson shall then read aloud the total points awarded to each of the applicants.
- 6. Once the point totals have been read aloud by the chairperson, the DSP may request interviews of the applicants with the highest point totals by the following procedure: Upon motion of one of the members, duly seconded by one of the other members, the DSP may vote to interview the applicants with the highest point totals.
- 7. If the DSP does not vote to conduct interviews, the DSP shall then vote to rank three (3) finalists in order of qualifications. If the DSP votes to conduct interviews, the DSP shall defer the ranking of the three (3) finalists until after the interviews have been concluded.
- 8. If the DSP votes to conduct interviews, the chairperson shall schedule the time and place of the interviews and written notice shall be given to the firms to be interviewed. Interviews shall be conducted in open session except that the chairperson may order competing firms, their agents and employees, to leave the meeting room during the interviews of their competitors. The MSBA may, within its discretion, develop standard questions to be answered or topics to be discussed by the applicants in the interview. Once the interviews have been concluded, at the direction of the chairperson, the DSP shall award points to the each of the firms in accordance with the procedures set forth in subsection 6(C)(5). Once the point totals have been read aloud by the chairperson, the DSP shall then vote to rank three (3) finalists in order of qualifications
- 9. In the event of a tie for the first, second or third highest point totals awarded to applicants by the DSP under Section 6(C)(5) or 6(C)(8), the chairperson shall determine, in his or her complete discretion, the procedure by which the tie shall be broken. The chairperson shall then read aloud the total points awarded to each of the applicants. Once the point totals have been read aloud by the chairperson, the DSP shall then vote to rank three (3) finalists in order of qualifications.

Once the DSP has voted to rank the top three (3) firms in order of qualifications, the MSBA shall transmit a list of the three (3) finalists ranked in order of qualifications to the Eligible Applicant along with a record of the final vote of the DSP on the selection and a written statement explaining the DSP's reasons for its ranking of the finalists.

Section 7: Award of Contract

- A. The authority to award a contract for designer services for a project that will receive funding from the MSBA is vested with the Eligible Applicant and subject to the approval of the MSBA.
- B. In the selection of a designer when the fee for designer services has been set prior to advertisement, the Eligible Applicant shall appoint a designer from the ranked list transmitted by the MSBA to the Eligible Applicant in the order of qualifications as determined by the DSP. If the Eligible Applicant proposes to select any designer other than the one ranked first by the DSP, it shall file a written justification for the proposed appointment with the DSP and shall not proceed until it has obtained written approval to proceed from the Executive Director.
- C. When the fee for designer services is to be negotiated, the Eligible Applicant shall review the list transmitted by the MSBA in the order of qualifications as determined by the DSP and may exclude any designer from the list if a written statement of reasons for the exclusion is filed with the DSP. The Eligible Applicant shall then appoint a designer based upon a successful fee negotiation. The Eligible Applicant shall first negotiate with the first ranked designer remaining on the list. Should the Eligible Applicant be unable to negotiate a satisfactory fee with the first ranked designer within thirty (30) days, negotiations shall be terminated and negotiations undertaken with the remaining designers, one at a time, in the order in which they were ranked by the DSP, until an arrangement is reached. Should the Eligible Applicant be unable to negotiate a successful fee with any designer initially selected by the DSP, the DSP shall recommend additional finalists in accordance with a procedure to be determined by the chairperson of the DSP that is not inconsistent with the procedures set forth in Section 6(B) above. The Eligible Applicant may require a finalist with whom a fee is being negotiated to submit a fee proposal and to provide current cost and pricing data on the basis of which the designer's fee proposal may be evaluated.

Section 8: Continued or Extended Services

- A. The Eligible Applicant may appoint a designer to perform continued or extended services that were not contemplated in the original public notice if the following conditions are met:
 - 1. A written statement is filed with the DSP explaining the reasons for the continuation or extension of services;
 - 2. The program for the design services is filed with the DSP;
 - 3. MSBA staff has made a written determination that the request for continued or extended services is otherwise in compliance with the MSBA's regulations, policies, procedures, and guidelines and the provisions of the feasibility study agreement, project scope and budget agreement, and/or project funding agreement, as applicable;

4. The DSP approves the appointment of the designer for continued or extended services and states the reason therefore.

Section 9: Emergency Designer Selection Process

- A. If a situation arises in accordance with Chapter 7C, Section 53, which has been declared an "emergency" by the Executive Director, an Eligible Applicant may request an emergency selection of a designer.
- B. In consultation with the technical staff of the MSBA, the Eligible Applicant shall prepare a proposed scope of work, an estimate of the cost of construction for the designer's services, and submit this, and any other relevant information to the Executive Director.
- C. In lieu of public advertisement, the Executive Director or his/her designee will consult with the Eligible Applicant to select three to six qualified firms who have Master File Brochures on file, to solicit to perform this work.
- D. The MSBA staff will poll an ad-hoc committee of three members of the DSP to select at least three qualified finalists and forward the names of the finalists to the Eligible Applicant with a written statement explaining the committee's reasons for its choice(s).
- E. The Eligible Applicant will select one of the three finalists to perform the work and forward the name of the selected firm to the DSP with a written statement explaining the reasons for its choice.

Section 10: Statutory Representations by the MSBA

- A. The projects of the MSBA and the Eligible Applicants are not subject to the jurisdiction of the Division of Capital Asset Management and Maintenance.
- B. The DSP procedures substantially incorporate the procedures required of the Commonwealth's Designer Selection Board in M.G.L. Chapter 7C, Section 45 through 53, inclusive, and Section 55.

```
THINONALLY LEFT BLANK
```



Greater Fall River Vocational School District

THE DIMAN REGIONAL VOCATIONAL TECHNICAL HIGH SCHOOL

END OF REQUEST FOR DESIGNER SERVICES (RFS)

THANK YOU!

